INTERACTIONS AMONG COLLEGE AND UNIVERSITY FACULTY AND STUDENTS INVOLVED IN ACADEMIC STUDENT ORGANIZATIONS: AN ANALYSIS OF QUALITATIVE AND QUANTITATIVE ENGAGEMENT

A Dissertation

by

PEGGY CAROL HOLZWEISS

Submitted to the Office of Graduate Studies of Texas A&M University in partial fulfillment of the requirements for the degree of

DOCTOR OF PHILOSOPHY

August 2010

Major Subject: Educational Administration

INTERACTIONS AMONG COLLEGE AND UNIVERSITY FACULTY AND STUDENTS INVOLVED IN ACADEMIC STUDENT ORGANIZATIONS: AN ANALYSIS OF QUALITATIVE AND QUANTITATIVE ENGAGEMENT

A Dissertation

by

PEGGY CAROL HOLZWEISS

Submitted to the Office of Graduate Studies of Texas A&M University in partial fulfillment of the requirements for the degree of

DOCTOR OF PHILOSOPHY

Approved by:

Co-Chairs of Committee,	Bryan Cole
	Kelli Peck Parrott
Committee Members,	Fred Bonner
	Richard Woodman
Head of Department,	Fredrick Nafukho

August 2010

Major Subject: Educational Administration

ABSTRACT

Interactions among College and University Faculty and Students Involved in Academic Student Organizations: An Analysis of Qualitative and Quantitative Engagement. (August 2010)

Peggy Carol Holzweiss, B.S., Texas A&M University; M.S., Texas A&M University Co-Chairs of Advisory Committee: Dr. Bryan Cole Dr. Kelli Peck Parrott

The purpose of this study was to describe what kind of student-faculty interactions are occurring in the context of academic student organizations as well as identify the quality and quantity of such interactions and what factors are involved with meaningful interactions. The study also determined how these interactions might differ from those occurring in other college activities and how the factors of classification, organizational status, and institutional size relate to quality and quantity of interactions in both contexts.

An instrument was developed for the study using all student-faculty interactions identified by previous researchers. A total of 104 undergraduate students from four different institutions responded to the instrument. All were members of an academic student organization.

Results indicated that almost all of the students had faculty advisors for their organizations. In addition, 99% of participants had at least one interaction with faculty since they started college. Another 81% had at least one interaction with faculty within

their academic student organization, and 96% had at least one interaction with faculty within their other college activities. Over three-quarters (78%) said they had interactions with faculty in both their academic student organization and other college activities.

Interactions were found to occur but were infrequent overall and did not differ significantly between academic student organizations and other college activities. However, there was a trend for participants to have a higher quantity of interactions within their organizations than through other activities. Most of the interactions reported by participants lasted longer than 10 minutes, which was the standard by which some researchers measured quality.

For institutional size, a pattern of responses indicated that participants from small institutions may have a higher quality and quantity of interactions with faculty than their peers from large institutions.

Whether or not a student served as a member or a leader in their academic student organization did appear to impact the interactions they had with faculty. Leaders reported more interactions with faculty in their organizational context than did members.

DEDICATION

TO:

All three of my boys – Bob, Nick, and Jake.

You are my inspiration, my support, and my life.

We go together like PB-N-J.

ACKNOWLEDGEMENTS

I would like to thank both of my committee chairs - Dr. Bryan Cole and Dr. Kelli Peck Parrott. Dr. Cole, you provided many words of wisdom and an abundance of knowledge for this process. I always knew where I stood and what I needed to do to accomplish the necessary tasks. Thank you for taking so much time to meet with me and provide feedback on whatever I was doing. Dr. Peck Parrott, you were an endless source of encouragement and support and I would not have been able to make it through the program without you. I am delighted to call you a mentor and a friend.

I also would like to thank my other committee members – Dr. Fred Bonner and Dr. Richard Woodman. You provided invaluable advice and support throughout this process. You are truly appreciated as an integral part of my learning curve.

I want to express a special appreciation for the students at each campus who either participated in the pilot testing phase or the actual data collection phase of this project. Thank you for being so willing to share your experiences and your time.

To the ladies in Student Life Studies – You laughed when I finally broke down and said I would willingly pursue a doctorate after swearing it would never happen, but you have been nothing but supportive during the process. I have learned a great deal from observing all of you over the years and you have shaped who I have become.

To Beth Pang, Nicolette Yevich, and Bari Brookins – You are amazing friends who could always be counted on to provide a listening ear, a word of encouragement, and endless humor. Thanks for putting up with all of the lunches, phone calls, and emails that it took to get me here. Know that I am up for returning the favor through your own doctoral processes.

To my mom, Ruth Philpot, who never stopped telling me how proud she was of me. I have never forgotten whose daughter I am, and I am proud to let people know.

To my dad, Jerry Philpot, you provided wonderful support and encouragement from afar. Even though we do not live in the same state, I know you are always there for me and I appreciate it more than you know.

To the rest of my family and friends, you helped distract me with other important events like holiday gatherings and graduations. Thanks for the emails, phone calls, Facebook exchanges, and support you have provided.

To Jake and Nick – You helped me stay focused on the important things in life like spending time together with activities such as reading, watching movies, playing games, and exploring new activities. You are my joy in life.

And, finally, to my husband, Bob. This never would have happened without you. I stopped counting all of the times you served as the primary parent so I could attend class, finish papers, or participate in group projects. You readily overlooked my minimal participation in household activities and picked up my considerable slack. And you put up with my complaints, frustrations, and exhaustion. You are my world and I would marry you all over again. There is no woman more blessed than me because I have you.

TABLE OF CONTENTS

ABSTRACT		
DEDICATION		
ACKNOWLEDGEMENTS	vi	
TABLE OF CONTENTS	viii	
LIST OF TABLES	X	
CHAPTER		
I INTRODUCTION	1	
Background Statement of the Problem Significance of the Study Significance Theoretical Framework Research Questions Operational Definitions Operational Definitions Organization of the Dissertation History of Student-Faculty Interactions	1 8 9 17 19 21 23 25 25	
Importance of Student-Faculty Interactions History of Student Involvement in Organizations Importance of Student Organization Involvement Summary	29 41 46 49	
III METHODOLOGY	50	
Research Paradigm Used for the Study Population and Sample Instrument Data Collection Data Analysis	50 51 55 57 59	

CHAPTER		Page
IV	RESULTS	66
	Purpose of the Study	66
	Demographics of Participants	66
	Research Question 1	68
	Research Question 2	71
	Research Question 3	77
	Research Question 4	82
	Research Question 5	91
	Research Question 6	94
	Research Question 7	101
	Limitations	106
	Summary of Findings	109
V	DISCUSSION AND CONCLUSIONS	114
	Discussion of Key Issues	114
	Recommendations for Practice	121
	Recommendations for Further Research	124
	Conclusions	137
REFEREN	REFERENCES	
APPENDIX	X A SOURCE OF ITEMS USED IN SURVEY INSTRUMENT	155
APPENDIX	X B SURVEY INSTRUMENT	158
VITA		166

LIST OF TABLES

TABLE		Page
1	Undergraduate Populations and Responses by Institution	67
2	Gender and Classification by Institution	67
3	Overall Occurrence of SFIs for All Study Participants	69
4	Frequency of SFIs by Organizational Status	70
5	Quantity of SFIs for All Study Participants Since They Started College	72
6	Low and High Quantity SFIs for All Study Participants	75
7	Quantity of SFIs for ASOs and Other College Activities	76
8	Quality of SFIs for All Study Participants	78
9	Low and High Quality SFIs for All Study Participants	80
10	Quality of SFIs for ASOs and Other College Activities	81
11	Composite Score Means for Overall SFIs – Descending Order	84
12	Combined Quantity and Quality of SFIs for Study Participants – Low Composite – Descending Order	86
13	Combined Quantity and Quality of SFIs for Study Participants – High Composite – Descending Order	87
14	Combined Quantity and Quality of SFIs for ASOs and Other College Activities	89
15	Frequency of SFIs by Classification	92
16	Occurrence of SFIs in ASOs and Other College Activities by Classification	93
17	Frequency of SFIs by Institutional Size	94

TABLE

18	Occurrence of SFIs in ASOs and Other College Activities by Institutional Size	96
19	Combined Quantity and Quality of SFIs for ASOs by Institutional Size	98
20	Combined Quantity and Quality of SFIs for Other College Activities by Institutional Size	99
21	Frequency of SFIs by Organizational Role	101
22	Occurrence of SFIs for ASOs and Other College Activities by Organizational Role	103
23	Combined Quantity and Quality of SFIs for ASOs by Organizational Role	104
24	Combined Quantity and Quality of SFIs for Other College Activities by Organizational Role	105

Page

CHAPTER I

INTRODUCTION

Background

According to Burke (2004), American society once trusted institutions of higher education to provide a quality education to citizens from all parts of the population. Financial support flowed without question as to how the money was being spent. Citizens and politicians expected institutions to spend the money on enhancing the quality of education and creating productive citizens but rarely asked for proof that these outcomes actually occurred. This social contract began to show signs of strain in the 1970s because of an economic recession. As financial revenues tightened, state governments started controlling the monetary support provided to higher education and demanded more accountability regarding how money was being spent and the benefits of that spending. In addition, financial support began shifting from higher education to other public services and caused higher education to begin competing for funds.

During the 1980s, a report called *A Nation at Risk* criticized the nation's public schools for not providing a quality education to students (Burke, 2004). All levels of education responded to the report by establishing goals and objectives for what students would learn at their institutions or within their systems. The state governments, in turn, started holding the schools accountable for the processes created to meet those goals and

This dissertation follows the style of the Journal of Higher Education.

objectives. This approach moved into the 1990s as a way to control the goals of an institution but still allow flexibility in how those goals were achieved.

In the current decade, the slowing economy and increasing financial burden on the state and federal governments has amplified concerns for how taxpayer funds are being spent and what is achieved as a result of those funds. In addition, higher education is increasingly being seen as more of a private rather than public benefit and government funding is shifting away from higher education and into direct public benefits such as health care and elementary education (Burke, 2004).

Constituents of higher education – students, parents, alumni, businesses, state governments, federal government – generally want a college education to mean that graduates will become productive members of society, increase their knowledge, obtain well-paying jobs, contribute to a stable economy, and develop practical life skills (Arnold & Kuh, 1999; Ewell, 1991). However, institutions are regularly challenged to do more to enhance these developmental outcomes while experiencing decreases in their financial support (Kuh, 1996; Massy & Zemsky, 1994).

As calls for accountability grow and government funding declines, institutions must compete for and retain students, especially highly qualified, paying students who can contribute to the overall reputation of the institution as well as provide a steady stream of income. On a basic level, the more students an institution can put in a classroom, the more money it has to conduct educational business and compete with other institutions to attract more students (Nadler & Nadler, 2001). However, students have many choices regarding where they pursue their college education. One way institutions compete for students is by increasing the reputation of the faculty. Faculty may be hired because they are well-known researchers in their academic discipline or the institution may demand an increase in research productivity and scholarly publications from existing faculty in order to increase the academic reputation of the department or college. By obtaining faculty members who are viewed as the best in their field, institutions can promote the idea that they provide a high quality education in return for the dollars being spent on it.

Unfortunately, with the focus on research and academic reputations, faculty members feel pressure to build their scholarly portfolios, sometimes to the detriment of other activities. Massy and Zemke (1994) call this the "academic ratchet" (p. 2). As the pressure to conduct research builds, other activities the faculty might have found fulfilling and productive must be set aside. This has the unintended consequence of faculty trying to decrease their teaching loads, institutional service, and meaningful interactions with students just to keep up with the demands of research. Altbach (2005) described additional pressures such as economic problems forcing publishers and grant foundations to cut back on opportunities for research. This increases the academic ratchet because faculty, especially those seeking tenure, must now spend even more time trying to achieve the feats of scholarly works that are expected to move along the tenure ladder. As faculty focus more on research and other administrative responsibilities, the potential for student learning decreases because faculty do not have the time to devote to this important endeavor (Nadler & Nadler, 2001). And some faculty who do set aside

time to interact with students may be viewed as neglecting their responsibilities (Golde & Pribbenow, 2000).

Complicating the issue beyond the ratchet, several scholars have demonstrated that faculty are working harder than ever. For instance, as budgets are cut, departments hire more part-time faculty in order to save money on salaries and benefits. However, as the numbers of part-time faculty increase, the number of full-time faculty decreases. This, in turn, limits the number of faculty available to participate in governance bodies, administrative committees, and curriculum development (Anderson, 2002). These responsibilities are in addition to the traditional teaching and research activities required of faculty. The result is that a typical faculty member works an average of 50 or more hours per week (Jacobs, 2004), teaches approximately four classes per academic semester, holds an average of seven office hours per semester, and writes approximately twelve scholarly publications (e.g., books, articles, reviews, etc.) over two years (Anderson, 2002). One national study concluded that only a small portion of faculty were able to excel at both teaching and research responsibilities simultaneously (Fairweather, 2002).

The report *Learning Reconsidered* encouraged institutions to reestablish student learning as the central mission of higher education and begin integrating classroom learning with student development processes (Keeling, 2004). Boyer (1987) explains that students want to learn more about themselves while attending college. They are motivated to understand how their college education will impact them after graduation and how they can best find a job that meets their personal goals and interests. This motivation can provide part of the requirements needed to create informal learning opportunities but institutions need to provide the organizational structure necessary to encourage learning in this manner. Terenzini, Pascarella, and Bliming (1996) recommend that institutions "blur the boundaries between students' academic and outof-class lives" by creating opportunities that can "promote academic or cognitive development" (p. 158).

As Kuh, Kinzie, Schuh, and Whitt (2005) explained, "Good things go together. Once students engage in a meaningful way with something that excites them, doors to other educationally purposeful activities often open up, and students discover that being in college is the single best place for them to be" (p. 269). Scholars recommend that institutions find ways to bring students and faculty together to create meaningful experiences that could lead to learning (Arnold & Kuh, 1999; Kuh, Palmer, & Kish, 2003; Skipper & Argo, 2003). Terenzini and Pascarella (1980) explained that "faculty members are traditionally viewed as role models who, intentionally or not, serve to acculturate students into the world of ideas" (p. 521).

Researchers consistently demonstrate that when faculty do have time to interact with students outside of the classroom, the interactions can be very powerful (Astin, 1993; Chickering & Gamson, 1987; Clark, Walker, & Keith, 2002; Cotten & Wilson, 2006; Feldman & Newcomb, 1969; Kuh, 1995; Light, 2001; Pascarella & Terenzini, 1998, 2005; Pascarella, Terenzini, & Wolfe, 1986; Terenzini & Pascarella, 1980; Terenzini, Theophilides, & Lorang, 1984a; Tinto, 1975), especially those that are friendly, personal, and address a variety of issues important to the student (Endo & Harpel, 1982).

For instance, as a result of their interactions with faculty, students improve academic abilities and skills (Cotten & Wilson, 2006; Endo & Harpel, 1982; Kuh, 1995; Pascarella & Terenzini, 2005); academic performance (Anaya & Cole, 2001; Cotten & Wilson, 2006; Kuh, 1996; Romanski, 1987); self-confidence (Cotten & Wilson, 2006; Kuh, 1995; Plecha, 2002; Sax, Bryant, & Harper, 2005); persistence in college (Kuh, 1995; Pascarella & Terenzini, 1979b; Pascarella & Terenzini, 1980b; Romanski, 1987; Sax, et al., 2005; Tinto, 1975); cognitive and emotional development (Halawah, 2006; Milem & Berger, 1997; Pascarella, Duby, Terenzini, & Iverson, 1983; Pascarella & Terenzini, 1980b); career and personal skills (Endo & Harpel, 1982; Pascarella & Terenzini, 2005; Sax, et al., 2005; Strayhorn, 2008); inclination for attending graduate school (Pascarella & Terenzini, 2005; Romanski, 1987; Sax, et al., 2005); attitudes and interests (Thompson, 2001); and overall satisfaction with the college experience (Astin, 1993; Cotten & Wilson, 2006; Endo & Harpel, 1982; Kuh & Hu, 2001; Lamport, 1993; Pascarella & Terenzini, 2005; Romanski, 1987).

Researchers also have demonstrated that informal interactions with faculty have a greater impact on students than formal interactions because they are outside of the normal contextual environment (Endo & Harpel, 1982). While college students report minimal interactions with faculty (Anaya & Cole, 2001; Cotten & Wilson, 2006; Fusani, 1994; Hagedorn, Maxwell, Rodriguez, Hocevar, & Fillpot, 2000; Jaasma & Koper, 1999; Lewallen, 1995; Nadler & Nadler, 2001; Snow, 1973; Wilson, Woods, & Gaff, 1974), the ones that do occur are typically associated with a student's major (Alderman, 2008; Iverson, Pascarella, & Terenzini, 1984) or "some special group or activity that brings them into directed and intense one-on-one contact with faculty outside the classroom" (Cotten & Wilson, 2006, p. 498).

One possible solution for bringing students and faculty together outside the classroom is by utilizing existing student organizations as a venue. Many academic departments and colleges have student organizations designed to further the knowledge and skill acquisition of a specific discipline or set of disciplines. Students who participate in these organizations have opportunities to attend lectures; discuss various career paths; understand what is required to be successful in their disciplines; and network with peers and others who share their academic and career interests (Holzweiss, Rahn, & Wickline, 2008).

Students agree that "having faculty members involved in student organizations and clubs" would be a good way of encouraging the two parties to interact (Alderman, 2008, p. 75) while some report that they are already interacting with faculty in the context of student organizations (Lohr, 2004). A few researchers (Bean & Kuh, 1984; Pike, 1999; Skipper & Argo, 2003) have found that students who participate in campus organizations are more likely to interact with faculty than students who do not participate in organizations. In addition, the interactions usually center on goals for the future such as career plans and graduate school (Alderman, 2008). These are the same topics that are covered in academic student organizations so there may be a natural connection that can be made between faculty and students who are involved in these groups.

In general, involvement in organizations has been shown to contribute a variety of positive outcomes for students such as an increase in skill development for critical thinking (Hernandez, Hogan, Hathaway, & Lovell, 1999; Pascarella & Terenzini, 2005), public speaking (Astin, 1993), leadership skills (Astin, 1993), and interpersonal communication (Astin, 1993; Hernandez, et al., 1999; Moore, Lovell, McGann, & Wyrick, 1998); an increase in career-related skills (Pascarella & Terenzini, 2005), psychosocial development (Foubert & Grainger, 2006; Stanford, 1992), intellectual development (Graham & Gisi, 2000), and self-confidence (Huang & Chang, 2004); a sense of community among students (Kuh, et al., 2005); overall satisfaction with college (Astin, 1993); an increase in college persistence (Astin, 1999; Pascarella & Terenzini, 2005; Tinto, 1975); and an increase in the likelihood that students will graduate (Kuh, et al., 1991) and seek additional educational opportunities (Moore, et al., 1998).

Statement of the Problem

While interactions with faculty and involvement in campus organizations are separately recognized as contributing to student development and learning, very few studies have examined whether or not combining the two activities has an impact (Kuh & Hu, 2001; Volkwein, King, & Terenzini, 1986). And no existing literature situates student-faculty interactions in the context of academic student organizations. This descriptive study will determine whether or not student-faculty interactions are occurring in the context of academic student organizations as well as identify the quality and quantity of such interactions and what factors are involved with meaningful interactions. It is hypothesized that when specific student-faculty interactions are explored within academic student organizations, the findings will reveal that there are both quantity and quality interactions occurring.

By situating faculty in their own academic disciplines with students who are interested in the discipline, both parties might be motivated to interact more and therefore increase the quantity and quality of interactions that can lead to student learning. The overall result could be that students better understand their chosen disciplines; students would be motivated to pursue graduate degrees; the reputation of the faculty would increase because of the time dedicated to helping students learn their disciplines in and out of the classroom; companies would want to hire graduates who exhibited a greater understanding and level of experience in their disciplines; and the academic departments and institution would gain a reputation for providing a quality education that equates to graduates finding good jobs. This cycle could prove, at least in one way, that a college education has value to those who share the costs for it.

Significance of the Study

This study will improve upon previous research in four main ways. First, this study examines the context of one type of student organization rather than treating involvement as a universal set of outcomes. Second, this study defines new types of student-faculty interactions occurring on modern college campuses that have been overlooked in many existing investigations. Third, this study compares a small number of institutions in detail rather than combining a large number of institutions from a national database on general outcomes. And fourth, this study creates a new way of defining the quality and quantity of student-faculty interactions.

The first area of improvement over previous research is isolation of one type of student organization. Very few studies have examined interactions between faculty and students in campus organizations. While these studies found a minimal quantity of interactions and limited impact on student learning and development (Kuh & Hu, 2001; Volkwein, et al., 1986), the researchers also failed to consider an important detail that could have impacted the findings – the specific context of the student organizations. Instead, they, along with most existing research conducted on student organizations, treat involvement as a broad activity that equates to the same outcomes for all students in all organizations; an observation also noted by Pascarella and Terenzini (1991).

Modern institutions of higher education have a wide range of student organizations with differing missions and goals. According to Chickering and Reisser's Seven Vectors theory of development (1993), students develop a sense of purpose that causes them to select activities that will benefit them in some way such as making friends, participating in community service, or learning about a career. Some scholars (Beeny, 2003; Gellin, 2003) suggest that research needs to individually examine the variety of organizations available because students select them for diverse reasons.

Academic student organizations, out of all of the other types of organizations available to students, have a direct relationship with career development and scholarly practices. Students are motivated to join these organizations in order to experience the benefits they offer. A study using data from the National Survey on Student

10

Engagement (NSSE) (Sarraf, Johnson, Davis, & Ahren, 2008, March 31) explored opinions of the benefits and detriments of being involved in organizations. The subjects were students involved in different types of student organizations (e.g., academic, service, religious, governance, Greek, housing) across the country. The researchers found that students involved exclusively in academic student organizations, compared to students involved exclusively in the other types organizations, were the least likely to agree that organizations "are for pure enjoyment" and were the most likely to agree that organizations "enhance my academic work" and "will enhance my resume and make me look more attractive to future employers."

In addition, Holzweiss, Rahn, and Wickline (2008) demonstrated that students who chose to join academic organizations (i.e., those associated with an academic department or college) did so because of a future orientation. These students sought academic organizations to help them learn more about their chosen careers; network with other students pursuing similar careers; and participate in activities that would help them find a job in their field after graduation. They also reported benefiting from the organizations by learning about academic resources; further clarifying concepts learned in class; and having a constant reminder of the future goals they wanted to achieve. It is interesting to note that the majority of these students had a junior or senior classification. Chickering and Reisser's development theory (1993) explains that students become more focused on career-related goals as they develop a purpose for their lives so it would make sense that students become more focused on their professional development the closer they get to graduation (Arminio & Loflin, 2003).

By comparison, students who joined non-academic organizations (i.e., those not associated with an academic department or college) did so because of a present orientation (Holzweiss, et al., 2008). These students sought non-academic organizations to help them meet current goals such as making friends, practicing hobbies, and engaging in fun activities that could distract them from their academic responsibilities. The classifications for these students were equally proportioned among all four groups – freshmen, sophomores, juniors, and seniors – lending credibility to the idea that academic organizations may attract students who are getting closer to graduation.

The present study will build upon previous research that has explored academic student organizations and take it one step further to discover what kinds of interactions students in those organizations have with faculty members. One researcher (Alderman, 2008) found that student-faculty interactions were occurring in student organizations and that most of the organizations that provided these interactions were related to academic disciplines. If there are interactions occurring between students and faculty in campus organizations, the most likely place to find them would be within those focused on an academic purpose.

The second way this study will improve on previous research is by clarifying what kind of interactions now may be occurring between students and faculty on a modern college campus. Most of the existing research uses generic statements such as how often students "met informally with faculty to help resolve a personal problem" to define the types of interactions that occur on any college campus. These statements were created 30 years ago when researchers first decided it was a topic worthy of consideration (Pascarella, et al., 1983; Pascarella & Terenzini, 1979a, 1979b, 1981; Terenzini & Pascarella, 1980). At that time in the history of higher education, campuses were smaller and students and faculty naturally may have had more opportunities to interact with each other outside of the classroom. Services such as student counseling centers were not as readily available as they are today and students at that time may have felt that faculty were the only ones they could turn to for assistance.

On modern campuses, however, faculty members are not as enmeshed in the general collegiate environment as they once were due to increasing demands to conduct research and perform other teaching and administrative responsibilities. In addition, some campuses have experienced tremendous growth in student and faculty populations making it very difficult for the parties to even be acquainted with one another let alone discuss "personal problems." As the campuses have grown, so have the abundance of services available to students to address many of the needs previously tended to by faculty.

The type of interactions created for the very first investigation of how students and faculty make connections probably would have been applicable to any collegiate environment of that time. However, subsequent studies merely used the original statements as the ones that were likely to occur on a modern college campus rather than considering how colleges have changed. For example, Einarson and Clarkberg (2004) found that at least some faculty on modern campuses interacted with students at athletic events, arts and cultural events, social functions, field trips, presentations and workshops, and student organization meetings. None of these interactions were considered relevant for the first studies conducted on student-faculty interactions and therefore are not included in subsequent studies even though they now may be common methods of bringing students and faculty together.

The third way this study will improve on previous research is by focusing on four institutions and examining them in detail based on size. Some previous studies rely on data from national surveys such as the National Survey on Student Engagement (NSSE), College Student Experiences Questionnaire (CSEQ), or the Cooperative Institutional Research Program (CIRP) (e.g., Alderman, 2008; Anaya & Cole, 2001; Iverson, Pascarella, & Terenzini, 1984; Kuh, 2003; Lewallen, 1995; Terenzini, et al., 1995; etc.) to help identify the impact of student-faculty interactions on student learning and development. While these studies can be very helpful in defining what issues to examine further, the data cannot provide much more than a benchmark. By their very nature, national surveys must remain consistent and applicable to a wide audience. Questions must be asked the same way each year in order to provide comparisons over time. Because of the different contextual environments of each participating campus, students may not understand or relate to the research questions. This could result in responses that cannot be interpreted easily or applied broadly.

One investigation provides a good example of how students fail to understand some questions asked of them because of a lack of context. Cotten and Wilson (2006) asked students participating in a focus group to describe all of the interactions they had with faculty members. Even though these students participated in a residential learning community that had significant and sustained interactions with faculty as part of the community activities, the students neglected to mention these connections until they were asked specifically to describe them. They initially focused on interactions occurring within the context of the classroom such as talking briefly with a faculty member before or after class or taking advantage of office hours. The findings of the study demonstrate that students may only be thinking of their direct academic experiences when they are asked about generic interactions with faculty outside of the classroom. It is necessary to ask about a variety of possible interactions in the context of specific environments in which they would occur (e.g., "tell me about your interactions with faculty in your residential community") in order to gauge the true nature of these relationships.

These contextually specific questions do not lend themselves well to national surveys since they would not be applicable to every campus and would take up needed space for questions that may be relevant for a wider audience. Research on student-faculty interactions may be more effective if it isolated one specific contextual environment (e.g., academic student organizations), asked about the likely interactions occurring in that environment, then compared responses from students attending different institutions.

The final improvement this study will make over previous research is to introduce a new definition of "meaningful" interactions between faculty and students. Previous studies have focused on either the quantity of interaction or the quality of interactions, or have used them as two separate factors in the same study. There appears to be an ongoing debate regarding which one results in a greater impact on students, with

15

one argument stating that only quality interactions can result in student learning (Clark, et al., 2002; Pascarella, et al., 1983; Pascarella & Terenzini, 2005; Strauss & Volkwein, 2004; Terenzini & Pascarella, 1980; Terenzini, et al., 1984a; Volkwein, et al., 1986) while the other argument explains that frequent interactions can produce learning benefits even if some of those interactions would not meet a high quality standard (Cotten & Wilson, 2006; Endo & Harpel, 1982; Kuh & Hu, 2001; Kuh, et al., 2005; Pascarella & Terenzini, 1975, 1979b; Terenzini & Pascarella, 1980). One researcher noted that "there may be no magic formula as to the frequency and type of facultystudent out-of-classroom interaction" (Alderman, 2008, p. 94). Other researchers (Kuh & Hu, 2001; Sax, et al., 2005) acknowledge that both quantity and quality of the interactions are important to student learning. This study will be the first to combine the previously separate measures of quantity and quality into a composite measurement and address the possibility that there is a direct relationship between the two.

The purpose of this study is to describe what kind of student-faculty interactions exist within academic student organizations and other college activities, document the quality and quantity of such interactions, and create a detailed plan for further research. If the anticipated findings are realized and there are, in fact, meaningful interactions occurring between students and faculty within academic student organizations that are not necessarily available to students through other college activities, there could be valuable information on which to base future policies and student development activities.

Theoretical Framework

To determine how student-faculty interactions might manifest themselves in academic student organizations, it is helpful to use Astin's theory of involvement to understand what occurs when energy is exerted toward meaningful experiences. The theory begins with the simple premise that students will learn if they devote enough effort to their chosen activities. As explained by Astin (1984), "a highly involved student is one who, for example, devotes considerable energy to studying, spends much time on campus, participates actively in student organizations, and interacts frequently with faculty members and other students" (p. 298).

While Astin acknowledges that motivation for learning is an important factor to consider, of greater importance is what kind of behavior the student exhibits that translates into learning. Even if a student has low motivation to meet with a faculty member outside of class, the plain fact that the meeting occurs will likely provide some additional information to the student to assist him or her in the learning process.

There are additional factors that must be understood in order for the theory to be applied successfully. Astin (1984) clarifies his theory by offering five postulates. The first postulate states that involvement occurs when energy is invested in any kind of object or activity. The second postulate explains that involvement is on a continuum and the amount of energy devoted to the objects or activities will vary by individuals and by situations. The third postulate describes involvement as having "quantitative and qualitative features" (p. 298), meaning that factors exist that can have numerical features such as how many times an activity occurs, as well as descriptive features such as how meaningful the activity was for the parties involved. The fourth postulate builds on the third one by stating that learning and development occurring within an experience is proportional to the quality and quantity of involvement a student exerts for the experience. And finally, the fifth postulate explains that educational policies and practices are only effective if they can increase student involvement.

Taking this study's main focus of student-faculty interactions outside of the classroom, Astin's theory can be used to explain why many scholars have consistently found important outcomes in these interactions. When students and faculty interact in a more informal environment, regardless of topic, both parties are actively engaged in the exchange. This is compared to typical classroom environments in which the faculty member dominates the exchange through lecture methods and students are primarily passive learners. In the informal interactions, it is likely that the students will offer increased communication, thus exerting more energy into the experience. This type of experience has been described as powerful learning (Morrill, 2007) or deep learning (Cleveland-Innes & Emes, 2005) and stems from the exchange between the student and faculty member and the relationship that is formed between them.

These interactions involve both quantitative and qualitative features. For example, a student's interaction with faculty can be both a factor of time (how often and what length) as well as a factor of quality (the meaningfulness of the interaction). Considering both avenues of involvement is important to understanding the learning experience. If a student interacts with a faculty member after class in order to clarify an upcoming assignment, the encounter may be brief with little meaning outside of the accomplishment of answering a quick question. However, this same student may decide to visit the faculty member during office hours later in the week to discuss a specific classroom topic further. More time and energy would be devoted to this encounter and it would result in additional learning for the student.

Addressing the final postulate, many institutions of higher education place considerable demands on faculty members in terms of teaching and research. These practices do not necessarily increase student involvement and may, in fact, take faculty further away from possible involvement with students. Adjusting these practices and encouraging faculty to spend more time with students could increase the opportunities for students to engage with faculty outside of the classroom. According to Pascarella and Terenzini (2005), Astin's theory highlights the need for an institutional environment that offers students meaningful opportunities to become involved in educational activities.

Research Questions

Because this study is the first to explore whether or not student-faculty interactions are occurring in academic student organizations, the research questions are designed to produce descriptive results. The purposes of this study and the related research questions are as follows:

Research Purpose #1: Determine what type of student-faculty interactions are occurring within academic student organizations compared to those that occur in other settings.

RQ1: What types of student-faculty interactions exist within academic student organizations and how do they compare to those that occur in other settings?

Research Purpose #2: Determine the quantity and quality of the student-faculty interactions occurring within academic student organizations compared to those that occur in other settings.

RQ2: What is the quantity of student-faculty interactions within academic student organizations and how does it compare to those that occur in other settings?

RQ3: What is the quality of student-faculty interactions within academic student organizations and how does it compare to those that occur in other settings?

RQ4: What is the combined quantity and quality of student-faculty interactions within academic student organizations and how does it compare to those that occur in other settings?

RQ5: What are the differences between classifications for the quantity and quality of student-faculty interactions within academic student organizations and how do they compare to those that occur in other settings?

Research Purpose #3: Determine the relationship between the quality and quantity of student-faculty interactions and institutional size, as well as the relationship between the quality/quantity of student-faculty interactions and

student organization role, as they occur within academic student organizations and other settings.

RQ6: Does the quantity and quality of student-faculty interactions in academic student organizations differ by institutional size and how do they compare to those that occur in other settings?

RQ7: Does the quantity and quality of student-faculty interactions in academic student organizations differ by student role (i.e., member versus leader) and how do they compare to those that occur in other settings?

Operational Definitions

For the purpose of this research, the following operational definitions will be used:

Student-Faculty Interactions (SFIs). Any face-to-face contact between an undergraduate student and a faculty member outside of the formal classroom environment. The interactions could be for any length of time, for any purpose, and occur on or off the campus. Hereinafter referred to as "SFIs."

Academic Student Organization (ASOs). A group, officially approved by its institution, which contains only student members, has a mission that focuses on a particular academic discipline or set of disciplines, and belongs to an academic department or college. Hereinafter referred to as "ASOs."

Institutional Size: The institutional descriptions for four-year colleges and universities were taken from The Carnegie Foundation for the Advancement of Teaching (Carnegie

Foundation for the Advancement of Teaching, 2009). The four institutions selected for this study, which will be combined by size for analysis, were:

- *Small four-year, primarily nonresidential (SNR).* This institutional type has a total enrollment of 1,000 to 2,999 students. Less than 25 percent of students live on campus.
- *Small four-year, primarily residental (SR).* This institutional type has a total enrollment of 1,000 to 2,999 students. More than 25 percent of students live on campus.
- *Large, four-year, primarily nonresidential (LNR).* This institutional type has a total enrollment of more than 10,000 students. Less than 25 percent of students live on campus.
- *Large, four-year, primarily residential (LR).* This institutional type has a total enrollment of more than 10,000 students. More than 25 percent of students live on campus.

Leader. Any student who holds an official position within his/her student organization (e.g., Committee Chair, President). "Leaders are those who accept and engage in consistent and deliberate operation of the organization. They expend considerable psychological and behavioral energy on behalf of the organization" (Arminio & Loflin, 2003, p. 38).

Member. Any student who does not hold an official position within his/her student organization. "The member level of involvement suggests consistent attendance as well as accepting some minimal responsibility with the sponsoring organization that might

include any of the following: paying dues, voting, participating in discussion regarding group functions, or identifying with the organization" (Arminio & Loflin, 2003, p. 38). *Quality*. The length of SFIs to include over 60 minutes, between 31 and 60 minutes, between 16 and 30 minutes, between 10 and 15 minutes, less than 10 minutes, or has not occurred (Nadler & Nadler, 2001; Terenzini, et al., 1984a; Volkwein, et al., 1986). For this study, low quality will be defined as SFIs less than 10 minutes in length and high quality will be defined as SFIs lasting 10 minutes or longer.

Quantity. The frequency of SFIs to include 5 or more times, 4 times, 3 times, 2 times, 1 time, or has not occurred (Pascarella & Terenzini, 1979a; Terenzini & Pascarella, 1980; Terenzini, et al., 1984a; Terenzini, Theophilides, & Lorang, 1984b; Volkwein, et al., 1986). For this study, low quantity will be defined as occurring once or twice since starting college while high quantity will be defined as occurring three or more times since starting college.

Organization of the Dissertation

This dissertation contains five chapters along with references and appendices. Chapter I provides an introduction of the problem, the significance of the study, the theoretical framework being used for the study, the research questions, and operational definitions. Chapter II examines the existing literature for the history of student-faculty interactions, the importance of student-faculty interactions, the history of student organizations, and the importance of student organizational involvement. Chapter III describes the methodology used for the study including the population, instrument design, data collection, and data analysis. Chapter IV provides an in-depth analysis of the results obtained from the survey. Chapter V summarizes the findings and conclusions of the study and provides recommendations for practice and future research.

CHAPTER II REVIEW OF THE LITERATURE

This chapter provides a review of the relevant literature concerning studentfaculty interaction as well as student organization involvement. The literature review is organized into five sections. The first section describes the history of student-faculty interactions in American higher education while the second section discusses the importance of such interactions and what they contribute to student learning and development. The third section describes the history of student organization involvement on college campuses and the fourth section illustrates the importance of this involvement and its impact on student learning and development. The fifth section summarizes the important findings of the existing literature and highlights what the current study will contribute to the literature base.

History of Student-Faculty Interactions

When American higher education began in the 1600s, it used a structure, based on English and Scottish practices, which emphasized classical subjects such as ancient languages and religious philosophy (Rudolph, 1990) as well as promoted "a system of almost monastic common life and discipline" (Sheldon, 1901, p. 81). A lack of adequate financing caused these early colleges to implement a standard curriculum and teaching methods, "which in turn led to the adoption of the recitation method of instruction" that was used in all classrooms (Sheldon, 1901, p. 85). Faculty were entrusted with turning students, mostly young men who had become a burden to their wealthy parents, into public leaders or respectable clergymen (Cohen, 1998). Prayers were required daily and students often had to summarize weekend sermons as a classroom exercise (Sheldon, 1901). In addition, students were limited as to what they could do when they were not in class. According to Sheldon (1901), "A student might not lie down on his own bed in daytime nor spend his own money without first securing the consent of authorities. He was strictly prohibited from leaving his own room except at certain specified hours" (p. 88).

Faculty mostly lived apart from students on the residential campuses and taught only a few specific subjects. They also viewed their teaching positions as transitional roles until they could move on to full-time clergy life or become statesmen (Cohen, 1998). Faculty members assumed a parental role with students and were "duty bound to capture and punish all rebels against the severe college discipline of the time" (Sheldon, 1901, p. 95). Students reacted by demonstrating disdain for any positive interactions their peers tried to have with faculty. As Sheldon (1901) explains, "A class looked with contempt on any of its members who should enter a recitation room before the ringing of the bell, or remain after the close of a recitation to ask a question" (p. 96).

The main academic interaction students had was with their tutors, who were typically recent graduates waiting to train for clergy life. Tutors were required to live with the students, could not marry, and taught most of the classes. They also implemented most of the disciplinary practices but were usually inexperienced at

26

supervising young men, causing the relationships with students to be mostly adversarial (Sheldon, 1901).

As the newly formed nation expanded into the west and railroads established connections between settlements, new information flowed between geographic regions and helped stimulate interest in modern issues such as industrial production, trade, and scientific developments (Cohen, 1998). However, the college curriculum remained focused on religious and classical topics. Frustrated students began rebelling against their strict educational environment and decided to educate themselves by creating literary societies (Sheldon, 1901). These societies centered upon debating current issues and reading materials that were never allowed in their classrooms. The societies also focused on collecting written works that were not available in their campuses libraries. Their resulting society libraries "in every case…were superior to the college libraries" (Rudolph, 1990, p. 143). For awhile, faculty and administrators tried to contain the students' newfound freedom of thought and pull them back on the path of religious obedience and controlled learning environments. However, the faculty role also was changing and they would soon join the students in pursuit of intellectual enlightenment.

As the number of campuses grew to accommodate the expanding population, the demand for faculty increased. Men began to see faculty positions as offering a career rather than merely something to do between jobs (Cohen, 1998). Yet there were not enough trained men available to take these positions. In order to create a qualified faculty, tutors and recent college graduates were sent to Germany to obtain an advanced education and were then expected to return to American campuses to assume faculty

27

positions. They came back with new ideas on pursuing knowledge through research and intellectual inquiry. These ideas began to take root on college campuses as more faculty looked to the German model as the way to pursue learning and develop new instructional techniques. This approach was cemented in American higher education with the founding of Johns Hopkins University in 1876 (Cohen, 1998). This institution increased the reputation of faculty by treating them as authorities of information. Their reputation grew further through the practice of only allowing well-trained students to work on faculty research projects and conduct advanced studies (Cohen, 1998). It was during this period of time when students began looking to faculty for knowledge and seeking their guidance for intellectual development.

With the success of Johns Hopkins University, more colleges began seeing their faculty as a point of prestige and competed for the best ones. Faculty reputations grew as their research solved modern societal problems. In addition, wealthy patrons, trying to avoid paying taxes, developed philanthropic foundations to support the research activities of the faculty (Cohen, 1998). With additional funding and a growing recognition of their research, faculty helped form national groups to set the standards of inquiry in specific disciplines. In turn, the faculty began focusing more on their scholarly fields rather than the individual campuses on which they worked, further separating them from students. It was during this time period that the first student organizations, the literary societies, decreased in importance because the faculty were now addressing all of the students' intellectual needs and there was no reason to duplicate those efforts outside of class (Rudolph, 1990).

As a result of the growth of faculty, the administrative ranks also grew. Students spent more time out of class than they did in it and staff members were needed to provide supervision in the residential areas of campus, distribute needed discipline, and coordinate the increasing number of student activities such as athletics, debate societies, and fraternities (Cohen, 1998). These staff members assumed the original roles of faculty as the managers of student behavior and the leaders of the campus environment while the faculty moved further into their new roles as authorities of intellectual development who were not to be bothered with the social lives of students.

Importance of Student-Faculty Interactions

Since the initial rise of the faculty, American higher education has grown dramatically. The amount and variety of institutions has increased in order to provide for all the training and educational needs of society. And a wide-ranging curriculum has emerged that covers a diversity of academic disciplines and helps train students for jobs after graduation. The growth of the national economy depends on being able to find qualified workers to fill jobs so it is essential to help students make the most of the learning while they are in college.

Astin's theory of involvement explains that college students can direct some of their own learning if they exert the time and energy to developing opportunities for learning (Astin, 1985). For instance, when students focus on meaningful interactions with the wide variety of people available on a college campus such as their peers, staff members, and faculty, their potential for learning is increased (Pascarella & Terenzini, 2005). However, some relationships are not as powerful as others when it comes to the learning process.

When students interact with staff members on a college campus, the staff often have a variety of roles ranging from supportive resources to administrative services (Love, 1995). These roles, as well as the reasons behind why a student is interacting with staff, combine to create a potential learning experience. For instance, staff members serving in mainly administrative roles may be approached by students who must turn in required documents. Other than educating students on the processes and procedures of college, these staff members may not have direct responsibilities that impact significant learning for students.

However, staff members serving in more supportive roles may have opportunities to form deeper relationships with students and therefore contribute something more valuable such as providing a social environment that helps students feel welcome and encourages them to stay in college (Berger & Braxton, 1998; Cheng, 2004; Kuh, et al., 2005; Love, 1995; Pascarella & Terenzini, 2005). These staff members also can provide resources such as counseling and mentoring programs that fulfill students' needs outside of class, thereby contributing to their overall college experience and pursuit of academic achievement (Kuh, et al., 2005). Outside of contributing to a positive campus environment and one study that demonstrated a positive impact on knowledge acquisition (Volkwein, et al., 1986), little is known about the impact of staff on learning (Love, 1995). In contrast, there is considerable evidence that student interactions with their peers and faculty have a tremendous impact on learning outcomes deemed important by higher education (Astin, 1993; Kuh, et al., 2005; Pascarella & Terenzini, 2005). For instance, interactions with peers are influential because of the amount of time students spend together both in and out of the classroom. When students are interacting with their peers in a variety of informal situations, they can gain experience with differing perspectives and beliefs as well as support each other in the pursuit of learning.

Faculty interactions, while not as frequent as peer interactions, can be just as powerful because of their impact on the concept of "deep" learning. Cleveland-Innes and Emes (2005) explained that there were three approaches that students use to engage with their learning environment—deep, surface, and achievement learning. With deep learning, there is intrinsic motivation to learn and students actively search for knowledge throughout their environment, such as being curious about a specific topic and searching for information to satisfy that curiosity. Surface learning is extrinsically motivated and occurs when students use a minimal amount of effort to accomplish a required task. An example of this is when students meet only the minimum requirements for an assignment in order to finish it and move on to other activities. While they do review the material, they are not working with it enough to fully understand it and apply it. Achievement learning is based on students' motivation to reach a loosely related goal such as a good grade. The learning itself is not the motive; rather, students are seeking the result of that learning which can benefit them in other ways such as getting into medical school. They

31

may not care about the material they learned in their English class but they do care about the grade they achieved and how it relates to their other grades.

The authors (Cleveland-Innes & Emes, 2005) explain that the approach students take for learning is both a process and an outcome. They found that deep learning and achievement learning were most often associated with SFIs and that as those interactions increased, the students' deep learning increased while achievement learning decreased. In other words, the students may initiate interactions with faculty in order to accomplish a specific achievement goal such as understanding what they have to do to get a good grade in a class. However, the more students interact with faculty, the more they begin to develop a natural curiosity about the material and want to learn more in order to satisfy that curiosity. This conclusion is supported by other researchers (Astin, 1993; Kuh, 1995; Kuh, et al., 2005; Pascarella & Terenzini, 2005; Thompson, 2001) and places importance on increasing SFIs outside of the classroom environment.

The exchange of knowledge that occurs when students and faculty interact within a classroom environment is easy to understand. When students participate in the formal learning of a classroom, they engage in the planned distribution of knowledge. Topics are defined by faculty and formal assessment measures of student learning, such as tests and written assignments, are implemented in order to evaluate what students have gained from their classes. However, Boyer (1987) suggests that "the most important teaching may go on outside the classroom" (p. 157) so it is also important to understand how informal learning occurs. According to Marsick and Watkins (2001), informal learning is intentional, loosely structured, and directed by the learner. It can generate from a variety of situations such as attempting a task through trial and error, or emerge from relationships such as having a mentor to guide personal decisions. In these informal situations, learners may not even be aware of the gains they have made so it is helpful to understand the authors' Informal and Incidental Learning Model so all of the processes that factor into informal learning can be identified.

The model first explains that all informal learning occurs within the specific context of the learner (Marsick & Watkins, 2001). The context includes the immediate environment, the interaction between the learner and people in that environment, and the activities conducted by the learner within the environment. This context influences how learners make meaning of the information they have received. The process of informal learning occurs when new information disrupts the learner's contextual environment. Learners will evaluate the new information through their personal view of the world as they attempt to understand why the information is causing a disruption. They will compare the new event to situations and information they have encountered in the past and determine what additional knowledge has been contributed, thereby resulting in informal learning.

While informal learning can occur anywhere and at anytime, the authors (Marsick & Watkins, 2001) recommend three conditions to increase the likelihood that informal learning will occur. First, the environment needs to be structured for informal learning opportunities such as creating purposeful situations where new ideas may be exchanged. Second, learners should be encouraged to look for those opportunities and explore them. And third, learners should be encouraged to reflect on any new ideas generated during these opportunities in order to gain insight into their impact on learning.

When the model is applied to the context of informal interactions between faculty and students, it is easy to understand how learning occurs. Through their enrollment and participation in college life, students are already in a contextual environment that promotes learning. They attend class and are expected to perform the requirements to successfully earn an academic degree. Most of their knowledge acquisition occurs in the confines of a classroom with the faculty member serving in the role of expert. When students and faculty interact informally outside of the classroom, they are creating a new experience that is outside of their normal contextual environment. This can help trigger the learning process and signal to students that new information is coming to them from people they view as knowledge experts.

However, it is not always easy to define the type of interactions that provide these learning opportunities. Some researchers have focused on SFIs within specific contexts such as living-learning communities (LLC). The early LLC movement required faculty to reside alongside students in campus residences or spend a substantial amount of time within these residential environments such as through teaching classes in the living space or holding office hours there rather than in academic buildings. Yet a major problem with this type of community is that faculty members do not always want to get involved in these time-intensive environments because they are not rewarded for doing so, do not personally enjoy spending that much time with a younger generation, or do not feel they can handle the personal issues of students that can emerge from the increased intimacy between them and students (Golde & Pribbenow, 2000).

Because of these reasons, the modern LLCs are increasingly focused on peer interaction within the academic environment, such as peers taking a number of classes together and attending programs directed toward their academic disciplines. Structured this way, these communities are easier to manage and can still increase the frequency of SFIs (Garrett & Zabriskie, 2003; Inkelas & Weisman, 2003; Kuh, Palmer, & Kish, 2003; Pascarella & Terenzini, 1981; Terenzini, et al., 1996); the quality of SFIs (Pascarella & Terenzini, 1980b); or result in additional benefits such as increased involvement in student organizations (Pike, 1999), improved academic performance (Pascarella & Terenzini, 1981), and persistence in college (Pascarella & Terenzini, 1981). However, the purpose, structure, and availability of LLCs may vary between campuses and may not accomplish the goal of bringing students and faculty together in close and frequent contact.

Another opportunity for meaningful interaction between students and faculty that has been explored is working together on research projects. These partnerships can help students become actively involved in their own learning (Pascarella & Terenzini, 2005) as well as provide faculty with low-cost assistance for their research agendas. In fact, one study concluded that working with faculty on just one research project during college might result in a transformative learning experience for students (Kuh, et al., 2005). According to the authors, "It is hard to imagine a richer educational setting for

35

student-faculty interaction than working side by side with a faculty member on a research project. Students not only observe an expert at work, but they also contribute to that work by applying in-class learning to the research project. And because many such projects extend beyond a single academic term, they provide students and faculty with many opportunities to discuss topics related and unrelated to the research" (p. 214). Working on research projects with faculty members can be a powerful experience for students in a variety of ways such as building skills like interpersonal communication, project management, and leadership (Nikolova Eddins, Williams, Bushek, Porter, & Kineke, 1997); increasing self-awareness and overall satisfaction with college (Nikolova Eddins, et al., 1997); and increasing the likelihood of graduation and pursuit of an advanced degree (Walpole, 2003).

While one study found that some faculty members were interacting with students on research projects a few times a month (Einarson & Clarkberg, 2004), these opportunities may not be available to a variety of students or could come so infrequently that they would never benefit a large segment of the undergraduate population. It is important to identify interactions that do not center exclusively on research projects or residential life so that the diversity of students attending modern college campuses can have an opportunity to pursue interactions with faculty. Researchers have demonstrated that a variety of less intensive SFIs also contribute to students' development including having informal conversations with faculty about grades, assignments, and career plans (Alderman, 2008; Astin, 1993; Cotten & Wilson, 2006; Kuh, et al., 2005); working with faculty on campus committees (Kuh, et al., 2005); being a guest in a professor's home (Astin, 1993); having a faculty mentor (Cotten & Wilson, 2006); participating in a course designed to introduce students to the functions of a college campus (Maisto & Tammi, 1991); and interacting with faculty in student organizations or campus activities such as athletics and conferences (Alderman, 2008).

There are a variety of interactions that can occur between students and faculty but they are often minimal (Einarson & Clarkberg, 2004; Fusani, 1994; Nadler & Nadler, 2001) and even may be non-existent for some students (Cotten & Wilson, 2006). The existing literature explains that anywhere from two-thirds to almost all of undergraduate students experience SFIs (Jaasma & Koper, 1999; Lewallen, 1995; Rosenthal, et al., 2000) and that those interactions usually occur before or after class or during a faculty member's office hours (Cotten & Wilson, 2006; Jaasma & Koper, 1999) and last less than ten minutes (Jaasma & Koper, 1999; Lewallen, 1995). SFIs typically occur when students need help with coursework (Anaya & Cole, 2001; Cotten & Wilson, 2006; Fusani, 1994; Nadler & Nadler, 2001); advice on academic or career issues (Alderman, 2008; Cotten & Wilson, 2006; Fusani, 1994); or have a personal interest in a faculty member's area of expertise (Gaff, 1973). While the majority of the literature found infrequent SFIs, it is important to note that Terenzini and Wright (1987) found that SFIs occurred five to seven times per year for academic issues and up to two times per year for non-academic issues.

Students typically initiate all of the interactions but only if they have a reason to do so and feel comfortable approaching faculty. According to the Cotten and Wilson (2006), a number of factors influence the comfort level felt by students. For instance,

maturity is a key factor in helping students feel comfortable with faculty members. Some researchers demonstrated that junior and senior level students are more likely to have SFIs than freshmen and sophomores (Alderman, 2008; Cotten & Wilson, 2006). In addition, as students progress in their college program, they become more interested in interactions with faculty and what they could gain from faculty in terms of knowledge and career development (Arminio & Loflin, 2003; Cotten & Wilson, 2006) as well as understand more about what faculty are required to do within their positions (Cotten & Wilson, 2006). Terenzini, et al., (1984a) theorized that SFIs may be more important in the later years of an academic career than the earlier years because students have a greater self-awareness and understanding of the academic environment and can gain more from interacting with faculty.

Another factor influencing the comfort level of students is how they perceive faculty receptiveness to interacting with them. According to one study, students are unsure whether faculty want interactions or if they have enough time to do so (Cotten & Wilson, 2006). Students also need some level of recognition between faculty members and themselves in order to feel comfortable in initiating interaction. If they felt faculty members did not know who they were because of large classes, for instance, they might choose not to initiate contact.

Smaller campus environments often have opportunities for more informal interactions (Kezar & Kinzie, 2006; Pike & Kuh, 2005) and may help students feel that faculty are more accessible and open to interactions (Alderman, 2008; Dilley, 1967; Gaff, 1973). On the other hand, larger institutions provide increased opportunities for

other powerful interactions such as participating in learning communities and faculty research projects (Kezar & Kinzie, 2006). Astin (1993) agreed by saying, "In all likelihood, large institutions, in comparison to small ones, offer more diverse social opportunities (social organizations, parties, and so on), more frequent cultural events on campus, a greater number of extracurricular activities, and fewer regulations governing campus life" (p. 284).

Yet, institutional size may not matter as much as how a faculty member teaches and interacts with students in the classroom setting. If faculty members create an intimate classroom environment by sharing personal information or using interactive teaching techniques, students can perceive that faculty are receptive to interactions outside of class and students may be more willing to initiate contact (Baxter Magolda, 1987; Cotten & Wilson, 2006; Wilson, et al., 1974). One study found that faculty members who were identified by students as good teachers had a desire to engage with students on a variety of issues both in and outside of the classroom (Gaff, 1973).

Even when students feel comfortable enough to initiate interaction, they may choose not do so because of other barriers. For instance, once a connection has been established with a faculty member, the student can perceive increased expectations for academic performance (Cotten & Wilson, 2006). In other words, getting to know a faculty member means that students will stand out and their classroom performance will be noticed. Students may avoid interactions with faculty so they do not have to exert extra effort to perform well or run the risk of disappointing their instructors. Students also may need a catalyst to help them interact with faculty such as pre-planned activities that bring both groups together (Anderson & Carta-Falsa, 2002; Cotten & Wilson, 2006).

Understanding the motivations for students to seek interactions with faculty is just one part of the equation. Another challenge comes from the faculty perspective. With the growing reliance on adjunct faculty and increased teaching loads for tenured faculty, there may not be enough time for faculty to engage with students in meaningful ways unless institutions provide the support and appropriate rewards for doing so (Boyer, 1987; Kuh, et al., 2005; O'Meara & Braskamp, 2005). However, one study found that institutional support was available for faculty who wanted to interact with students but some chose not to because of personal preferences or discomfort with interpersonal communication (Einarson & Clarkberg, 2004). While some faculty may choose not to interact with students, it is still important to provide students with what they need to be successful in college and advance their intellectual development. Boyer (1987) states that "faculty and administrators turn their backs on life outside the classroom, where there is so much learning that either enhances or diminishes the quality of the undergraduate experience" (pp. 192-193).

Some researchers have identified characteristics of faculty who may be more inclined to interact with students outside of the classroom and experience their own personal gains from doing so. Einarson and Clarkberg (2004) discovered that faculty who spend a lot of time with research and teaching activities are more likely to also engage in SFIs and will do so even if there is no institutional support or direct rewards. But they may find they receive personal rewards such as meaningful relationships (Gaff, 1973); a sense of accomplishment (Wilson, et al., 1974); an improvement in their teaching techniques (Frankel & Swanson, 2002; Golde & Pribbenow, 2000); a greater understanding of student needs and interests (Wilson, et al., 1974); and greater satisfaction with their job responsibilities (Wilson, et al., 1974).

One way to encourage faculty and students to spend some of their time together is by making it easier for the interactions to happen. Besides research projects and learning communities, another formal activity already exists on college campuses that has the potential to bring faculty and students together on a regular basis and for common interests. Student organizations have a long history of fulfilling the developmental and social needs of students and, over time, involving faculty in their activities.

History of Student Involvement in Organizations

The first historical references to student organizations occur between the fifth and tenth centuries in Europe (Sheldon, 1901). These early organizations were founded for political purposes when students came together to advocate for individual rights that colleges and communities would not automatically grant them. In seventeenth century England, organizations began to form that concentrated more on personal interests rather than political issues. The students, captivated with the leisurely lifestyle of the nobility, developed groups to promote activities such as fox hunting, playing cards, and attending horse races (Sheldon, 1901).

Student organizations shifted back to political issues during the eighteenth century with the formation of debating societies (Sheldon, 1901). These societies varied

in numbers depending on the political topics circulating throughout the communities. Literary societies also existed but were described as less critical than debating societies and focused on reading popular essays and discussing them. Because of the lower status of college students within the local communities, students from debating and literary societies often came together to form unions which would then represent its members in issues important to the local community as well as collect scholarly works for the purpose of enhancing intellectual development (Sheldon, 1901).

When student organizations made their first appearance in the American colonies in 1753 at Yale University, it was in the form of literary societies (Rudolph, 1990). These societies became prominent in colonial colleges because "the athlete had not yet arisen as a college hero, so the orator and writer represented the ideals of the academic youth" (Sheldon, 1901, p. 133). Like the European student unions, the American literary societies focused on activities such as "prepared orations, debates, declamations, and critical papers" (Sheldon, 1901, p. 129); writing comedies and performing them for members (Cutting, 1871); soliciting student works and publishing them in societysponsored literary magazines (Rudolph, 1990); and bringing controversial speakers to campus in order to stimulate discussion (Rudolph, 1990). Given the adversarial nature between students and faculty at that time (Sheldon, 1901), many of the works prepared by society members consisted of caricatures of the faculty along with humorous pieces designed to criticize them (Cutting, 1871). The societies often held competitions and judged all of the member-created works for quality (Sheldon, 1901). The literary societies also followed their European counterparts in the collection of modern literary works. In one example of the impact of these collections, Cutting (1871) describes how, in the 1800s, the Amherst College library contained a single case of books whereas the literary societies collected over 8,100 texts. Colleges often limited their collections to religious and classical literature while students worked to solicit modern pieces from community leaders and alumni (Sheldon, 1901).

It is reported that there were numerous skills that students developed as members of these literary societies such as interpersonal communication, written and oral communication, meeting management, and critical thinking (Sheldon, 1901). As explained by Rudolph (1990),

In a sense, the literary societies and their libraries, the clubs, journals, and organizations which compensated for the neglect of science, English literature, history, music, and art in the curriculum—this vast developing extracurriculum was the student response to the classical course of study. It brought prestige to the life of the mind. It helped to liberate the intellect on the American campus. It was one answer to the Yale Report of 1828, an answer so effective that by the end of the century at Yale itself there would be real concern over which was really more fundamental, which more important, the curriculum or the extracurriculum (p. 144).

Once they graduated, the societies helped form such a strong bond between students that alumni would often come back to the campus to participate in the commencement exercises prepared by the societies (Sheldon, 1901).

As membership in literary societies increased and virtually all students joined,

the leadership of the societies became increasingly political, with smaller cliques

controlling the elections of officers. For that reason, students began forming social

fraternities and secret societies to help gain power on campus (Sheldon, 1901). With the

power struggles occurring between students and the curriculum incorporating modern issues, literary societies vanished and other organizations began fulfilling unmet needs for students. However, the legacy of the literary societies continued through activities such as intercollegiate debating competitions and professional associations created to enhance learning in fields such as medicine, law, and science (Sheldon, 1901).

If literary societies served to fill the gaps of the academic curriculum, then Greek societies helped the students increase their social development by creating strong emotional bonds between the young men. According to Rudolph (1990), Greek societies emerged in 1825 to offer "an escape from the monotony, dreariness, and unpleasantness of the collegiate regimen" (p. 146) as well as diminish the influence of the students who controlled the literary societies (Cutting, 1871). The activities of the Greek societies included practices that were discouraged or banned by the church-run colleges such as drinking, smoking, and gambling. The Greek groups created a strong sense of loyalty among members and contributed to the demise of the literary societies, especially when institutions began focusing more on methods of intellectual development in the classroom and made the literary societies obsolete for the academic environment.

While the Greek organizations fueled the social needs of students, gymnastics clubs addressed their physical conditioning needs. The emergence of the German gymnasium movement gave rise to such clubs in the late 1840s (Rudolph, 1990) and created a network of campus recreational activities that provided physical development as well as interaction that solidified social connections between students (Sheldon,

44

1901). Eventually, the college athlete surpassed the literary scholar as the hero on college campuses (Sheldon, 1901).

With the growing list of student organizations and the increased attention on faculty as knowledge experts, the tension between students and faculty began to diminish (Sheldon, 1901). As faculty began to teach modern topics rather than just the classical ones, students began inviting them to join campus groups and further contribute to the students' intellectual development (Cutting, 1871). Faculty served as honorary members of the organizations, judges for contests, and advisors for events and activities (Cutting, 1871).

Through all of these extracurricular activities, students formed personal interests, developed a variety of skills, and enhanced their knowledge. While the academic curriculum provided theoretical and foundational knowledge, the interactions students had outside of the classroom with their peers and faculty provided the practical knowledge they needed to succeed in the business world. As their skills began to increase, additional organizations emerged to help address students' changing campus needs. Student government associations formed at the end of the nineteenth century (Rudolph, 1990) and it was at this time that faculty roles were seen as purely academic. No longer did faculty assume any responsibility for behavior management. They instead focused more on the growing research movement and their scholarly contributions to academic disciplines. This perpetuated the need for more personally fulfilling activities that allowed students to have a sense of well-being and support while attending college and they increasingly turned to their peers to fulfill these needs.

Importance of Student Organization Involvement

In his book *What Matters in College? Four Critical Years Revisited*, Astin (1993) described the peer group as "the single most potent source of influence on growth and development during the undergraduate years" (p. 398). Researchers have validated this claim by linking participation in student organizations to improvements in a variety of skills and developmental needs as well as fostering a sense of community on the campus and increasing the likelihood that students will graduate (Astin, 1993; Foubert & Grainger, 2006; Graham & Gisi, 2000; Hernandez, et al., 1999; Huang & Chang, 2004; Kuh, et al., 2005; Kuh, et al., 1991; Moore, et al., 1998; Pascarella & Terenzini, 2005; Stanford, 1992; Tinto, 1975).

Being involved in organizations also appeals to employers and increases the likelihood that a student will obtain a job offer after graduation (Pascarella & Terenzini, 2005). One group of researchers studied what job recruiters look for in education, engineering, and business fields in terms of collegiate preparation and experience (Moore, et al., 1998). They found that some recruiters preferred involvement in organizations over academic performance. Specifically, recruiters for education and engineering fields preferred students who had a high level of experience in campus organizations and average grades over students with high grades and average activities experience. Recruiters for the business field, however, did prefer students with high grades and at least average activities experience, indicating that for some fields, academic performance may still be of greater importance but that experience with student organizations stands prominently alongside it. Recruiters are not the only group who recognize the importance of organizational involvement on career success. Kuh et al. (1991) found that alumni attributed at least some of their job success to their involvement in student organizations while attending college. Involvement also impacts alumni behavior by increasing the likelihood that they will join community groups and become civically engaged in their local areas (Johnson, 2004). In addition, involvement increases positive perceptions of the institution and helps build loyalty (Kuh, et al., 1991), which can result in increased donations and support from alumni in later years. This loyalty begins with students joining organizations which connect them to the campus and their peers, while also providing an experiential environment where students can learn and refine practical skills needed for the workplace (Kuh, et al., 2005). As summarized by Pascarella and Terenzini (2005), "failure to capitalize on students' out-of-class experiences risks increasing learning only at the margins" (p. 647).

While many researchers have found benefits to being involved in student organizations (e.g., Astin, 1993; Pascarella & Terenzini, 2005; Terenzini, Pascarella & Bliming, 1996; etc.), they tend to treat all student organizations the same, making the assumption that any involvement equates to the same outcomes for every student in every situation. According to Terenzini, Pascarella, and Bliming (1996), "most researchers have either operationalized the concept as a global variable, or they have made no clear distinction between extracurricular and peer involvement" (p. 152).

Even though there have been a few studies conducted on involvement in specific types of student organizations, the research has been mostly confined to studies of Greek

organizations and intercollegiate athletics (see Astin, 1993; Pascarella & Terenzini, 2005). Beeny (2003) suggested that future research might focus on "examining whether different types of student organizations influence the amount of expressed learning or the skills or competencies students report gaining" (p. 87). Gellin (2003) echoed the suggestion saying that "traditionally, scholars have used the moniker *clubs and organizations* to represent the large number of sponsored activities available on college campuses" but that the broad focus "may be limited" and that a "closer examination of specific clubs and organizations may be warranted" (p. 759).

A few researchers have found that when student organizations focus on academic issues and career development topics, students, especially juniors and seniors, benefit in a variety of ways such as improved career planning (Wessel, Christian, & Hoff, 2003); networking with others who can help them prepare for their career (Holzweiss, et al., 2008); and overall learning (Terenzini, et al., 1996). In fact, Pascarella and Terenzini (2005) explain that student involvement that is not relevant to academic purposes can inhibit intellectual development.

Since student involvement in the general sense has been consistently demonstrated to increase student development and learning, it is necessary to begin delving deeper into specific types of organizations, starting with the ones that align most closely with the academic missions of institutions. In particular, it would be valuable to explore what important activities, such as interactions between students and faculty, are already occurring in academically-focused organizations. Some researchers have found that student involvement can lead to interactions with faculty (Bean & Kuh, 1984; Pike, 1999; Skipper & Argo, 2003) but questions remain regarding what kind of interactions are occurring, in what environments, and how students benefit from them.

Summary

Several conclusions can be drawn from the review of the literature. First, interactions between students and faculty have been occurring since the 1800s and have typically centered on academic and intellectual topics. Second, interactions between students and faculty can have a powerful impact on student learning and development even though they may be brief or occur infrequently. Third, there is a lengthy history of students being involved in campus organizations. While faculty have not been involved in these organizations to a large degree, there has been some documented involvement, especially when the subject matter addresses academic and career topics. And fourth, very few studies have merged the two topics together and explored what kind of SFIs are occurring within campus organizations. The ones that have researched this topic neglected to investigate the impact of the contextual environment.

This study contributes to the existing literature by examining one type of campus organization which promotes academically-oriented subject matter. It is within this type of organization that faculty and students may be more naturally drawn together to share their common interests. By better understanding what kinds of interactions are occurring between students and faculty within these academic organizations, institutions can determine how to structure the collegiate environment so it promotes more of these interactions and increase student learning both in and out of the classroom.

CHAPTER III

METHODOLOGY

This study employed survey methodology to gather data. The purpose of using survey methodology was to provide adequate description of the topic being studied as well as maintain objectivity towards the study participants and potential findings (Smith, 1983). This chapter will describe the paradigm used for the study, the population and sample, the survey instrument, and data collection and analysis processes.

Research Paradigm Used for the Study

Given the research questions posed to resolve this problem, the post-positivist paradigm is the appropriate structure to use as a research model. As explained by Guba (1990), the fundamental belief of the post-positivism paradigm is that human beings cannot fully understand reality. However, interpretations can be made from reality and the "researcher must have some competently gathered evidence...to back up the interpretation that is being given" (Phillips & Burbules, 2000, p. 79). The researcher can be an observer of what is being studied and act in a neutral manner towards participants in the study (Smith, 1983). This is achieved by "carefully selecting a representative sample from a population of interest and then, by means of a statistical inference, calculating the probability that findings based on the sample are characteristic of the population" (Schwandt, 2001, p. 106).

The purpose of this study was to discover what is happening between students and faculty in previously unexamined areas of the collegiate environment. To better understand student learning, it was important to conduct the study in such a way that the results could provide a thorough description of what is occurring and how future research could be constructed to fully assess the topic. Merriam (1991) explained that the post-positivist paradigm utilizes tools such as "scales, tests, surveys, and computers to collect data" and that this allows the researcher to maintain "as much distance from the researched as possible so as to remain objective" (p. 45). Therefore, the research design was structured to obtain this objectivity by using the recommended procedures, subject selection, and instrumentation. Along with obtaining objectivity, the study also was designed to be descriptive in nature. According to Isaac and Michael (1990), descriptive research "does not necessarily seek or explain relationships, test hypotheses, make predictions, or get at meanings and implications" (p. 46). This study sought to document, for the first time, a contextual environment and the interactions that are occurring within it.

Population and Sample

As outlined by the post-positivist paradigm, the study utilized survey methodology to obtain information regarding student-faculty interactions in academic student organizations. To combat the non-random nature of the sample, different types of four-year institutions were pre-selected as target populations to ensure students from different collegiate environments were included. The sample came from four institutions in the state of Texas which covered different categories of the Carnegie classifications. Those institutions included the University of St. Thomas in Houston (small four-year, primarily nonresidential), McMurry University in Abilene (small fouryear, primarily residential), Texas State University in San Marcos (large four-year, primarily nonresidential), and Texas A&M University in College Station (large four-year, primarily residential).

In addition to selecting different institutional types, a pre-selection process for the ASOs was conducted by viewing the student organization list at each institution. Only organizations with similar missions and purposes on each campus were selected for the study. The resulting organizations represented the academic disciplines of chemistry, psychology, English, health, and education. Participants in the five ASOs at each institution were combined into one institutional group in order to overcome any variance that occurred due to active membership numbers.

The following mission statements were obtained from organizational websites to highlight the similarities between the organizations on different campuses. *Chemistry*

- *American Chemical Society*: "The purpose of the ACS Student Affiliate Chapter is to promote interest in the chemical sciences. We are designed to provide a place for undergraduate chemistry majors (and anyone else who is interested in chemistry) to meet and get to know one another. The organization also brings chemistry to the campus and community, performs community service and attends the National ACS Conference" (ACS Profile, 2009).
- *Chemistry Club*: "The Chemistry Club is affiliated with the Chemistry & Biochemistry Department" and has the goal of providing "resources to further the needs of students pursuing a degree in chemistry or enrolled in chemistry

courses, as well as to create a long-term network among students" (About Chemistry Club, n.d.).

Education

- Association of Texas Professional Educators (ATPE): "With the strength of more than 112,000 educators as its foundation, ATPE is a constant champion for public education. ATPE believes it takes everyone working together to make Texas public schools successful. This collaborative philosophy is evident in our diverse membership and is what distinguishes ATPE from other organizations. ATPE members are committed to making positive contributions to the education profession and to the lives of the students" (Association of Texas Professional Educators, 2009).
- *Kappa Delta Pi*: Kappa Delta Pi is an International Honor Society in Education and "was established to foster excellence in education and promote fellowship among those dedicated to teaching. The founders chose the name from the Greek words to represent knowledge, duty, and power. Pioneering from its beginning by including women as well as men, Kappa Delta Pi grew form a local chapter to the international organization it is today, comprising 582 chapters and more than 45,000 members" (About KDP, 2008).

Health

 Health Occupations Students of America (HOSA): "HOSA is a national student organization endorsed by the U.S. Department of Education and the Health Science Technology Education Division of ACTE. HOSA's two-fold mission is to promote career opportunities in the health care industry and to enhance the delivery of quality health care to all people. HOSA's goal is to encourage all health occupations instructors and students to join and be actively involved" (What Is HOSA?, n.d.).

- *Phi Epsilon Kappa:* "Phi Epsilon Kappa Fraternity is a national professional fraternity for persons engaged in or pursuing careers in physical education, health, recreation, dance, human performance, exercise science, sports medicine and sports management. Membership is open to persons interested in the purposes of the Fraternity and in providing time and energy for the benefit of these areas" (Phi Epsilon Kappa, 2009).
- Health Information Management Student Association: The purpose of this
 organization is "to promote student engagement in the HIM program and to
 promote the HIM profession." Activities include "social activities, guest
 speakers, community service, and professional involvement" (Health Information
 Management Student Association, n.d.).

Psychology

Psi Chi: "Psi Chi is the National Honor Society in Psychology, founded in 1929 for the purposes of encouraging, stimulating, and maintaining excellence in scholarship, and advancing the science of psychology. Membership is open to graduate and undergraduate men and women who are making the study of psychology one of their major interests, and who meet the minimum qualifications" (About Psi Chi, 2009). *Psychology Club:* "The Psychology Club is open to all students regardless of major, who are interested in becoming involved in an academic as well as a social organization." The organization "is designed to facilitate students' knowledge about the field of psychology and other related possibilities of psychology. Additionally, the club assists students in gaining experience in volunteer work, learning about graduate school, and meeting new friends" (Psychology Club/Psi Chi, n.d.).

English

For English, the organization Sigma Tau Delta was found on all campuses.

 Sigma Tau Delta: "Sigma Tau Delta is the International English Honor Society. Its purpose is to confer distinction upon undergraduates, graduates, and scholars in academia as well as upon professional writers who have recognized accomplishments in linguistic or literary realms of the English language. Sigma Tau Delta affords exceptional students in the field of English opportunities for furthering culture, for formulating ethical principles for developing skills in creative and critical writing, and for fostering a spirit of fellowship" (Sigma Tau Delta profile, 2009).

Instrument

The researcher-designed survey instrument combined a collection of 24 statements regarding student-faculty interactions that were used in previous studies on SFIs (See Appendix A). Participants were asked to view the list of interactions three different times and respond to a different question each time. The first time through the list, participants were asked to consider the interaction and select all applicable options in regard to why the interaction occurred. The purpose of this section was to gather basic information on what SFIs were occurring as a result of participants' involvement in ASOs compared to SFIs stemming from other collegiate experiences.

The second time through the list, participants were asked to select how many times the interaction occurred between them and a faculty member since they started college. The purpose of this section was to obtain the quantity of SFIs as experienced by participants. Previous studies that have examined the quantity of SFIs have used the frequency of such interactions as the basis for judging quantity (Pascarella & Terenzini, 1979a; Terenzini & Pascarella, 1980; Terenzini, et al., 1984a, 1984b; Volkwein, et al., 1986).

The final time through the list, participants were asked to consider the average length of interactions they have had with faculty. The purpose of this section was to obtain the quality of SFIs as experienced by participants. Previous studies have examined the quality of SFIs through the length of such interactions as the basis for judging quality, with interactions over 10 minutes in length being of good quality (Nadler & Nadler, 2001; Terenzini, et al., 1984a; Volkwein, et al., 1986).

In addition to the list of SFIs, participants were asked to provide some demographic information such as gender and classification along with additional information such as their role within the ASO (i.e., member or leader) and whether or not they were members of a learning community on their campus. This information was used to explore differences between demographic groups as well as establish an overview of each institution's participants.

Data Collection

An online survey service, QuestionPro, was obtained in order to design and implement the survey. The service met several criteria important to the implementation of the survey such as secure data collection, management of participants (e.g., tracking who responded, sending reminders, etc.), and access to raw data for the analysis process. Once the initial survey draft was developed, the researcher asked two ASOs at Texas A&M University that were not selected to participate in the actual study to take the survey twice over the course of fourteen days in order to determine test-retest reliability (Isaac & Michael, 1990). A reliability analysis was conducted between the two administrations of the survey and resulted in an overall coefficient alpha of .912. Split half reliability analysis was conducted on all three sets of questions. For the first set of questions, reliability was .766; the second set of questions was .955; and the third set of questions was .988.

The pilot participants also were asked to provide written feedback regarding clarity and validity of the instrument items. As a result of the feedback received, a few minor changes were made to the instrument including the rephrasing of one non-SFI question for clarity purposes as well as correcting a few grammatical issues. Once the changes were made, a final version of the survey was prepared for the formal data collection period. The final survey instrument appears in Appendix B. Pilot testers reported an average time of ten minutes to complete the survey. To recruit participants for the study, the researcher first contacted staff representatives at each pre-selected institution who had responsibilities for campus organizations. The purpose of this communication was to request their assistance in identifying the approximate date when the organizations concluded their recruiting and selection processes for the fall semester as well as identifying, for each ASO, the appropriate student representatives, staff advisors, and all of their contact information. Based on this initial contact, one institution originally selected to be part of the study had to be replaced with another institution. The reason for the replacement was that the original institution did not have the administrative structure to identify current student representatives for their campus ASOs. Since there was no method of identifying and contacting the appropriate organizations, another institution with the same basic characteristics was substituted.

Contact was made with the student representatives to explain the project and ask for their assistance in recruiting members of their organization. Each representative was asked to share information with their organizational members regarding the project, ask for volunteers to take the survey, and collect email addresses of all willing members prior to October 30, 2009. All members of each organization were invited to participate. In a few cases, the student representatives did not respond to the request. When this happened, staff advisors for the ASOs were contacted to assist with recruiting organizational members. Out of the twenty organizations identified for the study, only one organization's staff advisor and student representative never responded. Contact

58

was established with either the student representative or the staff advisor with the remaining nineteen organizations.

A few days before the survey was launched, an email message was sent to each participant reminding them of the impending survey, reiterating the importance of the project and their participation in it, and requesting that they take ten minutes of their time to complete the survey when they received it. The survey was emailed to each participant on November 2, 2009. In an effort to continue recruiting from the organizations that only had a few members volunteer for the study or did not respond to participation requests, the survey link was sent to the student representatives for these organizations with a request to forward the survey link to their members.

To encourage participants to complete the survey, reminder emails were sent every three working days to those who had not yet taken the survey. The survey closed on November 16, 2009, for analysis. As part of the survey procedures and information, participants were notified on the last question of the survey that they could receive a copy of the findings when the study was completed. Emails entered voluntarily on the survey for the purpose of receiving the findings were removed from the database prior to analysis and stored in a secure location for future use.

Data Analysis

Once data collection concluded, the database was prepared for analysis. This included removing responses that did not meet the initial criteria. One unforeseen problem was the extended membership of some of the ASOs in the study. Because these organizations were professional in nature, some of the membership rosters included graduate students, faculty members, and alumni. A total of 45 responses had to be removed from the database because of this issue. Another fifteen responses had to be removed because the participants did not have active status in the pre-selected organizations. And finally, eleven responses had to be removed because of incomplete information.

Several types of analysis were conducted in order to obtain relevant descriptive data for the research objectives and questions. All analysis was conducted using SPSS statistical software. Data analysis began with an examination of response rates and participant demographics by institution and ASO type. This was followed by a simple frequency analysis to address the first research objective of identifying where the SFIs had occurred. In addition, an analysis of the overall SFI frequency was conducted to determine how many participants had experienced any of the interaction in their ASOs, in their other college activities, and in both contexts. These overall interactions were analyzed by institutional size, classification, and organizational status to determine if there were any differences between groups. The comparisons were conducted using a Pearson chi-square analysis. The chi-square analysis was selected because it is a calculation technique that can address small sample sizes in independent groups (Spatz, 2005). Small sample sizes were expected in this study due to the varying nature of the organizational membership at each institution.

To answer the second research objective, several analysis techniques were employed. First, a simple frequency analysis was conducted to determine how often the interactions occurred for participants since they started college. Means and standard

60

deviations also were obtained to help describe the responses more thoroughly. Participants who never experienced the specific interactions were removed prior to analysis because they could not respond to the question of how often the interactions occurred.

Second, the quantity variable was recoded to indicate low quantity of 1 to 2 interactions and high quantity of 3 or more interactions. The rationale for recoding the variable was that more frequent interactions were associated with growth in personal and cognitive development (Endo & Harpel, 1982; Pascarella & Terenzini, 1975). A simple frequency analysis then was conducted in order to understand the categorized quantity of each SFI.

Once the frequency analysis concluded, specific comparisons were made between low and high quantity for interactions occurring in ASOs and those occurring in other college activities in order to determine whether or not differences existed. A chi square analysis was used to compare the groups. As suggested by Isaac and Michaels (1990), interactions that had fewer than five participants in their frequency count were not analyzed for differences because the cell sizes were too small to make comparisons. Several interactions could not be compared because of small cell sizes.

The process conducted for the quantity variable was repeated for the quality variable. A simple frequency analysis as well as means and standard deviations were obtained for the original responses. Participants who did not experience the interactions were removed from the analysis. Then, the variable was recoded to indicate a low quality of less than 10 minutes and a high quality of 10 minutes or more, which has been used as a quality measure in previous studies (Johnson, 1997; Terenzini & Pascarella, 1980; Volkwein, et al., 1986). A simple frequency analysis was conducted on the recategorized variable to better describe the overall status of interaction quality. Then, a chi square analysis was used to compare the quality of interactions in ASOs and other college activities. Similar to the quantity comparisons, some quality data could not be compared due to small cell sizes.

To answer the fourth research question, a composite score was calculated for participants who experienced each SFI. This score was obtained by multiplying the participant's quantity score by his/her quality score. The rationale for this approach was that existing research on student-faculty interactions emphasizes the importance of both quantity and quality measures (Endo & Harpel, 1982; Kuh & Hu, 2001; Pascarella, et al., 1983; Pascarella & Terenzini, 1979a; Pascarella & Terenzini, 1980b; Terenzini, et al., 1984a; Volkwein, et al., 1986) and some researchers highlighted the need for understanding the quantity of SFIs in relation to the quality of SFIs (Alderman, 2008; Sax, et al., 2005). While some of the previous studies looked at both quantity and quality as separate factors, none brought them together as one factor by which to measure SFIs.

It was this researcher's contention that generating one composite score would yield a more effective estimate of student-faculty interactions. For example, when examined as separate factors, quality and quantity of SFIs often yield the same results for student learning. Quality interactions, or those over 10 minutes in length, were found to contribute to intellectual development (Terenzini & Pascarella, 1980; Volkwein, et al.,

62

1986); personal development (Terenzini & Pascarella, 1980); and persistence (Johnson, 1997). Quantity interactions, or those occurring more frequently over time, also were found to contribute to intellectual development (Endo & Harpel, 1982); personal development (Endo & Harpel, 1982); and persistence (Johnson, 1997; Pascarella & Terenzini, 1975).

Because of the similarity in outcomes and the repeated findings that SFIs are mostly brief and infrequent for many students (Feldman & Newcomb, 1969; Jaasma & Koper, 1999; Lewallen, 1995; Nadler & Nadler, 2001), it can be logically concluded that something more powerful is occurring in these interactions that has yet to be identified. For example, students experiencing frequent and lengthy interactions with a faculty member would mostly exhibit greater benefits than students who had a one-time, brief encounter. In another example, students who repeatedly spend five minutes with a faculty member during the semester to ask about homework may not have the same kind of benefits as students who have one-time, hour-long conversation with a faculty member that covered personal, developmental issues. Thus, student learning from SFIs could very well be generated by the interaction of both quantity and quality. A composite score was one way to examine this possibility.

Prior to the calculation of each composite score, it was determined that the possible range was one to 25, with one being the lowest possible composite score (1 X 1) and 25 being the highest possible composite score (5 X 5). The arithmetic mean of the possible scores was nine. Once the composite scores were calculated for each SFI, means, standard deviations, and confidence intervals at the 95% level were calculated to

help describe the responses. To continue the descriptive analysis, the composite score was then recoded into two levels. Because nine was the arithmetic mean of the possible scores, a low composite score was defined as less than nine while a high composite score was defined as nine or more. A simple frequency analysis was then conducted to determine the composite score occurrences for each interaction. A chi square analysis was conducted to examine differences between ASOs and other college activities. Small cell sizes were again a problem for some of the interactions and analysis could not be successfully conducted for these interactions.

To answer the final research question in the second research objective, classification was recoded into two categories – underclassmen (freshmen, sophomores) and upperclassmen (juniors, seniors). A simple frequency analysis was conducted to determine the composite scores categories by classification for each interaction. A chi square analysis was conducted in order to analyze any differences that occurred between classification groups for ASOs and other college activities. There were not enough underclassmen participants to provide enough data for comparisons. Therefore, the chi square analysis was not conducted for some comparisons and the fifth research question could not be fully addressed.

For the final research objective and questions, a simple frequency analysis was conducted on the composite scores by organizational role (member, leader) and by institutional size (small, large) to help describe the distribution of responses. A chi square analysis then was conducted for each characteristic in order to identify any differences occurring for ASO interactions and the interactions occurring in other college activities. Small cell sizes were an issue for some of the interactions and they could not be successfully analyzed.

CHAPTER IV RESULTS

This chapter provides the results of the study. It is organized into ten sections. The first section addresses the demographics of the participants as well as their responses to some descriptive questions. The second through eighth sections address each of the research questions that were generated to help determine where SFIs were occurring along with the quality and quantity of those interactions. A ninth section discusses the limitations of the study while a tenth section provides an overview of the findings.

Purpose of the Study

The purpose of this study was to describe what kind of student-faculty interactions are occurring in the context of academic student organizations as well as identify the quality and quantity of such interactions and what factors are involved with meaningful interactions. The study also determined how these interactions might differ from those occurring in other college activities and how the factors of classification, organizational status, and institutional size relate to quality and quantity of interactions in both contexts.

Demographics of Participants

Institutional undergraduate populations for the 2009 fall semester were collected prior to the implementation of the survey instrument. Table 1 illustrates these populations as well as the participant numbers from each institution.

TABLE 1. Undergraduate Populations and Responses by Institution

Population and Participants	University of St. Thomas	McMurry University	Texas State University	Texas A&M University
Undergraduate Population	1,792	1,515	24,810	38,958
Participants	33	13	24	34

Demographics for participants also were calculated. Table 2 depicts the gender and classification by institution. As illustrated by the table, the survey participants were mostly female and had a junior or senior classification. For gender, it is a common problem in survey research conducted with college students for females to respond in greater proportion than males (Sax, Gilmartin, & Bryant, 2003; Underwood, Kim, & Matier, 2000). With such a high percentage of females responding, it is possible that results could be impacted in favor of more relationship-based interactions which have been recognized as prevalent among women (Gilligan, 1982).

Organizational '	Гуре	University of St. Thomas (n=33)	McMurry University (n=13)	Texas State University (n=24)	Texas A&M University (n=34)
Gender	Female	94%	85%	83%	85%
	Male	6%	15%	17%	15%
Classification	Freshman	3%	8%		3%
	Sophomore	9%	8%	13%	12%
	Junior	36%	31%	17%	38%
	Senior	52%	54%	71%	47%

TABLE 2. Gender and Classification by Institution

All participants (n=104) were asked what academic discipline their ASOs represented. Almost a third (32%) said psychology, 23% said chemistry, 23% said health, 12% said English, and 11% said education. Each participant was asked to estimate the active membership for their ASOs. Those estimates ranged from 4 to 44 members. Almost all of the participants (94%) had a faculty advisor for their ASO.

In addition, participants were asked if they were part of a formal living-learning community where they live with other students, take classes together, and engage in academic activities together. Only 8% (n=8) said they were in a living-learning community. Of those eight students, five said they interacted with faculty as part of the community. A final question asked participants about their status in the ASOs selected for the study. Over half (59%) were members while 41% were leaders.

Research Question 1

The first research question asked "what types of student-faculty interactions exist within academic student organizations and how do they compare to those that occur in other settings?" A simple frequency analysis, contained in Table 3, revealed that many participants asked faculty members for guidance on courses, academic programs, career plans, and feedback on assignments. The least cited interactions with faculty included attending athletic and cultural events, teaching class, being a guest in a faculty member's home, and working with faculty on a committee or project that was affiliated with a group. In terms of the ASO environment, all SFIs were experienced by a range of 2% to 62% of the participants. For other college activities, the range was 7% to 81%.

Student-Faculty Interaction	n	Occurred in ASOs	Occurred in Other College Activities	Occurred in Both ASOs and Other College Activities	Did not occur
Asked a faculty member for comments or criticisms about your work (tests, papers, etc.)	104	6%	81%	6%	8%
Asked a faculty member for information or guidance related to a course you were taking (grades, make-up work,					
assignments, etc.)	104	10%	79%	7%	5%
Assisted a faculty member in teaching a class	103	3%	12%	1%	85%
Discussed ideas for a term paper or other class project with a faculty member Had a faculty member advise or supervise you on a faculty	103	7%	70%	3%	20%
research project	102	11%	31%	2%	56%
Had a faculty member advise or supervise you on a student research project (e.g., Honors thesis, independent study, etc.)	102	7%	40%	2%	51%
Had coffee, sodas, snacks, or meals with a professor (either in on-campus dining locations or off-campus at restaurants)	104	26%	19%	19%	36%
Met informally with faculty to discuss a campus issue or problem	103	16%	24%	6%	54%
Met informally with faculty to discuss matters related to my	105	1070	2470	070	5470
future career such as career plans and ambitions	103	22%	52%	15%	12%
Met informally with faculty to get basic information and advice about my academic program Met informally with faculty to help resolve a personal	102	17%	59%	13%	12%
informally with faculty to help resolve a personal lem ed informally (and outside of class) with an instructor it current events, campus activities, or other common	101	3%	32%	2%	63%
about current events, campus activities, or other common interests Was a guest in a professor's home for a meal or social	103	18%	38%	19%	25%
function	103	11%	13%	3%	74%
Worked with a faculty member outside of class on a committee or project	102	12%	8%	3%	78%
Had a faculty member accompany you to an athletic competition or event (this could include other people as well)	102	2%	7%		91%
Had a faculty member attend an arts or cultural event with					
you (this could include other people as well)	101	7%	11%	2%	80%
Had a faculty member supervise a student social function (e.g., party) that you attended	100	27%	18%	6%	49%
Personally invited a faculty member to speak at an event sponsored by a group you were affiliated with	102	27%	7%	2%	65%
Had a faculty member help organize or attend a field trip with a group you were affiliated with	103	29%	15%	2%	54%
Had a faculty member assist in organizing a presentation or workshop sponsored by a group you were affiliated with	100	35%	11%	3%	51%
Had a faculty member speak at a presentation or workshop sponsored by a group you were affiliated with	102	42%	11%	4%	43%
Had a faculty member attend meetings for a group you were affiliated with	101	60%	14%	4%	22%
Had a faculty member actively participate in meetings for a group you were affiliated with	103	62%	11%	4%	23%
Worked with a faculty member outside of class on a committee or project sponsored by a group you were affiliated with	101	18%	7%	1%	74%

TABLE 3. Overall Occurrence of SFIs for All Study Participants

Overall, 99% of participants had at least one interaction with faculty since they started college. Another 81% had at least one interaction with faculty within their ASO environment, and 96% had at least one interaction with faculty within their other college activities. Over three-quarters (78%) said they had interactions with faculty in both their ASO and other college activities.

The next step of analysis was to compare where the interactions had occurred by the different categories of participants. A chi square analysis was conducted for each comparison, and two significant results were found. Organizational status made a difference as to whether or not participants had any interactions with faculty within the context of their ASOs. As illustrated in Table 4, leaders were more likely than members to report SFIs in the ASO context and when the ASO context and other college activities were observed together. Institutional size and classification did not affect whether or not participants had any interactions with faculty or interactions in the two different contexts of ASOs and other college activities.

Type of Student-Faculty Interaction	Members	n	Leaders	n	χ^2
Experienced any SFI					
	100%	61	98%	43	
Experienced any SFI in the ASO context					
	71%	61	95%	43	**
Experienced any SFI in the Other College Activities					
context	97%	61	95%	43	
Experienced SFIs in both the ASO context and the Other College Activities context	67%	61	93%	43	**

TABLE 4. Frequency of SFIs by Organizational Status

df=1, * =p<0.05; ** =p<0.01

The findings reinforce previous research that two-thirds to almost all of students experience interactions with faculty while in college. However, the findings do not support conclusions made by past researchers that most interactions are occurring before or after class or during a faculty member's office hours. While the most frequently cited interactions could fall into this category, other interactions experienced by the participants in this study occurred outside of the classroom environment such as having coffee with a faculty member, attending a student social function, participating in organizational workshops or presentations, and attending organizational meetings. In addition, a majority of the participants experienced at least one interaction because of their involvement in an ASO, which is outside of the immediate classroom environment. It is clear from these data that there are more kinds of interactions occurring outside of the classroom environment than were previously recognized.

Research Question 2

The second research question asked "what is the quantity of student-faculty interactions within academic student organizations and how does it compare to those that occur in other settings?" First, a simple frequency analysis was conducted in order to obtain how frequently these interactions occurred since the participants started college. The scale used for the quantity questions was 1=1 Time, 2=2 Times, 3=3 Times, 4=4 Times, and 5=5 or More Times. Table 5 illustrates the responses along with the means and standard deviations for each interaction. Those who did not experience the interaction were removed from the analysis.

TABLE 5. Quantity of SFIs for All Study Participants Since They Started College

Student-Faculty Interaction	n	Mean (sd)	1 Time	2 Times	3 Times	4 Times	5 or Mor Times
Asked a faculty member for comments or criticisms about your work (tests, papers, etc.)	98	4.00 (1.27)	3%	15%	15%	11%	55%
Asked a faculty member for information or guidance related	70	(1.27)	370	1370	1570	1170	5570
to a course you were taking (grades, make-up work,		1.00					
assignments, etc.)	99	4.08 (1.36)	10%	5%	12%	12%	61%
	99	2.52	10%	J 70	1 2 70	1 2 70	0170
Assisted a faculty member in teaching a class	23	(1.56)	39%	13%	26%		22%
Discussed ideas for a term paper or other class project with a		3.60					
aculty member	86	(1.42)	12%	13%	19%	17%	40%
Had a faculty member advise or supervise you on a faculty research project	42	3.07 (1.61)	26%	17%	10%	19%	29%
	42	(1.01)	20%	1/%	10%	19%	29%
Had a faculty member advise or supervise you on a student		3.20					
research project (e.g., Honors thesis, independent study, etc.)	50	(1.69)	28%	10%	14%	10%	38%
Had coffee, sodas, snacks, or meals with a professor (either							
n on-campus dining locations or off-campus at restaurants)	65	3.20	1.40/	260/	150/	150/	200/
At informally with faculty to discuss a campus issue or	65	(1.46) 2.79	14%	26%	15%	15%	29%
problem	47	(1.35)	19%	26%	32%	4%	19%
	• •	(1.00)	1970	2070	02/0	170	1970
Met informally with faculty to discuss matters related to my future career such as career plans and ambitions		3.57					
*	93	(1.51)	15%	10%	24%	7%	45%
Met informally with faculty to get basic information and		3.83					
dvice about my academic program	90	(1.42)	12%	7%	17%	14%	50%
Met informally with faculty to help resolve a personal problem	42	2.83	29%	17%	21%	10%	24%
Falked informally (and outside of class) with an instructor	42	(1.55)	2970	1 / 70	2170	10%	2470
bout current events, campus activities, or other common		3.44					
nterests	73	(1.56)	16%	16%	16%	8%	43%
Was a guest in a professor's home for a meal or social		2.04					
unction	28	(1.26)	50%	18%	14%	14%	4%
Worked with a faculty member outside of class on a	•	2.90	1.5.4			4 = 4 /	4 = 0 /
committee or project	29	(1.37)	17%	28%	21%	17%	17%
Had a faculty member accompany you to an athletic		3.33					
competition or event (this could include other people as well)	12	(1.23)	8%	17%	25%	33%	17%
Had a faculty member attend an arts or cultural event with		· /					
you (this could include other people as well)		2.32					
	22	(1.46)	41%	23%	14%	9%	14%
Had a faculty member supervise a student social function e.g., party) that you attended	53	2.87	23%	25%	17%	15%	21%
	55	(1.47)	23%	2370	1 / 70	1,3 70	∠170
Personally invited a faculty member to speak at an event		2.79					
ponsored by a group you were affiliated with	39	(1.42)	28%	13%	23%	23%	13%
Had a faculty member help organize or attend a field trip		2.21					
vith a group you were affiliated with	48	(1.25)	40%	23%	21%	10%	6%
Had a faculty member assist in organizing a presentation or		2.72					
vorkshop sponsored by a group you were affiliated with	53	(1.38)	25%	21%	30%	8%	17%
	55	(1.50)	2070	21/0	5070	070	1770
Had a faculty member speak at a presentation or workshop		2.89					
ponsored by a group you were affiliated with	65	(1.48)	23%	22%	22%	11%	23%
Had a faculty member attend meetings for a group you were		3.81					
ffiliated with	77	(1.41)	10%	10%	16%	16%	48%
Had a faculty member actively participate in meetings for a group you were affiliated with	75	3.72 (1.44)	11%	13%	16%	13%	47%
Worked with a faculty member outside of class on a	15	(1.44)	1170	1370	1070	1.5 %	+ / /0
committee or project sponsored by a group you were		2 17					
		3.17					

The most frequent interactions were those relating to academic issues such as seeking advice for assignments or obtaining advice about an academic program. The next most frequently occurring interactions were having faculty members attend and actively participate in meetings for groups with which students are involved. The least frequent interactions were going to a faculty's home for a function, having faculty help organize a field trip, and attending an arts or cultural event with faculty. The quantity results could be influenced by the characteristics of the millennial generation (Strange, 2004). This generation may see faculty members as authority figures who have rules that students want to understand and follow in their academic programs, so they will seek them out for advice. Or they may prefer to interact with faculty only when they are in the company of their peers such as organization meetings rather than spending time with them in smaller environments such as the faculty member's home. Future studies may need to examine the role of generational characteristics to determine whether or not it has an impact on how students describe and prefer interactions with faculty.

In order to conduct comparisons and provide additional descriptive data, the responses were re-categorized into low quantity (1 to 2) and high quantity (3 or more) interactions. Table 6 demonstrates that the majority of the participants who experienced each interaction did so in high quantity since they started college. However, most of the participants in this study were upperclassmen, meaning that interactions occurred over three or four years and would amount to one or two interactions per year. This is infrequent when a student's entire college career is considered.

Following the re-categorization of the quantity variable, a chi-square analysis was conducted to determine if differences existed between the quantity of SFIs for the ASO environment and other college activities. Two specific interactions could not be successfully analyzed due to small cell sizes in the ASO context. The analysis, which can be viewed in Table 7, immediately following Table 6, revealed that there were no significant differences for any of the interactions in terms of quantity. When the interactions occurred in both contexts, the majority of the participants experienced them in high quantity.

From this study, it does not appear that quantity significantly differs for SFIs that occur in either ASOs or other college activities. However, there appears to be a pattern of responses that could be explored further. In 17 of the 24 listed interactions, the participants who experienced the interactions in an ASO environment had a greater portion of high quantity interactions than participants who experienced the interactions in their other college activities. It may be that students who join academic student organizations have a few more opportunities to interact with faculty than students who do not join these organizations. Another study with a large number of participants might help determine if this pattern reoccurs or is just an anomaly for this study.

Student-Faculty Interaction	n	Low Quantity: 1-2 Times	High Quantity: 3 or More Times
Asked a faculty member for comments or criticisms about your work (tests, papers, etc.)	98	15%	85%
Asked a faculty member for information or guidance related to a course you were taking grades, make-up work, assignments, etc.)	99	5%	95%
Assisted a faculty member in teaching a class	23	13%	87%
Discussed ideas for a term paper or other class project with a faculty member	86	13%	87%
Had a faculty member advise or supervise you on a faculty research project	42	17%	83%
Had a faculty member advise or supervise you on a student research project (e.g., Honors hesis, independent study, etc.)	50	10%	90%
Had coffee, sodas, snacks, or meals with a professor (either in on-campus dining locations or off-campus at restaurants)	65	26%	74%
Met informally with faculty to discuss a campus issue or problem	47	26%	75%
Met informally with faculty to discuss matters related to my future career such as career plans and ambitions	93	10%	90%
Met informally with faculty to get basic information and advice about my academic program	90	7%	93%
Met informally with faculty to help resolve a personal problem	42	17%	83%
Talked informally (and outside of class) with an instructor about current events, campus activities, or other common interests	73	18%	82%
Was a guest in a professor's home for a meal or social function	28	18%	82%
Worked with a faculty member outside of class on a committee or project	29	28%	72%
Had a faculty member accompany you to an athletic competition or event (this could nclude other people as well)	12	17%	83%
Had a faculty member attend an arts or cultural event with you (this could include other people as well)	22	27%	73%
Had a faculty member supervise a student social function (e.g., party) that you attended	53	25%	76%
Personally invited a faculty member to speak at an event sponsored by a group you were affiliated with			
Had a faculty member help organize or attend a field trip with a group you were affiliated with	39 48	15% 23%	85% 77%
Had a faculty member assist in organizing a presentation or workshop sponsored by a group you were affiliated with	53	21%	79%
Had a faculty member speak at a presentation or workshop sponsored by a group you were affiliated with	65	22%	79%
Had a faculty member attend meetings for a group you were affiliated with	77	10%	90%
Had a faculty member actively participate in meetings for a group you were affiliated with	75	13%	87%
Worked with a faculty member outside of class on a committee or project sponsored by a group you were affiliated with	-	-	

TABLE 6. Low and High Quantity SFIs for All Study Participants

	ASOs			Other College Activities				
Student-Faculty Interaction	Low Quantity	High Quantity	n	Low Quantity	High Quantity	n	χ^2	
Asked a faculty member for comments or criticisms about your work (tests, papers, etc.)	8%	92%	12	15%	85%	88	.36	
Asked a faculty member for information or guidance related to a course you were taking (grades, make-up								
work, assignments, etc.)	0%	100%	17	6%	94%	86	1.09	
Assisted a faculty member in teaching a class	0%	100%	4	25%	75%	12	***	
Discussed ideas for a term paper or other class project with a faculty member Had a faculty member advise or supervise you on a	11%	89%	9	11%	89%	71	.00	
faculty research project Had a faculty member advise or supervise you on a	0%	100%	13	22%	79%	27	3.40	
student research project (e.g., Honors thesis, independent study, etc.) Had coffee, sodas, snacks, or meals with a professor	0%	100%	8	13%	87%	39	1.15	
(either in on-campus dining locations or off-campus at restaurants)	17%	83%	46	31%	69%	36	1.97	
Met informally with faculty to discuss a campus issue or problem	32%	68%	19	30%	70%	23	.00	
Met informally with faculty to discuss matters related to my future career such as career plans and ambitions	5%	95%	38	9%	91%	65	.53	
Met informally with faculty to get basic information and advice about my academic program Met informally with faculty to help resolve a personal	3%	97%	30	9%	91%	67	.98	
problem Talked informally (and outside of class) with an instructor about current events, campus activities, or	20%	80%	5	19%	81%	31	.00	
other common interests Was a guest in a professor's home for a meal or social	16%	84%	37	13%	87%	54	.19	
function Worked with a faculty member outside of class on a	15%	85%	13	33%	67%	15	1.20	
committee or project Had a faculty member accompany you to an athletic	18%	82%	11	40%	60%	10	1.22	
competition or event (this could include other people as well)	0%	100%	2	40%	60%	5	***	
Had a faculty member attend an arts or cultural event with you (this could include other people as well)	1.40/	960/	7	220/	790/	9	16	
Had a faculty member supervise a student social function (e.g., party) that you attended	14% 28%	86% 72%	7 32	22% 22%	78% 78%	9 18	.16 .21	
Personally invited a faculty member to speak at an event sponsored by a group you were affiliated with								
Had a faculty member help organize or attend a field trip	16%	84%	25	25%	75%	8	.33	
with a group you were affiliated with Had a faculty member assist in organizing a presentation or workshop sponsored by a group you were affiliated	15%	85%	27	39%	62%	13	2.81	
with	24%	77%	34	20%	80%	10	.06	
Had a faculty member speak at a presentation or workshop sponsored by a group you were affiliated with	21%	80%	44	15%	85%	13	.17	
Had a faculty member attend meetings for a group you were affiliated with	10%	90%	60	19%	81%	16	.93	
Had a faculty member actively participate in meetings for a group you were affiliated with	11%	89%	62	31%	69%	13	3.26	
Worked with a faculty member outside of class on a committee or project sponsored by a group you were affiliated with	6%	94%	16	14%	86%	7	.40	

df=1, * =p<0.05; ** =p<0.01, ***=Could Not Compute

Research Question 3

The third research question asked "what is the quality of student-faculty interactions within academic student organizations and how does it compare to those that occur in other settings?" A simple frequency analysis first was conducted in order to demonstrate the quality of these interactions since the participants started college. The scale used for the quality questions was 1=Less than 10 Minutes, 2=10 to 15 Minutes, 3=16 to 30 Minutes, 4=31 to 60 minutes, and 5=More than 60 Minutes. Table 8 illustrates the responses along with the means and standard deviations for each interaction. Those who did not experience the interaction were not included in the analysis.

The highest quality interactions tended to be functions that were designed to be longer in length such as social functions and meetings. The lowest quality interactions were those involving informal discussions for campus issues or personal guidance that could vary in length.

TABLE 8. Quality of SFIs for All Study Participants

Student-Faculty Interaction	n	Mean (sd)	Less than 10 minutes	10 to 15 minutes	16 to 30 minutes	31 to 60 minutes	More than 60 minute
Asked a faculty member for comments or criticisms		2.67					
about your work (tests, papers, etc.)	91	(1.27)	21%	30%	21%	19%	10%
Asked a faculty member for information or guidance							
related to a course you were taking (grades, make-up		2.63					
work, assignments, etc.)	95	(1.19)	20%	28%	27%	17%	7%
Assisted a faculty member in teaching a class	~~	2.96		2 4 6 4	4 501		
Discussed ideas for a term paper or other class project	25	(1.46) 2.53	20%	24%	16%	20%	20%
with a faculty member	80	(1.19)	24%	29%	24%	19%	5%
Had a faculty member advise or supervise you on a		3.29					
faculty research project	41	(1.45)	17%	12%	24%	17%	29%
Had a faculty member advise or supervise you on a student research project (e.g., Honors thesis, independent		3.15					
student research project (e.g., Honors thesis, independent study, etc.)	52	(1.45)	17%	21%	14%	25%	23%
Had coffee, sodas, snacks, or meals with a professor	02	(11.0)	1770	21/0	11/0	2070	2070
(either in on-campus dining locations or off-campus at		3.57					
restaurants)	61	(1.28)	12%	10%	13%	41%	25%
Met informally with faculty to discuss a campus issue or problem	42	2.40 (1.33)	33%	24%	21%	12%	10%
	42	(1.55)	3370	2470	2170	12/0	10%
Met informally with faculty to discuss matters related to		3.02					
ny future career such as career plans and ambitions	85	(1.19)	12%	22%	28%	27%	11%
Met informally with faculty to get basic information and	02	2.81	150/	280/	200/	220/	70/
Idvice about my academic program Met informally with faculty to help resolve a personal	83	(1.16) 2.53	15%	28%	28%	23%	7%
problem	40	(1.18)	25%	23%	33%	15%	5%
Falked informally (and outside of class) with an							
nstructor about current events, campus activities, or		2.51	2004	2.10/	2504	100/	0.04
other common interests Was a guest in a professor's home for a meal or social	67	(1.28) 3.96	28%	24%	25%	13%	9%
function	27	(1.53)	15%	7%	4%	15%	59%
Worked with a faculty member outside of class on a		3.37					
committee or project	27	(1.42)	11%	22%	15%	22%	30%
Had a faculty member accompany you to an athletic		2 (0					
competition or event (this could include other people as well)	13	2.69 (1.49)	31%	15%	23%	15%	15%
,	15	(1.47)	5170	1570	2370	1570	1370
Had a faculty member attend an arts or cultural event with you (this could include other people as well)		3.37					
	19	(1.34)	5%	32%	11%	26%	26%
Had a faculty member supervise a student social function e.g., party) that you attended	47	4.04 (1.23)	4%	11%	15%	17%	53%
	47	(1.23)	470	11/0	1570	1770	5570
Personally invited a faculty member to speak at an event sponsored by a group you were affiliated with		2.66					
	38	(1.36)	29%	13%	34%	11%	13%
Had a faculty member help organize or attend a field trip with a group you were affiliated with	44	3.20 (1.37)	11%	23%	27%	11%	27%
Had a faculty member assist in organizing a presentation	77	(1.57)	11/0	<i></i>	21/0	11/0	21/0
or workshop sponsored by a group you were affiliated		3.10					
vith	49	(1.30)	14%	14%	39%	12%	20%
Had a faculty member speak at a presentation or		3.56					
workshop sponsored by a group you were affiliated with	59	(1.09)	3%	14%	29%	32%	22%
Had a faculty member attend meetings for a group you		3.81					
vere affiliated with	74	(.99)	1%	10%	23%	39%	27%
Had a faculty member actively participate in meetings	71	3.63	60/	1.40/	160/	410/	240/
for a group you were affiliated with Worked with a faculty member outside of class on a	71	(1.16)	6%	14%	16%	41%	24%
committee or project sponsored by a group you were		3.13					
affiliated with	32	(1.41)	16%	22%	19%	22%	22%

In order to conduct comparisons and provide additional descriptive data, the responses were re-categorized into low quality (less than 10 minutes) and high quality (10 minutes or more) interactions. Table 9 demonstrates that the majority of the participants had high quality interactions with faculty since they started college. This contradicts previous literature that concludes that most interactions between students and faculty last less than 10 minutes (Jaasma & Koper, 1999; Lewallen, 1995).

Following the re-categorization of the quality variable, a chi-square analysis was conducted to determine if differences existed between the quality of SFIs for the ASO environment and other college activities. The same two interactions that presented cell size problems for the quantity analysis also caused the same problems for the quality analysis. For the rest of the interactions, comparisons were made and can be viewed in Table 10, depicted immediately after Table 9. The results revealed that there were no significant differences for any of the interactions in terms of quality. When the interactions occurred in both contexts, the majority of the participants had high quality interactions.

sekked a faculty member for comments or criticisms about your work (tests, papers, etc.)9121%sekked a faculty member for information or guidance related to a course you were taking grades, make-up work, assignments, etc.)9521%sessisted a faculty member in teaching a class2520%biscussed ideas for a term paper or other class project with a faculty member8024%lad a faculty member advise or supervise you on a faculty research project4117%lad a faculty member advise or supervise you on a student research project (e.g., Honors nesis, independent study, etc.)5219%lad offee, sodas, snacks, or meals with a professor (either in on-campus dining locations) refricampus at restaurants)6112%At informally with faculty to discuss a campus issue or problem4233%det informally with faculty to discuss matters related to my future career such as career lans and ambitions8316%fet informally with faculty to lelp resolve a personal problem4025%Vorked with a faculty member outside of class on a committee or project2711%lad a faculty member accompany you to an athletic competition or event (this could aclude other people as well)1331%lad a faculty member actult as or cultural event with you (this could include other ecople as well)1331%lad a faculty member help organize or attend a field trip with a group you were affiliated with ad a faculty member help organize or attend a field trip with a group you were affiliated with filiated with4914%lad a faculty member actively participate in me	Quality: High Than 10 Quality tes or Mor	nutes
grades, make-up work, assignments, etc.)9521%assisted a faculty member in teaching a class2520%biscussed ideas for a term paper or other class project with a faculty member8024%Iad a faculty member advise or supervise you on a faculty research project4117%Iad a faculty member advise or supervise you on a student research project (e.g., Honors5219%Iad coffee, sodas, snacks, or meals with a professor (either in on-campus dining locations6112%Act informally with faculty to discuss a campus issue or problem4233%Act informally with faculty to get basic information and advice about my academic8512%Iad offee, sodas, snacks, or meals with a professor (either in on-campus dining locations8512%Act informally with faculty to get basic information and advice about my academic8512%Iat informally (and outside of class) with an instructor about current events, campus6728%Vas a guest in a professor's home for a meal or social function2711%Iad a faculty member accompany you to an athletic competition or event (this could nclude other people as well)1331%Iad a faculty member supervise a student social function (e.g., party) that you attended with a faculty member help organize or attend a field trip with a group you were affiliated with4411%Iad a faculty member attend meetings for a group you were affiliated with Iad a faculty member assist in organizing a presentation or workshop sponsored by a group you were affiliated with593%Iad a faculty member attend meeting	79%	
sessisted a faculty member in teaching a class2520%Discussed ideas for a term paper or other class project with a faculty member8024%Iad a faculty member advise or supervise you on a faculty research project (e.g., Honorseis, independent study, etc.)5219%Iad coffee, sodas, snacks, or meals with a professor (either in on-campus dining locations or off-campus at restaurants)6112%Act informally with faculty to discuss a campus issue or problem4233%Act informally with faculty to discuss matters related to my future career such as career lans and ambitions8512%Act informally with faculty to belp resolve a personal problem4025%Act informally with faculty to belp resolve a personal problem7115%Act a faculty member accompany you to an athletic competition or event (this could informally member accompany you to an athletic competition or event (this could include other common interests7628%Vas a guest in a professor's home for a meal or social function715%14%Iad a faculty member accompany you to an athletic competition or event (this could include other people as well)1316%Iad a faculty member supervise a student social function (e.g., party) that you attended7628%Iad a faculty member supervise a student social function or event (this could include other people as well)715%Iad a faculty member supervise a student social function (e.g., party) that you attended tith7678Iad a faculty member supervise a student social function (e.g., party) that you attended7678Iad a f	79%	
Naccussed ideas for a term paper or other class project with a faculty member8024%Iad a faculty member advise or supervise you on a faculty research project4117%Iad a faculty member advise or supervise you on a student research project (e.g., Honors resis, independent study, etc.)5219%Iad coffee, sodas, snacks, or meals with a professor (either in on-campus dining locations r off-campus at restaurants)6112%Act informally with faculty to discuss a campus issue or problem4233%At informally with faculty to discuss matters related to my future career such as career lans and ambitions8512%Act informally with faculty to get basic information and advice about my academic rogram6025%'alked informally with faculty to help resolve a personal problem4025%'alked informally (and outside of class) with an instructor about current events, campus citvities, or other common interests6728%Vorked with a faculty member outside of class on a committee or project1331%Iad a faculty member attend an arts or cultural event with you (this could include other ecople as well)195%Iad a faculty member supervise a student social function (e.g., party) that you attended tith4111%Iad a faculty member attend meetings for a group you were affiliated with4914%Iad a faculty member supervise a student social function (e.g., party) that you attended tith4111%Iad a faculty member attend negring a presentation or workshop sponsored by a group you were affiliated with293%	80%	
Iad a faculty member advise or supervise you on a faculty research project4117%Iad a faculty member advise or supervise you on a student research project (e.g., Honors nesis, independent study, etc.)5219%Iad a faculty member advise or supervise you on a student research project (e.g., Honors r off-campus at restaurants)6112%Iad coffee, sodas, snacks, or meals with a professor (either in on-campus dining locations r off-campus at restaurants)6112%Aet informally with faculty to discuss a campus issue or problem4233%Aet informally with faculty to get basic information and advice about my academic rogram8512%Aet informally with faculty to help resolve a personal problem4025%Alked informally (and outside of class) with an instructor about current events, campus ctivities, or other common interests6728%Vas a guest in a professor's home for a meal or social function2715%Iad a faculty member outside of class on a committee or project1331%Iad a faculty member accompany you to an athletic competition or event (this could nclude other people as well)1331%Iad a faculty member supervise a student social function (e.g., party) that you attended with4411%Iad a faculty member help organize or attend a field trip with a group you were affiliated with3829%Iad a faculty member help organizing a presentation or workshop sponsored by a group you were affiliated with4914%Iad a faculty member attend meetings for a group you were affiliated with593%Iad a	76%	
nessis, independent study, etc.)5219%Iad coffee, sodas, snacks, or meals with a professor (either in on-campus dining locations r off-campus at restaurants)6112%Adet informally with faculty to discuss a campus issue or problem4233%At informally with faculty to discuss matters related to my future career such as career lans and ambitions8512%At informally with faculty to get basic information and advice about my academic rogram8316%At informally with faculty to help resolve a personal problem4025%At informally and outside of class) with an instructor about current events, campus ctivities, or other common interests6728%Vas a guest in a professor's home for a meal or social function2715%Vorked with a faculty member outside of class on a committee or project2711%Iad a faculty member accompany you to an athletic competition or event (this could nclude other people as well)1331%Iad a faculty member supervise a student social function (e.g., party) that you attended with474%At a faculty member help organize or attend a field trip with a group you were affiliated with4914%Iad a faculty member speak at a presentation or workshop sponsored by a group you were ffiliated with4914%Iad a faculty member speak at a presentation or workshop sponsored by a group you were affiliated with593%Iad a faculty member attend meetings for a group you were affiliated with741%Iad a faculty member actively participate in meetings for a group you were affiliat	83%	
r off-campus at restaurants) 61 12% Aet informally with faculty to discuss a campus issue or problem 42 33% Aet informally with faculty to discuss matters related to my future career such as career lans and ambitions 85 12% Aet informally with faculty to get basic information and advice about my academic rogram 83 16% Aet informally with faculty to help resolve a personal problem 40 25% Talked informally (and outside of class) with an instructor about current events, campus ctivities, or other common interests 67 28% Vas a guest in a professor's home for a meal or social function 27 15% Vorked with a faculty member outside of class on a committee or project 27 11% Iad a faculty member accompany you to an athletic competition or event (this could aclude other people as well) 13 31% Iad a faculty member supervise a student social function (e.g., party) that you attended 47 4% tersonally invited a faculty member to speak at an event sponsored by a group you were affiliated with 42 33% Iad a faculty member help organize or attend a field trip with a group you were affiliated vith 11% Iad a faculty member supervise a student social function or workshop sponsored by a roup you were affiliated with 49 14% Iad a faculty member accompany pou were affiliated with 59 3% Iad a faculty member speak at a presentation or workshop sponsored by a group you were affiliated with 59 3% Iad a faculty member attend meetings for a group you were affiliated with 74 1% Iad a faculty member attend meetings for a group you were affiliated with 74 1% Iad a faculty member actively participate in meetings for a group you were affiliated with 74 1%	81%	
4233%Aet informally with faculty to discuss matters related to my future career such as career lans and ambitions85Aet informally with faculty to get basic information and advice about my academic rogram85Aet informally with faculty to help resolve a personal problem4025%28%Alked informally (and outside of class) with an instructor about current events, campus ctivities, or other common interests67Vas a guest in a professor's home for a meal or social function27Vorked with a faculty member outside of class on a committee or project27If ad a faculty member accompany you to an athletic competition or event (this could nclude other people as well)13If a faculty member supervise a student social function (e.g., party) that you attended47Aff a faculty member help organize or attend a field trip with a group you were affiliated with38Aga faculty member asist in organizing a presentation or workshop sponsored by a roup you were affiliated with49Id a faculty member speak at a presentation or workshop sponsored by a group you were affiliated with49Id a faculty member actively participate in meetings for a group you were affiliated with59Id a faculty member actively participate in meetings for a group you were affiliated with71	89%	
Act informally with faculty to discuss matters related to my future career such as career lans and ambitions8512%Act informally with faculty to get basic information and advice about my academic rogram8316%Act informally with faculty to help resolve a personal problem4025%Alked informally (and outside of class) with an instructor about current events, campus ctivities, or other common interests6728%Vas a guest in a professor's home for a meal or social function2715%Vorked with a faculty member outside of class on a committee or project2711%Iad a faculty member accompany you to an athletic competition or event (this could nclude other people as well)1331%Iad a faculty member supervise a student social function (e.g., party) that you attended with474%Personally invited a faculty member to speak at an event sponsored by a group you were affiliated with3829%Iad a faculty member assist in organizing a presentation or workshop sponsored by a roup you were affiliated with4914%Iad a faculty member speak at a presentation or workshop sponsored by a group you were affiliated with593%Iad a faculty member actively participate in meetings for a group you were affiliated with716%	67%	
Met informally with faculty to get basic information and advice about my academic rogram8316%Met informally with faculty to help resolve a personal problem4025%Alked informally (and outside of class) with an instructor about current events, campus ctivities, or other common interests6728%Vas a guest in a professor's home for a meal or social function2715%Vorked with a faculty member outside of class on a committee or project2711%Ida d faculty member accompany you to an athletic competition or event (this could nclude other people as well)1331%Ida d faculty member supervise a student social function (e.g., party) that you attended with a faculty member to speak at an event sponsored by a group you were affiliated with3829%Ida a faculty member assist in organizing a presentation or workshop sponsored by a roup you were affiliated with4914%Ida a faculty member speak at a presentation or workshop sponsored by a group you were affiliated with593%Ida a faculty member assist in organizing a presentation or workshop sponsored by a group you were affiliated with593%Ida a faculty member assist in organizing a presentation or workshop sponsored by a group you were affiliated with593%Ida a faculty member attend meetings for a group you were affiliated with741%Ida a faculty member attend meetings for a group you were affiliated with593%Ida a faculty member actively participate in meetings for a group you were affiliated with716%	88%	
Act informally with faculty to help resolve a personal problem4025%Yalked informally (and outside of class) with an instructor about current events, campus ctivities, or other common interests6728%Vas a guest in a professor's home for a meal or social function2715%Vorked with a faculty member outside of class on a committee or project2711%Had a faculty member accompany you to an athletic competition or event (this could nelude other people as well)1331%Had a faculty member attend an arts or cultural event with you (this could include other eeople as well)195%Had a faculty member supervise a student social function (e.g., party) that you attended with474%Personally invited a faculty member to speak at an event sponsored by a group you were ffiliated with3829%Had a faculty member speak at a presentation or workshop sponsored by a roup you were affiliated with4914%Had a faculty member speak at a presentation or workshop sponsored by a group you were affiliated with593%Had a faculty member attend meetings for a group you were affiliated with741%Had a faculty member actively participate in meetings for a group you were affiliated with716%	84%	
ctivities, or other common interests6728%Vas a guest in a professor's home for a meal or social function2715%Vorked with a faculty member outside of class on a committee or project2711%Had a faculty member accompany you to an athletic competition or event (this could nclude other people as well)1331%Had a faculty member attend an arts or cultural event with you (this could include other eople as well)195%Had a faculty member supervise a student social function (e.g., party) that you attended with474%Versonally invited a faculty member to speak at an event sponsored by a group you were ffiliated with3829%Had a faculty member help organize or attend a field trip with a group you were affiliated with4411%Had a faculty member speak at a presentation or workshop sponsored by a roup you were affiliated with593%Had a faculty member speak at a presentation or workshop sponsored by a group you were ffiliated with593%Had a faculty member attend meetings for a group you were affiliated with141%Had a faculty member actively participate in meetings for a group you were affiliated with716%	75%	
Vas a guest in a professor's home for a meal or social function2715%Vorked with a faculty member outside of class on a committee or project2711%Iad a faculty member accompany you to an athletic competition or event (this could nclude other people as well)1331%Iad a faculty member attend an arts or cultural event with you (this could include other eople as well)195%Iad a faculty member supervise a student social function (e.g., party) that you attended with474%Versonally invited a faculty member to speak at an event sponsored by a group you were ffiliated with3829%Iad a faculty member supervise a student social function or workshop sponsored by a roup you were affiliated with4914%Iad a faculty member speak at a presentation or workshop sponsored by a ffiliated with593%Iad a faculty member attend meetings for a group you were affiliated with741%Iad a faculty member attend meetings for a group you were affiliated with716%	72%	
IndexIndexIndexIndexIdad a faculty member accompany you to an athletic competition or event (this could nclude other people as well)1331%Idad a faculty member attend an arts or cultural event with you (this could include other eople as well)195%Idad a faculty member supervise a student social function (e.g., party) that you attended474%Versonally invited a faculty member to speak at an event sponsored by a group you were ffiliated with3829%Idad a faculty member help organize or attend a field trip with a group you were affiliated vith4411%Idad a faculty member speak at a presentation or workshop sponsored by a roup you were affiliated with4914%Idad a faculty member speak at a presentation or workshop sponsored by a group you were ffiliated with593%Idad a faculty member attend meetings for a group you were affiliated with741%Idad a faculty member actively participate in meetings for a group you were affiliated with6%	85%	
Include other people as well)1331%IAd a faculty member attend an arts or cultural event with you (this could include other eople as well)195%IAd a faculty member supervise a student social function (e.g., party) that you attended474%Versonally invited a faculty member to speak at an event sponsored by a group you were ffiliated with3829%IAd a faculty member help organize or attend a field trip with a group you were affiliated with4411%IAd a faculty member assist in organizing a presentation or workshop sponsored by a roup you were affiliated with4914%IAd a faculty member speak at a presentation or workshop sponsored by a group you were ffiliated with593%IAd a faculty member attend meetings for a group you were affiliated with741%IAd a faculty member actively participate in meetings for a group you were affiliated with716%	89%	
eople as well)195%Had a faculty member supervise a student social function (e.g., party) that you attended474%Versonally invited a faculty member to speak at an event sponsored by a group you were ffiliated with3829%Had a faculty member help organize or attend a field trip with a group you were affiliated with4411%Had a faculty member assist in organizing a presentation or workshop sponsored by a roup you were affiliated with4914%Had a faculty member speak at a presentation or workshop sponsored by a group you were ffiliated with593%Had a faculty member attend meetings for a group you were affiliated with741%Had a faculty member actively participate in meetings for a group you were affiliated with716%	69%	
Image: A standard a faculty member supervise a student social function (e.g., party) that you attended195%Had a faculty member supervise a student social function (e.g., party) that you attended474%Personally invited a faculty member to speak at an event sponsored by a group you were ffiliated with3829%Had a faculty member help organize or attend a field trip with a group you were affiliated vith4411%Had a faculty member assist in organizing a presentation or workshop sponsored by a roup you were affiliated with4914%Had a faculty member speak at a presentation or workshop sponsored by a group you were ffiliated with593%Had a faculty member attend meetings for a group you were affiliated with741%Had a faculty member actively participate in meetings for a group you were affiliated with716%		
474%Personally invited a faculty member to speak at an event sponsored by a group you were ffiliated with381029%11%11%12411%12411%12514%12612%12612%12712%12812%12914%12914%12914%12012%12012%12012%12112%12112%12112%12212%12312%12412%12512%12612%12712%12812%12912%12012%12012%12112% <td>95%</td> <td></td>	95%	
ffiliated with38Idad a faculty member help organize or attend a field trip with a group you were affiliated4411%Idad a faculty member assist in organizing a presentation or workshop sponsored by a roup you were affiliated with49Idad a faculty member speak at a presentation or workshop sponsored by a group you were ffiliated with59Idad a faculty member attend meetings for a group you were affiliated with59Idad a faculty member attend meetings for a group you were affiliated with71Idad a faculty member actively participate in meetings for a group you were affiliated with6%	96%	
vith4411%Had a faculty member assist in organizing a presentation or workshop sponsored by a roup you were affiliated with4914%Had a faculty member speak at a presentation or workshop sponsored by a group you were ffiliated with593%Had a faculty member attend meetings for a group you were affiliated with741%Had a faculty member attend meetings for a group you were affiliated with6%	71%	
roup you were affiliated with4914%Had a faculty member speak at a presentation or workshop sponsored by a group you were ffiliated with593%Had a faculty member attend meetings for a group you were affiliated with741%Had a faculty member actively participate in meetings for a group you were affiliated with6%	89%	
ffiliated with593%Had a faculty member attend meetings for a group you were affiliated with741%Had a faculty member actively participate in meetings for a group you were affiliated with716%	86%	
Iad a faculty member attend meetings for a group you were affiliated with741%Iad a faculty member actively participate in meetings for a group you were affiliated with716%	97%	
/1 0%	99%	
Vorked with a faculty member outside of class on a committee or project sponsored by a	94%	
roup you were affiliated with 32 19%	81%	

TABLE 9. Low and High Quality SFIs for All Study Participants

	ASOs			Other Col	lege Activitie	es	
Student-Faculty Interaction	Low Quality	High Quality	n	Low Quality	High Quality	n	χ ²
Asked a faculty member for comments or criticisms about your work (tests, papers, etc.)	25%	75%	12	21%	79%	82	.11
Asked a faculty member for information or guidance related to a course you were taking (grades, make-up work, assignments, etc.)	12%	88%	17	22%	78%	82	.91
Assisted a faculty member in teaching a class							
Discussed ideas for a term paper or other class project with a	25%	75%	4	9%	91%	11	**:
faculty member Had a faculty member advise or supervise you on a faculty research project	22% 17%	78% 83%	9 12	22% 12%	79% 89%	65 26	.00
	1 / %	83%	12	12%	89%	20	.15
Had a faculty member advise or supervise you on a student research project (e.g., Honors thesis, independent study, etc.)	13%	88%	8	11%	89%	37	.20
Had coffee, sodas, snacks, or meals with a professor (either in on-campus dining locations or off-campus at restaurants)	0.04		10		0.1.04		
Met informally with faculty to discuss a campus issue or	8%	93%	40	9%	91%	35	.03
problem	32%	68%	19	40%	60%	20	.30
Met informally with faculty to discuss matters related to my future career such as career plans and ambitions	9%	91%	34	12%	88%	60	.19
Met informally with faculty to get basic information and advice about my academic program Met informally with faculty to help resolve a personal	11%	89%	27	11%	89%	62	.00
problem Falked informally (and outside of class) with an instructor	20%	80%	5	14%	86%	28	.11
about current events, campus activities, or other common interests Was a guest in a professor's home for a meal or social	24%	76%	33	26%	74%	50	.03
function Worked with a faculty member outside of class on a	8%	92%	12	7%	93%	14	.01
committee or project	11%	89%	9	14%	86%	7	.04
Had a faculty member accompany you to an athletic competition or event (this could include other people as well)	50%	50%	2	0%	100%	4	**
Had a faculty member attend an arts or cultural event with you (this could include other people as well)	17%	83%	6	0%	100%	7	1.2
Had a faculty member supervise a student social function (e.g., party) that you attended	4%	96%	27	6%	94%	, 17	.11
Personally invited a faculty member to speak at an event sponsored by a group you were affiliated with	26%	74%	23	38%	63%	8	.38
Had a faculty member help organize or attend a field trip with a group you were affiliated with	12%	88%	25	0%	100%	11	1.4
Had a faculty member assist in organizing a presentation or workshop sponsored by a group you were affiliated with	10%	90%	30	11%	89%	9	.01
Had a faculty member speak at a presentation or workshop sponsored by a group you were affiliated with	2%	98%	41	0%	100%	11	.27
Had a faculty member attend meetings for a group you were affiliated with	2%	98%	55	0%	100%	15	.28
Had a faculty member actively participate in meetings for a group you were affiliated with Worked with a faculty member outside of class on a	5%	95%	57	8%	92%	12	.17
worked with a faculty member outside of class on a committee or project sponsored by a group you were affiliated with	7%	93%	15	0%	100%	5	.35

TABLE 10. Quality of SFIs for ASOs and Other College Activities

df=1, * =p<0.05; ** =p<0.01, ***=Could Not Compute

Research Question 4

The fourth research question asked "what is the combined quantity and quality of student-faculty interactions within academic student organizations and how does it compare to those that occur in other settings?" This was an opportunity to redefine how interactions were viewed based on the interaction of quantity and quality measures. To determine the composite scores, the quantity and quality responses were multiplied for each interaction, per participant. For instance, a participant who reported a one time interaction with a faculty member (quantity rating of 1) that lasted for less than 10 minutes (quality rating of 1) would have a composite score of 1 and would be considered a low composite interaction. A participant who reported more than five interactions with a faculty rating of 5) that lasted for an average of 60 minutes or more (quality rating of 5) would have a composite score of 25 and would be considered a high composite interaction. Composite scores ranged from 1 to 25, with an arithmetic mean of 9.

Once all composite scores were calculated, means and standard deviations were computed for each interaction. Considering the overall arithmetic mean of the possible composite scores, which was 9, seven interactions had means that fell below the arithmetic mean, another three had means that were around the arithmetic mean, and fourteen had means above the arithmetic mean. Based on where the interactions were in comparison to the arithmetic mean, more than half of the interactions were in the high quantity – high quality range. The overall mean and standard deviation were calculated for all of the composite scores for each interaction. Only one interaction was observed to be outside the 95% confidence interval – meeting informally with faculty to help resolve a personal problem. This interaction was below the lower control limit and rated in the lower portion of both the quantity and quality measures. While this is an interaction that appeared in the first studies regarding SFIs and several subsequent studies (See Appendix A), it does not seem to be an interaction that has high quantity and quality measures for this study's participants.

On the high quantity – high quality side, the interactions focused on having a faculty member attend meetings and actively participate in them had higher composite scores than most other interactions yet the means were within the control limits. Table 11 illustrates the composite score means for the overall occurrence of SFIs for the participants.

TABLE 11.	Composite Score Mean	ns for Overall SF	FIs – Descending Order

Student-Faculty Interaction	n	Mean (sd)	Outside 95% Confidence Level
Had a faculty member attend meetings for a group you were affiliated with	70	14.56 (7.39)	No
Had a faculty member actively participate in meetings for a group you were affiliated with	67	14.28 (7.69)	No
Had a faculty member supervise a student social function (e.g., party) that you attended	45	11.07 (6.42)	No
Had a faculty member speak at a presentation or workshop sponsored by a group you were affiliated with	57	11.07 (6.45)	No
Had coffee, sodas, snacks, or meals with a professor (either in on-campus dining locations or off-campus at restaurants)	59	11.97 (7.36)	No
Met informally with faculty to get basic information and advice about my academic program	82	11.35 (6.81)	No
Asked a faculty member for information or guidance related to a course you were taking (grades, make-up work, assignments, etc.)	93	11.43 (7.01)	No
Asked a faculty member for comments or criticisms about your work (tests, papers, etc.)	93 90	11.43 (7.01)	No
Met informally with faculty to discuss matters related to my future career such as career plans and ambitions	84	11.50 (7.32)	No
Worked with a faculty member outside of class on a committee or project sponsored by a group you were affiliated with	26	12.04 (8.14)	No
Discussed ideas for a term paper or other class project with a faculty member	20 76	12.04 (8.14)	No
Had a faculty member assist in organizing a presentation or workshop sponsored by a group you were affiliated with	44	8.73 (5.48)	No
Worked with a faculty member outside of class on a committee or project	44 22	9.45 (6.41)	No
Had a faculty member accompany you to an athletic competition or event (this could include other people as well)			
Had a faculty member advise or supervise you on a faculty research project	10 37	11.00 (8.10) 10.95 (8.31)	No No
Had a faculty member advise or supervise you on a student research project (e.g., Honors thesis, independent study, etc.)	46	10.57 (8.03)	No
Had a faculty member help organize or attend a field trip with a group you were affiliated with	40 39	7.44 (5.06)	No
Talked informally (and outside of class) with an instructor about current events, campus activities, or other common interests	65	9.25 (6.98)	No
Was a guest in a professor's home for a meal or social function	26	7.31 (5.24)	No
Met informally with faculty to discuss a campus issue or problem	39	6.69 (5.15)	No
Personally invited a faculty member to speak at an event sponsored by a group you were affiliated with	33	7.82 (6.32)	No
Assisted a faculty member in teaching a class	33 20	9.50 (8.10)	No
Had a faculty member attend an arts or cultural event with you (this could include other people as well)	17	8.18 (7.03)	No
Met informally with faculty to help resolve a personal problem	65	3.86 (6.27)	Yes

In order to conduct comparisons and provide additional description, the responses were re-categorized into low composite (less than 9) and high composite (9 or more) interactions. Table 12 demonstrates that eleven interactions could be classified as majority low composite, with one interaction that was evenly split between low and high composite scores. Table 13 depicts the findings that twelve interactions could be classified as classified as majority high composite.

Based on the results, students are obtaining high composite interactions with faculty during organizational meetings and events, personal meetings centered on academic and career issues, and socially-based interactions. The low composite interactions centered on infrequent activities such as research projects, teaching courses, and attending athletic events. However, the interactions also included activities previously explored by other researchers such as discussing current events, personal issues, and going to a faculty member's home for an event.

s a guest in a professor's home for a meal or social function d a faculty member help organize or attend a field trip with a group you were affiliated with t informally with faculty to discuss a campus issue or problem trked with a faculty member outside of class on a committee or project sonally invited a faculty member to speak at an event sponsored by a group you were affiliated with d a faculty member advise or supervise you on a faculty research project sisted a faculty member in teaching a class d a faculty member advise or supervise you on a student research project (e.g., Honors thesis, ependent study, etc.) ked informally (and outside of class) with an instructor about current events, campus activities, or other mono interests d a faculty member accompany you to an athletic competition or event (this could include other people as 1) d a faculty member assist in organizing a presentation or workshop sponsored by a group you were liated with d coffee, sodas, snacks, or meals with a professor (either in on-campus dining locations or off-campus at taurants) cussed ideas for a term paper or other class project with a faculty member wheth a faculty member for comments or criticisms about your work (tests, papers, etc.) d a faculty member for comments or criticisms about your work (tests, papers, etc.) d a faculty member speak at a presentation or workshop sponsored by a group you were liated with ked a faculty member for information or guidance related to a course you were affiliated with ked a faculty member for information or guidance related to a course you were taking (grades, make-up rk, assignments, etc.) d a faculty member supervise a student social function (e.g., party) that you attended t informally with faculty to get basic information and advice about my academic program d a faculty member actively participate in meetings for a group you were affiliated with	n	Low Composite: Less Than 9
Had a faculty member attend an arts or cultural event with you (this could include other people as well)	17	82%
Was a guest in a professor's home for a meal or social function	26	69%
Had a faculty member help organize or attend a field trip with a group you were affiliated with	39	67%
Met informally with faculty to discuss a campus issue or problem	39	62%
Met informally with faculty to help resolve a personal problem	36	61%
Worked with a faculty member outside of class on a committee or project	22	59%
Personally invited a faculty member to speak at an event sponsored by a group you were affiliated with		
Had a faculty member advise or supervise you on a faculty research project	33 37	58% 57%
Assisted a faculty member in teaching a class	20	55%
Had a faculty member advise or supervise you on a student research project (e.g., Honors thesis,	20	5570
independent study, etc.)	46	52%
Talked informally (and outside of class) with an instructor about current events, campus activities, or other common interests	65	51%
Had a faculty member accompany you to an athletic competition or event (this could include other people as	05	5170
well)	10	50%
Had a faculty member assist in organizing a presentation or workshop sponsored by a group you were affiliated with	44	48%
Had coffee, sodas, snacks, or meals with a professor (either in on-campus dining locations or off-campus at		1070
restaurants)	59	44%
Discussed ideas for a term paper or other class project with a faculty member	76	42%
Worked with a faculty member outside of class on a committee or project sponsored by a group you were affiliated with	•	10.01
Asked a faculty member for comments or criticisms about your work (tests papers etc.)	26	42%
Asked a faculty memory for comments of efficising about your work (lesis, papers, etc.)	90	41%
Had a faculty member speak at a presentation or workshop sponsored by a group you were affiliated with	57	40%
Asked a faculty member for information or guidance related to a course you were taking (grades, make-up work, assignments, etc.)		
Ind a faculty member supervise a student assis function (a.e., north) that you attended	93	39%
riad a factory member supervise a student social function (e.g., party) that you attended	45	38%
Met informally with faculty to discuss matters related to my future career such as career plans and ambitions	84	36%
Met informally with faculty to get basic information and advice about my academic program	82	34%
Had a faculty member actively participate in meetings for a group you were affiliated with	67	31%
Had a faculty member attend meetings for a group you were affiliated with	70	27%

TARIE 12 Combined	Quantity and O	Juglity of SEIs for Study Darticipants	- Low Composite - Descending Order
IADLE 12. Combined	Juanuty and O	Juanty of SI'18 for Study Farthenbands -	- LOW COMPOSITE $-$ Descending Order

Student-Faculty Interaction	n	High Composite: 9 or More
Had a faculty member attend meetings for a group you were affiliated with	70	73%
Had a faculty member actively participate in meetings for a group you were affiliated with	67	69%
Met informally with faculty to get basic information and advice about my academic program	82	66%
Met informally with faculty to discuss matters related to my future career such as career plans and ambitions	0.4	6404
Had a faculty member supervise a student social function (e.g., party) that you attended	84 45	64% 62%
Asked a faculty member for information or guidance related to a course you were taking (grades, make-up		
work, assignments, etc.)	93	61%
Had a faculty member speak at a presentation or workshop sponsored by a group you were affiliated with	57	60%
Asked a faculty member for comments or criticisms about your work (tests, papers, etc.)	90	59%
Discussed ideas for a term paper or other class project with a faculty member	76	58%
Worked with a faculty member outside of class on a committee or project sponsored by a group you were affiliated with	26	58%
Had coffee, sodas, snacks, or meals with a professor (either in on-campus dining locations or off-campus at restaurants)	59	56%
Had a faculty member assist in organizing a presentation or workshop sponsored by a group you were affiliated with	44	52%
Had a faculty member accompany you to an athletic competition or event (this could include other people as well)	10	50%
Talked informally (and outside of class) with an instructor about current events, campus activities, or other common interests	65	49%
Had a faculty member advise or supervise you on a student research project (e.g., Honors thesis,		
independent study, etc.)	46	48%
Assisted a faculty member in teaching a class	20	45%
Had a faculty member advise or supervise you on a faculty research project	37	43%
Personally invited a faculty member to speak at an event sponsored by a group you were affiliated with	33	42%
Worked with a faculty member outside of class on a committee or project	22	41%
Met informally with faculty to discuss a campus issue or problem	39	39%
Met informally with faculty to help resolve a personal problem	36	39%
Had a faculty member help organize or attend a field trip with a group you were affiliated with	39	33%
Was a guest in a professor's home for a meal or social function	26	31%
Had a faculty member attend an arts or cultural event with you (this could include other people as well)	17	18%

TABLE 13. Combined Quantity and Quality of SFIs for Study Participants – High Composite – Descending Order

Following the re-categorization of the composite scores, a chi-square analysis was conducted to determine if differences existed between the ASO environment and other college activities. The athletic events interaction and the teaching class interaction again had to be dropped from analysis due to small cell size. For the rest of the interactions, comparisons were made and can be viewed in Table 14. Only one significant result was discovered and involved students working with faculty members on a student research project. Students were more likely to have a high composite score if they obtained the interaction through their ASO.

While most of the interactions did not differ significantly between contexts, there was one pattern worth noting. In eleven interactions, the participants who experienced the interactions in an ASO environment had a greater portion of high composite interactions than did participants who experienced the interactions in their other college activities. These interactions included student and faculty research projects, seeking advice on academic matters, attending and participating in organizational meetings and events, and discussing personal issues or campus events. In three interactions, the pattern was reversed and students tended to have a greater portion of high composite interactions if they occurred because of other collegiate activities. These interactions included working with faculty on a committee or project (within an organizational context and in a general context) and being a guest in a faculty member's home.

	ASOs		Other College Activities				
Student-Faculty Interaction	Low Composite	High Composite	n	Low Composite	High Composite	n	χ^2
Asked a faculty member for comments or criticisms about your work (tests, papers, etc.)	25%	75%	12	43%	57%	81	1.43
Asked a faculty member for information or guidance related to a course you were taking (grades, make-up							
work, assignments, etc.)	18%	82%	17	41%	59%	81	3.22
Assisted a faculty member in teaching a class	50%	50%	4	36%	64%	11	***
Discussed ideas for a term paper or other class project with a faculty member Had a faculty member advise or supervise you on a	33%	67%	9	42%	59%	65	.22
faculty research project Had a faculty member advise or supervise you on a	42%	58%	12	58%	42%	24	.89
student research project (e.g., Honors thesis, independent study, etc.) Had coffee, sodas, snacks, or meals with a professor	13%	88%	8	53%	47%	36	4.28*
(either in on-campus dining locations or off-campus at restaurants)	30%	70%	40	43%	57%	35	1.34
Met informally with faculty to discuss a campus issue or problem Met informally with faculty to discuss matters	56%	44%	18	75%	25%	20	1.59
related to my future career such as career plans and ambitions	35%	65%	34	32%	68%	60	.13
Met informally with faculty to get basic information and advice about my academic program Met informally with faculty to help resolve a	26%	74%	27	30%	70%	63	.17
personal problem Talked informally (and outside of class) with an	40%	60%	5	57%	43%	28	.50
instructor about current events, campus activities, or other common interests Was a guest in a professor's home for a meal or	41%	59%	32	50%	50%	50	.69
social function Worked with a faculty member outside of class on a	67%	33%	12	50%	50%	14	.74
committee or project Had a faculty member accompany you to an athletic competition or event (this could include other people	67%	33%	9	43%	57%	7	.91
as well) Had a faculty member attend an arts or cultural	50%	50%	2	25%	75%	4	***
event with you (this could include other people as well) Had a faculty member supervise a student social	68%	33%	6	71%	29%	7	.03
function (e.g., party) that you attended Personally invited a faculty member to speak at an	30%	70%	27	38%	63%	16	.28
event sponsored by a group you were affiliated with	50%	50%	22	63%	38%	8	.37
Had a faculty member help organize or attend a field rip with a group you were affiliated with Had a faculty member assist in organizing a	53%	47%	15	50%	50%	8	.02
presentation or workshop sponsored by a group you were affiliated with Had a faculty member speak at a presentation or	37%	63%	30	33%	67%	9	.03
workshop sponsored by a group you were affiliated with	29%	71%	41	46%	55%	11	1.03
Had a faculty member attend meetings for a group you were affiliated with	21%	79%	53	47%	53%	15	4.03
Had a faculty member actively participate in meetings for a group you were affiliated with Worked with a faculty member outside of class on a	27%	73%	55	46%	55%	11	1.44
committee or project sponsored by a group you were							

df=1, * =p<0.05; ** =p<0.01, ***=Could Not Compute

Based on the results, generating a composite score from the quantity and quality features of SFIs may need to be explored further as a way to measure meaningful interactions. Some of the interactions promoted by previous researchers (See Appendix A) as powerful were included in the high composite category such as meeting with faculty to discuss academic and career topics and interacting with them in a social setting. However, other previously cited powerful interactions (See Appendix A) fell into the low composite category such as working on research projects together, visiting a faculty member's home, and discussing personal issues with a faculty member. Further exploration may help determine why these interactions did not end up in the high composite category if they were previously defined as powerful.

The results of this study also revealed the possibility that some SFIs that were not addressed by early researchers may offer opportunities for meaningful interactions. For example, attending and participating in organizational meetings was researched for the first time in 2004 (Einarson & Clarkberg, 2004), so not much is known regarding what kind of benefits this setting can provide. However, it appears from this study that interactions occurring within the context of organizations, especially ASOs, may increase the likelihood that students and faculty will engage in meaningful interactions. This may be that the ASO context provides a comfortable environment that is based on shared interests, all of which are necessary to build powerful relationships (Cotten & Wilson, 2006; Endo & Harpel, 1982). Further research of these contexts could be beneficial to understanding what happens on a modern campus and where students and faculty members could naturally interact in the future.

Research Question 5

The fifth research question asked "what are the differences between classifications for the quantity and quality of student-faculty interactions within academic student organizations and how do they compare to those that occur in other settings?" However, the data were difficult to analyze because of the low number of freshmen and sophomores who participated in the study. The low participation rate by underclassmen may be a recurring challenge for researchers trying to study ASOs. Existing literature explains that upperclassmen may naturally gravitate to activities that provide additional career development opportunities whereas underclassmen are content to explore other activities (Alderman, 2008; Cotten & Wilson, 2006; Holzweiss, et al., 2008; Terenzini, et al., 1984a). It may be a natural consequence of conducting research on ASOs that participants will be limited to junior and senior classifications.

While most analysis could not be conducted due to the low participation rate by underclassmen, an attempt was made to explore classification further by comparing underclassmen (freshmen and sophomores) and upperclassmen (juniors and seniors) regarding the frequency of SFIs in the ASO and other college activities contexts. A chisquare analysis was used to compare the groups and can be viewed in Table 15. No significant differences were discovered between classification groups although it does appear that upperclassmen are trending towards more SFIs in the ASO context than underclassmen.

TABLE 15.	Frequency	of SFIs by	Classification
-----------	-----------	------------	----------------

Type of Student-Faculty Interaction	Underclassmen	n	Upperclassmen	n	χ^2
Experienced any SFI	100%	13	99%	90	.15
Experienced any SFI in the ASO context	69%	13	83%	90	1.50
Experienced any SFI in the Other College Activities					
context Experienced SFIs in both the ASO context and the	100%	13	96%	90	.60
Other College Activities context	69%	13	80%	90	.78

df=1, * =p<0.05; ** =p<0.01

In addition to the overall SFI frequencies, a chi-square analysis was used to compare frequency differences between the two contextual environments. Cell size was an issue for some of the interactions. In addition to attending athletic events and teaching class, two more interactions had to be dropped from analysis because of small cell sizes. These interactions were attending arts or cultural events and working on a committee or project sponsored by a group with which they were affiliated. For the rest of the interactions, comparisons were made and can be viewed in Table 16. No significant differences were discovered between classification groups and no patterns were observed.

Student-Faculty Interaction	ASOs Underclass men	Upperclass men	n	Other College Underclass men	e Activities Upperclass men	n	χ ²
Asked a faculty member for comments or criticisms about your work (tests, papers, etc.)	0%	100%	12	14%	87%	89	1.84
Asked a faculty member for information or guidance related to a course you were taking (grades, make-up work, assignments, etc.)	6%	94%	17	13%	88%	88	.62
Assisted a faculty member in teaching a class	0%	100%	4	8%	92%	13	***
Discussed ideas for a term paper or other class project with a faculty member	10%	90%	10	10%	91%	74	.00
Had a faculty member advise or supervise you on a faculty research project	8%	92%	13	6%	94%	33	.04
Had a faculty member advise or supervise you on a student research project (e.g., Honors thesis, independent study, etc.) Had coffee, sodas, snacks, or meals with a professor	0%	100%	9	5%	95%	43	.44
(either in on-campus dining locations or off-campus at restaurants)	2%	98%	47	8%	92%	40	1.42
Met informally with faculty to discuss a campus issue or problem Met informally with faculty to discuss matters	0%	100%	22	10%	90%	30	2.34
related to my future career such as career plans and ambitions Met informally with faculty to get basic information	11%	90%	38	13%	87%	67	.19
and advice about my academic program Met informally with faculty to help resolve a	10%	90%	30	11%	89%	72	.03
personal problem Talked informally (and outside of class) with an	0%	100%	5	3%	97%	33	.16
instructor about current events, campus activities, or other common interests Was a guest in a professor's home for a meal or	8%	92%	38	9%	91%	58	.02
social function Worked with a faculty member outside of class on a	14%	86%	14	6%	94%	16	.54
committee or project Had a faculty member accompany you to an athletic	13%	87%	15	0%	100%	11	1.59
competition or event (this could include other people as well) Had a faculty member attend an arts or cultural event with you (this could include other people as	0%	100%	2	0%	100%	7	***
well) Had a faculty member supervise a student social	0%	100%	9	0%	100%	13	***
function (e.g., party) that you attended Personally invited a faculty member to speak at an	6%	94%	33	8%	92%	24	.11
event sponsored by a group you were affiliated with	3%	97%	29	0%	100%	9	.32
Had a faculty member help organize or attend a field trip with a group you were affiliated with Had a faculty member assist in organizing a presentation or workshop sponsored by a group you	13%	88%	32	0%	100%	17	2.31
were affiliated with Had a faculty member speak at a presentation or workshop sponsored by a group you were affiliated	5%	95%	38	0%	100%	14	.77
with Had a faculty member attend meetings for a group	7%	93%	43	18%	82%	11	1.31
you were affiliated with Had a faculty member actively participate in	8%	92%	65	17%	83%	18	1.30
meetings for a group you were affiliated with Worked with a faculty member outside of class on a	7%	93%	68	20%	80%	15	2.26
committee or project sponsored by a group you were affiliated with	0%	100%	19	0%	100%	8	***

TABLE 16. Occurrence of SFIs in ASOs and Other College Activities by Classification

df=1, * =p<0.05; ** =p<0.01, ***=Could Not Compute

It is difficult to identify any trends for classification due to the low participation rate of underclassmen. This study's results suggest that there are no differences between underclassmen and upperclassmen in terms of the frequency of interactions they may have with faculty in the different contextual environments. However, a stronger response rate from all classifications is needed before any patterns can be identified.

Research Question 6

The sixth research question asked "does the quantity and quality of studentfaculty interactions in academic student organizations differ by institutional size and how do they compare to those that occur in other settings?" The first analysis examined differences between institutions for the overall frequency of SFIs that participants experienced in various contextual environments. Table 17 reveals no significant differences between small and large colleges although it is interesting to note that small colleges appeared to have more SFIs in the ASO context than large colleges. This supports the existing literature that finds small colleges may have more opportunities for informal interactions (Kezar & Kinzie, 2006; Pike & Kuh, 2005).

Type of Student-Faculty Interaction	Small College	n	Large College	n	χ^2
Experienced any SFI	98%	45	100%	58	1.30
Experienced any SFI in the ASO context					
	88%	45	76%	58	2.86
Experienced any SFI in the Other College Activities					
context	93%	45	98%	58	1.66
Experienced SFIs in both the ASO context and the Other College Activities context	84%	45	74%	58	1.60
$df_{-1} * -n < 0.05$; $** -n < 0.01$					

TABLE 17. Frequency of SFIs by Institutional Size

df=1, * =p<0.05; ** =p<0.01

Following the overall analysis of interactions experienced, each interaction was then analyzed for differences using the chi-square statistic. The interactions dealing with athletic events and teaching a class could not be computed due to small cell sizes. Table 18 illustrates that one significant difference was observed (Asking a faculty member for guidance, $p \le .05$). For the interaction centering on asking a faculty member for advice related to an academic course, participants attending small colleges were more likely to experience the interaction in the ASO context while participants attending large colleges were more likely to experience the interaction through their other college activities. While not statistically significant, the same pattern was observed for seven more interactions including working on research projects; discussing assignments, campus issues, academic concerns, and career plans; and working on a committee or project sponsored by a group with which they are affiliated.

In addition to this pattern, institutional size appeared to have some influence on other interactions. Participants from small colleges had a greater proportion of ASO interactions in the areas of being a guest in a faculty member's home and discussing personal problems. In contrast, participants from large colleges had a greater proportion of ASO interactions in the areas of working with faculty on a committee or project, inviting a faculty member to speak at an event, and hearing a faculty member speak at a presentation or workshop. TABLE 18. Occurrence of SFIs for ASOs and Other College Activities by Institutional Size

	ASOs				ge Activities		
Student-Faculty Interaction	Small College	Large College	n	Small College	Large College	n	χ^2
Asked a faculty member for comments or criticisms about your work (tests, papers, etc.)	42%	58%	12	42%	58%	89	.00
Asked a faculty member for information or guidance related to a course you were taking (grades, make-up							
work, assignments, etc.)	71%	29%	17	39%	61%	88	5.91*
Assisted a faculty member in teaching a class	100%	0%	4	54%	46%	13	***
Discussed ideas for a term paper or other class project with a faculty member Had a faculty member advise or supervise you on a	60%	40%	10	47%	53%	74	.59
aculty research project Had a faculty member advise or supervise you on a	62%	39%	13	46%	55%	33	.97
tudent research project (e.g., Honors thesis, ndependent study, etc.) Had coffee, sodas, snacks, or meals with a professor	78%	22%	9	49%	51%	43	2.51
either in on-campus dining locations or off-campus t restaurants)	47%	53%	47	45%	55%	40	.03
Aet informally with faculty to discuss a campus ssue or problem Aet informally with faculty to discuss matters	64%	36%	22	47%	53%	30	1.47
elated to my future career such as career plans and mbitions Met informally with faculty to get basic information	58%	42%	38	40%	60%	67	3.02
nd advice about my academic program	57%	43%	30	42%	58%	72	1.92
Aet informally with faculty to help resolve a ersonal problem 'alked informally (and outside of class) with an	80%	20%	5	55%	46%	33	1.15
nstructor about current events, campus activities, or ther common interests Vas a guest in a professor's home for a meal or	47%	53%	38	48%	52%	58	.01
ocial function	79%	21%	14	56%	44%	16	1.67
Vorked with a faculty member outside of class on a ommittee or project Iad a faculty member accompany you to an athletic	47%	53%	15	36%	64%	11	.28
ompetition or event (this could include other people s well) Had a faculty member attend an arts or cultural	100%	0%	2	71%	29%	7	***
vent with you (this could include other people as yell) Iad a faculty member supervise a student social	33%	67%	9	69%	31%	13	2.76
unction (e.g., party) that you attended	46%	55%	33	50%	50%	24	.12
ersonally invited a faculty member to speak at an vent sponsored by a group you were affiliated with	45%	55%	29	33%	67%	9	.37
Iad a faculty member help organize or attend a field ip with a group you were affiliated with Iad a faculty member assist in organizing a	41%	59%	32	59%	41%	17	1.48
resentation or workshop sponsored by a group you vere affiliated with Iad a faculty member speak at a presentation or	47%	53%	38	57%	43%	14	.39
vorkshop sponsored by a group you were affiliated vith	49%	51%	43	36%	64%	11	.55
Iad a faculty member attend meetings for a group ou were affiliated with	48%	52%	65	50%	50%	18	.03
Iad a faculty member actively participate in neetings for a group you were affiliated with	50%	50%	68	53%	47%	15	.06
Vorked with a faculty member outside of class on a ommittee or project sponsored by a group you were	68%	32%			63%		

df=1, * =p<0.05; ** =p<0.01, ***=Could Not Compute

In the next step of analysis, the composite scores for each interaction were compared to determine whether or not differences existed based on institutional size. Table 19 contains the results for interactions that occurred within the ASO context. Cell size was a problem for eight interactions and those could not be compared. These interactions included teaching a class, attending athletic and cultural events, being a guest in a faculty member's home, working on a committee or project, working on a student research project, and discussing personal issues or ideas for a class project.

For the remaining interactions, one significant difference was observed. Participants from small colleges were more likely to have a high composite score when the interaction involved discussing current events. Participants from large colleges were more likely to have a low composite score for this interaction. In most other interactions, it was common for participants from both types of institutions to have high composite scores from their ASO-based interactions.

Comparisons were then conducted for interactions occurring because of other college activities. The results can be found in Table 20, which immediately follows Table 19. Cell size was a problem for eight interactions but some of those interactions differed from those identified for the ASO context. Those focused on activities within a group context such as attending a field trip, organizing a presentation for the group, having a faculty member speak at a group event, and working with a faculty member on a committee or project sponsored by a group. For the interactions that could be compared, no significant differences were found between small and large institutions and no discernible patterns were observed.

	Small College			Large College			
Student-Faculty Interaction	Low Composite	High Composite	n	Low Composite	High Composite	n	χ^2
Asked a faculty member for comments or criticisms about your work (tests, papers, etc.)	20%	80%	5	29%	71%	7	.11
Asked a faculty member for information or guidance related to a course you were taking (grades, make-up work, assignments, etc.)	170/	920/	10	2004	000/	-	02
Assisted a faculty member in teaching a class	17%	83%	12	20%	80%	5	.03
Discussed ideas for a term paper or other class project	50%	50%	4	0%	0%	0	***
with a faculty member Had a faculty member advise or supervise you on a	33%	67%	6	33%	675	3	***
faculty research project Had a faculty member advise or supervise you on a student research project (e.g., Honors thesis,	29%	71%	7	60%	40%	5	1.19
independent study, etc.) Had coffee, sodas, snacks, or meals with a professor	17%	83%	6	0%	100%	2	***
(either in on-campus dining locations or off-campus at restaurants)	30%	70%	30	30%	70%	20	.00
Met informally with faculty to discuss a campus issue or problem	42%	58%	12	83%	17%	6	2.81
Met informally with faculty to discuss matters related to my future career such as career plans and ambitions	21%	79%	19	53%	47%	15	3.83
Met informally with faculty to get basic information and advice about my academic program	27%	73%	15	25%	75%	12	.01
Met informally with faculty to help resolve a personal problem Talked informally (and outside of class) with an	50%	50%	2	0%	100%	1	***
instructor about current events, campus activities, or other common interests Was a guest in a professor's home for a meal or social	19%	81%	16	63%	38%	16	6.35
function Worked with a faculty member outside of class on a	60%	40%	10	100%	0%	2	***
committee or project Had a faculty member accompany you to an athletic competition or event (this could include other people as	50%	50%	4	80%	20%	5	***
well)	50%	50%	2	0%	0%	0	***
Had a faculty member attend an arts or cultural event with you (this could include other people as well)	50%	50%	2	75%	25%	4	***
Had a faculty member supervise a student social function (e.g., party) that you attended	25%	75%	12	33%	67%	15	.22
Personally invited a faculty member to speak at an event sponsored by a group you were affiliated with	36%	64%	11	64%	36%	11	1.64
Had a faculty member help organize or attend a field trip with a group you were affiliated with Had a faculty member assist in organizing a	50%	50%	6	56%	44%	9	.05
presentation or workshop sponsored by a group you were affiliated with Had a faculty member speak at a presentation or	33%	67%	15	40%	60%	15	.14
workshop sponsored by a group you were affiliated with	32%	68%	19	27%	73%	22	.09
Had a faculty member attend meetings for a group you were affiliated with	17%	83%	23	23%	77%	30	.28
Had a faculty member actively participate in meetings for a group you were affiliated with	19%	82%	27	36%	64%	28	2.05
Worked with a faculty member outside of class on a committee or project sponsored by a group you were							

TABLE 19. Combined Quantity and Quality of SFIs for ASOs by Institutional Size

df=1, * =p<0.05; ** =p<0.01, ***=Could Not Compute

	Small Colleg	-	Large College				
Student-Faculty Interaction	Low Composite	High Composite	n	Low Composite	High Composite	n	χ^2
Asked a faculty member for comments or criticisms about your work (tests, papers, etc.)	34%	66%	32	49%	51%	49	1.68
Asked a faculty member for information or guidance related to a course you were taking (grades, make-up							
work, assignments, etc.)	39%	61%	31	43%	57%	49	.14
Assisted a faculty member in teaching a class	40%	60%	5	33%	67%	6	.05
Discussed ideas for a term paper or other class project with a faculty member	38%	62%	29	43%	57%	35	.16
Had a faculty member advise or supervise you on a faculty research project	67%	33%	9	53%	47%	15	.41
Had a faculty member advise or supervise you on a student research project (e.g., Honors thesis, independent study, etc.) Had coffee, sodas, snacks, or meals with a professor	59%	41%	17	47%	53%	19	.47
(either in on-campus dining locations or off-campus at restaurants)	47%	53%	15	40%	60%	20	.16
Met informally with faculty to discuss a campus issue or problem	73%	27%	11	78%	22%	9	.07
Met informally with faculty to discuss matters related to my future career such as career plans and ambitions	27%	73%	22	35%	65%	37	.39
Met informally with faculty to get basic information and advice about my academic program	23%	77%	26	33%	67%	36	.77
Met informally with faculty to help resolve a personal problem Falked informally (and outside of class) with an	64%	36%	14	46%	54%	13	.90
nstructor about current events, campus activities, or other common interests Was a guest in a professor's home for a meal or social	46%	54%	26	54%	46%	24	.32
function Worked with a faculty member outside of class on a	29%	71%	7	71%	29%	7	2.5
committee or project Had a faculty member accompany you to an athletic	0%	100%	1	50%	50%	6	**1
competition or event (this could include other people as well)	33%	67%	3	0%	100%	1	***
Had a faculty member attend an arts or cultural event with you (this could include other people as well)	67%	33%	3	75%	25%	4	**1
Had a faculty member supervise a student social function (e.g., party) that you attended	25%	75%	8	50%	50%	8	1.0
Personally invited a faculty member to speak at an event sponsored by a group you were affiliated with	100%	0%	2	50%	50%	6	***
Had a faculty member help organize or attend a field trip with a group you were affiliated with	67%	33%	6	0%	100%	2	***
Had a faculty member assist in organizing a presentation or workshop sponsored by a group you were affiliated with	40%	60%	5	25%	75%	4	***
Had a faculty member speak at a presentation or workshop sponsored by a group you were affiliated with						7	***
Had a faculty member attend meetings for a group you were affiliated with	25% 29%	75% 71%	4 7	57% 63%	43% 38%	7 8	1.7
Had a faculty member actively participate in meetings for a group you were affiliated with	33%	67%	6	60%	40%	5	.78
Worked with a faculty member outside of class on a committee or project sponsored by a group you were							
affiliated with df=1, * =p<0.05; ** =p<0.01, ***=Could Not Compute	0%	100%	1	25%	75%	4	***

TABLE 20. Combined Quantity and Quality of SFIs for Other College Activities by Institutional Size

One problem that plagued this set of analysis was small cell sizes. While this could be due to inadequate participation, it may be possible that certain types of SFIs are rare for undergraduates on any size campus such as teaching a class or attending athletic events with faculty. This problem might persist even if larger numbers of students were studied. Different types of analysis such as qualitative methods may need to be considered to help identify the true nature of these interactions and how institutional size impacts those interactions.

In terms of the actual analysis that could be conducted, very few statistical differences were found between small and large institutions but there seemed to be a pattern of some interactions, especially those within the ASO context, occurring more in the small college environment. The literature does support the idea that small college environments may offer more opportunities than large colleges for students and faculty to come together in informal activities (Alderman, 2008; Dilley, 1967; Tinto, 1975). It does appear that the findings from this study would support that conclusion.

One final observation is that many interactions occurring in the ASO context resulted in high composite scores for both small and large institutions. It may be that when interactions occur within ASOs, they occur in higher quantity and quality regardless of the size of the institution. More research is needed to determine the impact of interactions within ASOs and what characteristics may be present to yield higher quantity and/or quality.

Research Question 7

The seventh, and final, research question asked "does the quantity and quality of student-faculty interactions in academic student organizations differ by student role (i.e., member versus leader) and how do they compare to those that occur in other settings?" The first set of analysis compared the overall occurrence of SFIs as well as where the interactions occurred since participants started college. Two significant differences were observed. Table 21 demonstrates that while members and leaders had a similar overall frequency of SFIs, leaders were more likely to experience SFIs in the context of their ASO than members. Leaders also were more likely than members to experience SFIs in both the ASO context and within their other college activities.

Type of Student-Faculty Interaction	Member	n	Leader	n	χ^2
Experienced any SFI	100%	61	98%	43	1.43
Experienced any SFI in the ASO context					
	71%	61	95%	43	10.03**
Experienced any SFI in the Other College Activities context	97%	61	95%	43	.13
Experienced SFIs in both the ASO context and the Other College Activities context	67%	61	93%	43	9.75**

TABLE 21. Frequency of SFIs by Organizational Role

df=1, * =p<0.05; ** =p<0.01

The second set of analysis explored differences for each interaction between members and leaders in the ASO context and within other college activities. Two interactions could not be compared due to small cell sizes – attending athletic events and teaching a class. For the rest of the interactions, three significant differences were observed and can be viewed in Table 22. The three interactions were having a faculty member speak at a presentation or workshop sponsored by a group, asking a faculty member for comments or criticism about coursework, and discussing campus issues with a faculty member. In all three cases, leaders were more likely to experience the interactions within their ASOs while members were more likely to experience the interactions through other college activities. In addition to the significant differences, the same pattern continued with other interactions. For the ASO environment, leaders experienced twenty interactions in greater frequency than members. For other college activities, members experienced sixteen interactions in greater frequency than leaders.

An attempt was made to compare the composite scores for members and leaders in both the ASO context and within other college activities. Several interactions in each context could not be compared due to small cell sizes. However, for those that could be compared, a few significant differences were observed. Tables 23 and 24, which immediately follow Table 22, contain the responses. For the ASO context, leaders were more likely than members to have a high composite score for the interactions of obtaining advice for their academic program and having drinks or meals with a faculty member. Similarly, in other college activities, leaders were more likely than members to have a high composite score for the interactions of seeking guidance for a course, discussing ideas for class projects, and having drinks or meals with a faculty member. Overall, leaders had more interactions that rated in the high composite category than did members in both the ASO environment and through other college activities. TABLE 22. Occurrence of SFIs for ASOs and Other College Activities by Organizational Role

	ASOs			Other College Activities					
Student-Faculty Interaction	Member	Leader	n	Member	Leader	n	χ^2		
Asked a faculty member for comments or criticisms									
about your work (tests, papers, etc.)	25%	75%	12	58%	42%	90	4.58*		
Asked a faculty member for information or guidance related to a course you were taking (grades, make-up									
work, assignments, etc.)	41%	59%	17	60%	40%	89	1.96		
Assisted a faculty member in teaching a class							***		
Discussed ideas for a term paper or other class	25%	75%	4	46%	54%	13	~ ~ ~		
project with a faculty member Had a faculty member advise or supervise you on a	40%	60%	10	55%	45%	75	.76		
faculty research project	54%	46%	13	53%	47%	34	.00		
Had a faculty member advise or supervise you on a student research project (e.g., Honors thesis, independent study, etc.)	44%	56%	9	47%	54%	43	.01		
Had coffee, sodas, snacks, or meals with a professor (either in on-campus dining locations or off-campus									
at restaurants)	34%	66%	47	45%	55%	40	1.09		
Met informally with faculty to discuss a campus issue or problem	14%	86%	22	55%	45%	31	9.30**		
Met informally with faculty to discuss matters related to my future career such as career plans and									
ambitions	45%	55%	38	57%	43%	68	1.56		
Met informally with faculty to get basic information and advice about my academic program	50%	50%	30	58%	43%	73	.49		
Met informally with faculty to help resolve a				58%					
personal problem Talked informally (and outside of class) with an	20%	80%	5	53%	47%	34	1.89		
instructor about current events, campus activities, or	250/	(20)	20	500/	100/	50	2.20		
other common interests Was a guest in a professor's home for a meal or	37%	63%	38	53%	48%	59	2.29		
social function	29%	71%	14	44%	56%	16	.74		
Worked with a faculty member outside of class on a committee or project	33%	67%	15	55%	46%	11	1.17		
Had a faculty member accompany you to an athletic competition or event (this could include other people									
as well)	50%	50%	2	86%	14%	7	***		
Had a faculty member attend an arts or cultural event with you (this could include other people as									
well)	22%	78%	9	62%	39%	13	3.32		
Had a faculty member supervise a student social function (e.g., party) that you attended	39%	61%	33	54%	46%	24	1.22		
Personally invited a faculty member to speak at an	5270	01/0	55	5170	1070	2.	1.44		
event sponsored by a group you were affiliated with	24%	76%	29	56%	44%	9	3.14		
Had a faculty member help organize or attend a field									
trip with a group you were affiliated with Had a faculty member assist in organizing a	44%	56%	32	41%	59%	17	.03		
presentation or workshop sponsored by a group you were affiliated with	42%	58%	38	50%	50%	14	.26		
Had a faculty member speak at a presentation or	T2/0	5670	50	5070	5070	17	.20		
workshop sponsored by a group you were affiliated with	47%	54%	43	91%	9%	11	6.99**		
Had a faculty member attend meetings for a group									
you were affiliated with Had a faculty member actively participate in	49%	51%	65	61%	39%	18	.80		
meetings for a group you were affiliated with	49%	52%	68	67%	33%	15	1.62		
Worked with a faculty member outside of class on a committee or project sponsored by a group you were									
affiliated with	26%	74%	19	50%	50%	8	1.42		

df=1, * =p<0.05; ** =p<0.01, ***=Could Not Compute

TABLE 23. Combined Quantity and Quality of SFIs for ASOs by Organizational Role

Student-Faculty Interaction	Members Low Composite	High Composite	n	Leaders Low Composite	High Composite	n	χ^2
Asked a faculty member for comments or criticisms about your work (tests, papers, etc.)	67%	33%	3	11%	89%	9	***
Asked a faculty member for information or guidance related to a course you were taking (grades, make-up							
work, assignments, etc.)	29%	71%	7	10%	90%	10	.98
Assisted a faculty member in teaching a class	100%	0%	1	33%	67%	3	***
Discussed ideas for a term paper or other class project with a faculty member Had a faculty member advise or supervise you on a	50%	50%	4	20%	80%	5	***
faculty research project Had a faculty member advise or supervise you on a	67%	33%	6	17%	83%	6	3.09
student research project (e.g., Honors thesis, independent study, etc.) Had coffee, sodas, snacks, or meals with a professor	25%	75%	4	0%	100%	4	***
(either in on-campus dining locations or off-campus at restaurants) Met informally with faculty to discuss a campus issue	55%	46%	11	21%	79%	29	4.35
or problem	100%	0%	2	50%	50%	16	***
Met informally with faculty to discuss matters related to my future career such as career plans and ambitions	47%	53%	15	26%	74%	19	1.52
Met informally with faculty to get basic information and advice about my academic program Met informally with faculty to help resolve a personal	46%	54%	13	7%	93%	14	5.34
problem Talked informally (and outside of class) with an	100%	0%	1	25%	75%	4	***
instructor about current events, campus activities, or other common interests Was a guest in a professor's home for a meal or social	56%	44%	9	35%	65%	23	1.16
function Worked with a faculty member outside of class on a	50%	50%	2	70%	30%	10	***
committee or project Had a faculty member accompany you to an athletic	100%	0%	1	63%	38%	8	***
competition or event (this could include other people as well)	100%	0%	1	0%	100%	1	***
Had a faculty member attend an arts or cultural event with you (this could include other people as well)	50%	50%	2	75%	25%	4	***
Had a faculty member supervise a student social function (e.g., party) that you attended	33%	67%	9	28%	72%	18	.09
Personally invited a faculty member to speak at an event sponsored by a group you were affiliated with	33%	67%	3	53%	47%	19	***
Had a faculty member help organize or attend a field trip with a group you were affiliated with Had a faculty member assist in organizing a	33%	67%	3	58%	42%	12	***
presentation or workshop sponsored by a group you were affiliated with Had a faculty member speak at a presentation or	40%	60%	10	35%	65%	20	.07
workshop sponsored by a group you were affiliated with	19%	81%	16	36%	64%	25	1.40
Had a faculty member attend meetings for a group you were affiliated with	25%	75%	24	17%	83%	29	.48
Had a faculty member actively participate in meetings for a group you were affiliated with	36%	64%	25	20%	80%	30	1.76
Worked with a faculty member outside of class on a committee or project sponsored by a group you were							
affiliated with df=1, * =p<0.05; ** =p<0.01, ***=Could Not Compute	100%	0%	3	25%	75%	12	***

TABLE 24. Combined Quantity and Quality of SFIs for Other College Activities by Organizational Role

Student-Faculty Interaction	Members Low Composite	High Composite	n	Leaders Low Composite	High Composite	n	χ^2
Asked a faculty member for comments or criticisms about your work (tests, papers, etc.)	51%	49%	45	33%	67%	36	2.58
Asked a faculty member for information or guidance related to a course you were taking (grades, make-up work, assignments, etc.)	51%	49%	47	27%	74%	34	4.94*
Assisted a faculty member in teaching a class							
Discussed ideas for a term paper or other class project with	60%	40%	5	17%	83%	6	2.21
a faculty member Had a faculty member advise or supervise you on a faculty	55%	46%	33	28%	72%	32	4.67*
esearch project Had a faculty member advise or supervise you on a student esearch project (e.g., Honors thesis, independent study,	50%	50%	10	64%	36%	14	.49
etc.)	69%	31%	16	40%	60%	20	2.95
Had coffee, sodas, snacks, or meals with a professor (either n on-campus dining locations or off-campus at restaurants)	73%	27%	15	20%	80%	20	9.96**
Met informally with faculty to discuss a campus issue or problem	80%	20%	10	70%	30%	10	.27
Met informally with faculty to discuss matters related to my future career such as career plans and ambitions	39%	61%	33	22%	78%	27	2.02
Met informally with faculty to get basic information and advice about my academic program	39%	61%	33	20%	80%	30	2.81
Met informally with faculty to help resolve a personal problem Talked informally (and outside of class) with an instructor	50%	50%	12	63%	38%	16	.44
about current events, campus activities, or other common interests	52%	48%	23	48%	52%	27	.08
Was a guest in a professor's home for a meal or social function	60%	40%	5	44%	56%	9	.31
Worked with a faculty member outside of class on a committee or project Had a faculty member accompany you to an athletic	33%	67%	3	50%	50%	4	***
competition or event (this could include other people as well)	33%	67%	3	0%	100%	1	***
Had a faculty member attend an arts or cultural event with you (this could include other people as well)	80%	20%	5	50%	50%	2	***
Had a faculty member supervise a student social function (e.g., party) that you attended	29%	71%	7	44%	56%	9	.42
Personally invited a faculty member to speak at an event sponsored by a group you were affiliated with	75%	25%	4	50%	50%	4	***
Had a faculty member help organize or attend a field trip with a group you were affiliated with	50%	23% 50%	4	50%	50%	4 6	***
Had a faculty member assist in organizing a presentation or workshop sponsored by a group you were affiliated with	50%	50%	4	20%	80%	5	***
Had a faculty member speak at a presentation or workshop sponsored by a group you were affiliated with	57%	43%	7	25%	75%	4	***
Had a faculty member attend meetings for a group you were affiliated with	50%	50%	8	43%	57%	4	.08
Had a faculty member actively participate in meetings for a group you were affiliated with	33%	67%	6	60%	40%	5	.78
Worked with a faculty member outside of class on a committee or project sponsored by a group you were affiliated with	0%	100%	2	33%	67%	3	***

df=1, * =p<0.05; ** =p<0.01, ***=Could Not Compute

Based on the results of this study, it does appear that the role a student has within a student organization may impact the quantity and quality of interactions they have with faculty. This seems especially true for the ASO context, possibly because faculty members serve as advisors for these organizations. It may be that leaders, through their increased roles in their ASOs, have greater access to faculty than do members. Student leaders, by virtue of their positions and responsibilities, would have to interact with the faculty advisors to conduct organizational business. These business-related interactions may then develop into opportunities for forming deeper relationships with the faculty advisors or increase the prospects for interactions with other faculty. By contrast, members of ASOs may just attend meetings and events and never have a reason to interact with the faculty advisor or other faculty associated with the organization.

In addition to having greater access to faculty through their ASOs, student leaders may be participating in more forms of campus governance or may be invited to more functions and activities than members. These additional activities may give student leaders more access to faculty members than what is immediately available in their ASOs.

Limitations

There are a number of limitations to this study that should be noted. First, even though the study was conducted under the post-positivist paradigm, it was designed to be descriptive in nature rather than provide generalizable results. The results help identify what additional research should be conducted to further explore student-faculty interactions within the specific context of academic student organizations. It does not represent the experience of all students participating in these organizations and should not be interpreted as such.

Second, although the academic student organizations were selected based on similar missions and purposes, it is possible that the activities of the organizations differed by institution. These unknown differences may have impacted the findings that were not addressed by this study.

Third, because of the limited availability of similar student organizations on each campus, some pre-selected organizations were honorary in nature, meaning membership is only open to students who perform well academically. These students may be predisposed to pursuing interactions with faculty in order to further develop their careers. It is possible that their participation in the study could have impacted the results.

Fourth, the instrument developed for the study may not capture the wide variety of SFIs that occur on a college campus. It is a research tool that may need further exploration in order to determine its usefulness in measuring interactions between students and faculty.

Fifth, this study defined "quality" in terms of the time that faculty and students spent together during their interactions. There could be other measures that could contribute to the quality of interactions between students and faculty such as level of satisfaction and the importance placed on the interactions by both parties. Additional studies would need to be conducted to determine whether length of time is a strong individual indicator of quality interactions.

Sixth, this study did not explore several areas suggested by other researchers as important factors in SFIs such as classroom structure (Cotten & Wilson, 2006) and teaching styles (Cotten & Wilson, 2006; Jaasma & Koper, 1999). These factors may have contributed to the quality and/or quantity of interactions in this study but were not examined.

Seventh, this study only examined the student perspective of SFIs. The faculty perspective is also needed in order to fully understand how interactions with students within the context of academic organizations may benefit and hinder the work, expectations, and values of faculty.

Eighth, females participated in this study in greater numbers than did males. Therefore, the male experience may not be fully captured by the results of this study.

Ninth, there are additional types of interactions that could be occurring between students and faculty that have yet to be fully examined. For instance, one study found that predominantly commuter campuses located in urban areas had interactions that were mainly conducted through technological means such as email and discussion boards (Kezar & Kinzie, 2006). It could be that some students are replacing face-to-face interactions with faculty through the use of technology, which could have resulted in decreased reporting of face-to-face interactions in this study.

And finally, some of the interactions examined by this study were not experienced by many participants. The result was several sets of analysis that could not be fully conducted due to small cell sizes. While it could be assumed that more participants were needed to make these comparisons, it is unclear whether or not this was the cause of the small cell size issue. Other factors may be impacting this issue and should be explored further before concluding whether or not more participants would result in the kind of responses needed to conduct analysis. It may be that these interactions are not prevalent among college students regardless of institutional size, classification, and organizational role and should not be included in future studies on the topic.

Summary of Findings

The purpose of summarizing the findings of this descriptive study is to highlight the key points that require additional consideration. There are several findings that could impact future research, ranging from the demographic information collected to the analysis conducted for each research question.

For demographics, there were three important results to consider. First, almost all of the participants indicated that their academic student organization had a faculty member serving as their advisor. Second, at the beginning of the study, the issue of living-learning communities was a consideration for how students may be obtaining interactions with faculty. However, only a small proportion of participants in this study belonged to such communities and even fewer reported interacting with faculty as part of that community. Third, there were more juniors and seniors participating in the study than freshmen and sophomores. There is some indication in the literature that upperclassmen may be more likely to join academic student organizations than underclassmen. It may be that future research on academic student organizations needs to focus exclusively on upperclassmen. In terms of answering the first research question, virtually all of the participants in this study experienced at least one interaction with faculty since they started college. This supports previous research that states almost all students will have an interaction with faculty during their college career (Lewallen, 1995; Rosenthal, et al., 2000). This study did provide some evidence, however, that asking where these interactions are occurring is necessary to truly understand student-faculty relationships. While most of the interactions reported in this study occurred through activities not associated with their academic student organizations, a majority of the participants experienced at least one interaction with faculty within the context of their organization. More importantly, three-quarters reported having interactions with faculty in both their academic student organization and their other college activities.

When research does not recognize that these interactions are occurring in specific contexts, it misses a key component of the college experience. This study reinforced the idea that contextual environments need to be considered and researched in order to understand student-faculty interactions. It also may be necessary to consider cultural influences such as socio-economic status to fully understand how students are operating within their contextual environments. For instance, students from a lower socio-economic status may not have the cultural capital to understand that interacting with faculty members can provide benefits (Walpole, 2003). It may be necessary to guide these students into interactions by providing them with a supportive environment.

Another observation worth noting was that most of the interactions occurring within the context of academic student organizations may be social in nature rather than

academic. For instance, faculty and students may interact within organizational meetings but it is unclear what topics they are discussing during those meetings. They could be focused on collecting canned goods for a service project rather than academic or career-related issues, which has been demonstrated to help students advance their intellectual development (Cotten & Wilson, 2006; Pascarella & Terenzini, 1980a; Terenzini & Pascarella, 1980). More research is needed before any firm conclusions can be made but it is an issue worth further exploration.

In regards to the quantity of interactions, the findings of this study support the existing literature that student-faculty interactions occur but are infrequent overall (Anaya & Cole, 2001; Cotten & Wilson, 2006; Fusani, 1994; Hagedorn, et al., 2000; Jaasma & Koper, 1999; Nadler & Nadler, 2001). The quantity of interactions that did occur for participants did not differ significantly between the context of academic student organizations and other college activities. However, there was a trend for participants to have a higher quantity of interactions within the ASO context than through other college activities. This trend may be influenced by the faculty advisors available in the organizations.

In terms of quality, most of the interactions reported by participants lasted longer than 10 minutes, which was the standard by which some researchers measured quality. The results of this study contradict previous literature (Jaasma & Koper, 1999; Nadler & Nadler, 2001) that concluded interactions between students and faculty typically last less than 10 minutes. It may be that the interactions explored in this study allowed for a

wider scope of activities experienced on a modern campus and that further research into these interactions may result in a new norm for the length of interactions.

While there were some discernible patterns for quantity, no differences or trends could be detected for quality. It may be that defining quality through length of time spent in the interaction is not the best way to determine quality. A better way to define quality may be what this study defined as a composite score of the frequency of the interaction multiplied by the length of time spent in the interaction. When composite scores were computed for this study, several of the traditional interactions defined as powerful by existing literature (See Appendix A) emerged as powerful interactions for the participants (e.g., higher frequencies and longer lasting). It should be noted, though, that some interactions thought to be powerful by previous researchers (See Appendix A) were not classified as high composite interactions by this study.

For institutional size, there were a few significant differences and a pattern of responses that indicated participants from small institutions may have a higher quality and quantity of interactions with faculty than their peers from large institutions. This idea is supported in the existing literature (Alderman, 2008; Dilley, 1967; Tinto, 1975). However, participants from large institutions also were experiencing interactions with faculty in both the academic student organization context and within other college activities so it is important not to dismiss what is occurring in both types of institutions. Institutional size should be considered a factor worth exploring in future research until a better understanding can be developed of what may be different about the interactions provided within both types of institutions.

For the last research question, whether or not a student served as a member or a leader in their academic student organization did appear to impact the interactions they had with faculty. Leaders reported more interactions with faculty in their organizational context than did members. This may be a direct result of these organizations having faculty advisors. Leaders must interact with advisors on a regular basis in order to perform responsibilities for the organization. This also would provide them greater access to faculty than the members who may just attend a general meeting and a few events throughout the school year.

The final chapter of this dissertation will provide further discussion of these findings along with a detailed plan for researching the topic more thoroughly.

CHAPTER V

DISCUSSION AND CONCLUSIONS

Because this study was descriptive in nature, the final chapter will provide a discussion of the important findings and identify additional questions and considerations that need to be explored further. The discussion also will address some of challenges discovered while implementing this study and how best to overcome them in future studies. The chapter will conclude with recommendations for practice and a detailed plan for future research on the topic of student-faculty interactions that occur in academic student organizations.

Discussion of Key Issues

The topic of student-faculty interactions has long been studied to determine what students learn from these encounters. Previous literature has found that interactions between students and faculty are minimal (Anaya & Cole, 2001; Cotten & Wilson, 2006; Fusani, 1994; Hagedorn, et al., 2000; Lewallen, 1995; Nadler & Nadler, 2001; Snow, 1973; Wilson, et al., 1974); typically associated with a student's major or career field (Alderman, 2008; Anaya & Cole, 2001; Fusani, 1994; Iverson, et al., 1984; Nadler & Nadler, 2001; Terenzini & Pascarella, 1980); usually occur before or after class or during a faculty member's office hours (Cotten & Wilson, 2006; Nadler & Nadler, 2001); and last less than 10 minutes (Jaasma & Koper, 1999; Nadler & Nadler, 2001). Research also identified characteristics of powerful interactions such as having a friendly exchange, addressing personal issues or interests, and students feeling comfortable with a faculty member (Baxter Magolda, 1987; Endo & Harpel, 1982).

Institutional size has been identified as another characteristic that could impact interactions with small campuses offering more opportunities for informal interactions such as visiting a faculty member's home while large campuses may offer more opportunities for formal interactions like working on a research project (Alderman, 2008; Dilley, 1967; Kezar & Kinzie, 2006; Tinto, 1975). The literature concludes that almost all college students experience an interaction with faculty while in college even though they may be brief and infrequent overall (Rosenthal, et al., 2000; Terenzini & Wright, 1987).

This study validated a few of these conclusions with most participants saying they had at least one interaction with a faculty member since they started college and that these interactions could be considered infrequent in the context of a student's entire college career. In addition, small institutions appeared to provide more opportunities for student-faculty interactions than did large institutions.

However, this study also found contradictory evidence to previous conclusions made by researchers. Participants in this study reported having at least one interaction with faculty which did not fall into the general category of occurring before or after class or during office hours. These interactions were occurring in the context of an academic student organization, with faculty members interacting with students during organizational meetings and activities. In addition, many of the participants reported interactions lasting more than 10 minutes in length, which was the standard used by some researchers to measure the quality of interactions (Nadler & Nadler, 2001; Terenzini, et al., 1984a; Volkwein, et al., 1986).

Research has repeatedly identified involvement in co-curricular activities, and serving as a leader in those activities, as a useful experience where students can learn new skills and network with peers and college officials (Astin, 1993, 1999; Foubert & Grainger, 2006; Graham & Gisi, 2000; Hernandez, et al., 1999; Huang & Chang, 2004; Kuh, et al., 2005; Kuh, et al., 1991; Moore, et al., 1998; Pascarella & Terenzini, 2005; Stanford, 1992; Tinto, 1975). While some researchers previously have explored the idea that student-faculty interactions may occur within students' involvement experiences, they concluded that no meaningful interactions were occurring in these settings (Kuh & Hu, 2001; Volkwein, et al., 1986). However, the basic premise of this research was flawed. Researchers who study involvement regularly defined it as a broad concept with no attention paid to the different activities and outcomes associated with the type of organization a student chooses to join (Kuh & Hu, 2001; Pascarella & Terenzini, 1991; Terenzini, et al., 1996; Volkwein, et al., 1986). Dismissing the contextual environment resulted in important discoveries being overlooked. This study illustrates that students and faculty are interacting through the context of student organizations, but the right questions must be asked in order to discover the true nature of those interactions.

For instance, past researchers neglected to ask students if they had a faculty advisor for their student organizations (See Appendix A). While faculty advisors may not be prevalent for all types of student organizations, it is likely that academic organizations will attract the attention of faculty because of the interests they share with students who join the organizations. In this study, almost all of the participants, who came from four institutions and nineteen student organizations, indicated that they had a faculty advisor for their academic organization. Student organization advisors interact with at least some student members on a regular basis in order to conduct organizational business and activities, so having a faculty member serving in this role provides some automatic interactions for students who belong to these organizations. This finding alone indicates that the research on student-faculty interactions within involvement opportunities is incomplete and should be reexamined for deficiencies.

Campuses have evolved over the past 30 years as societal needs have changed, yet research on student-faculty interactions routinely asks students about the interactions that were identified for the earliest investigations on the topic (See Appendix A). It has only been in the last seven years that new interactions have been explored, and those were from the faculty perspective and not from the student perspective (Einarson & Clarkberg, 2004). In addition, existing studies do not address where these interactions are occurring, only that they do occur somewhere in the college environment (See Appendix A). This study gathered all of the interactions that have been identified in the last thirty years and asked participants to identify not only if the interactions occurred but also in what context they occurred. Three-fourths of the participants in this study experienced interactions in their academic student organizations as well as their other college activities. This finding emphasizes the necessity of modernizing the research on this topic in order to obtain the true picture of what is occurring what interactions may

now be occurring on college campuses that were not available when the research topic was introduced (e.g., technology).

Another interesting finding of this study concerned the introduction of a composite score for measuring interactions. While existing literature explored the concepts of quantity and quality as separate entities (Clark, et al., 2002; Cotten & Wilson, 2006; Endo & Harpel, 1982; Kuh & Hu, 2001; Kuh, et al., 2005; Pascarella, et al., 1983; Pascarella & Terenzini, 1975, 1979b, 2005; Strauss & Volkwein, 2004; Terenzini & Pascarella, 1980; Terenzini, et al., 1984a; Volkwein, et al., 1986), this study put the two together in a composite score to try and understand the interaction of both characteristics. Astin's Theory of Involvement (1984) provided a rationale for creating a composite score and using it to measure interactions between students and faculty. When the interactions occur, both students and faculty are exerting energy in the exchange. These interactions have both qualitative and quantitative features that can impact how much energy is involved. For example, a longer interaction will require more energy from the participants. And the more energy they exert, the more meaningful the interaction will become. The composite score used in this study followed this logic to try and determine how much energy or "power" was being exhibited during the interactions.

The composite score for this study identified many of the traditional interactions (See Appendix A) as powerful, or of high quantity and high quality. These interactions included activities such as having sodas or snacks with a faculty member or discussing academic and career issues with them. These results help provide face validity the study

and highlight some things that might be changing on modern college campuses. For instance, some of the interactions such as interacting with faculty to gain advice on personal issues have traditionally been considered powerful but did not rate as powerful for this study. While it might be easy to dismiss this finding because of the descriptive nature of the study, the face validity provided from other results point to the possibility that additional factors may need to be considered such as the changes that have occurred on college campuses over time.

For instance, seeking advice from faculty regarding personal issues was identified as a meaningful interaction in the past but did not emerge as one for this study. Thirty years ago, faculty may have been seen as a resource for personal issues because students lacked immediate contact with people they normally would have turned to for advice. With the emergence of social networking and portable communication devices, students spend a great deal of time in constant contact with friends and family, obtaining immediate and ongoing support for whatever issues emerge in students' personal lives. In addition, these students may prefer to connect to other people through their online communities because it is a more immediate way of sharing their thoughts and interacting with a wide variety of people. Faculty may no longer serve the same role for students as they once did, so previously identified "powerful" interactions such as seeking advice for personal problems may no longer be relevant for today's students. The use of a composite score to measure the overall "power" of each interaction should be explored further to determine whether or not it is an appropriate and accurate measure of student-faculty interactions.

In addition to using a composite score for the first time, this study identified one characteristic that has been overlooked in previous research on the topic of student-faculty interactions – organizational role. While interactions with faculty were occurring for a majority of students in their academic student organizations, student leaders were found to have even more interactions with faculty in this context than did students who were members of the organization. This is an important finding because it addresses the importance of not only being involved in these organizations but of taking on additional responsibilities as a leader. While members merely attend meetings and activities, leaders must interact with advisors on a regular basis in order to perform responsibilities for the organization. This would provide them greater access to faculty interactions than the members who may just attend general meetings and a few events throughout the school year.

By virtue of their leadership position in a campus organization, leaders also may have access to other faculty members through campus governance activities or academic functions that require student participation. For instance, if the faculty advisors attend a department meeting where they are asked to identify students who could provide feedback on a proposed curricula change, the advisors may naturally think of the students with whom they work most closely and forward their names. The student leaders then are contacted by other faculty members in the department in order to provide that feedback. The frequency of interactions could continue to increase because of the leadership position students hold, with the result that leaders have more powerful interactions with faculty than members.

Recommendations for Practice

Kuh (1995) explained, "Policies and practices should be designed to encourage students to take responsibility for their own affairs, interact frequently with members of different groups in various settings, and apply knowledge gained in the classroom to other areas (for example, employment, community affairs)" (p. 150). While these are useful goals to have, researchers have demonstrated that students are unsure of how to meet and interact with faculty and need direct guidance and activities that bring the two groups together (Anderson & Carta-Falsa, 2002; Cotten & Wilson, 2006). This study demonstrated institutions may already have resources, in the form of academic student organizations, that can be examined as possible avenues for bringing students and faculty together.

For instance, institutions regularly provide orientation sessions for new students, especially incoming freshmen. These orientation sessions could include a list of academic student organizations available for specific majors and areas of study. In addition, the campus administrators and faculty who participate in the orientation sessions can encourage new students to join an academic organization. Once students are directed to these organizations, academic colleges and departments can turn the majority of their non-classroom attention to these organizations and begin to establish regular interactions. These interactions could take the form of faculty serving as the organizational advisor, leading an annual career-related field trip, or speaking to the members of an organization about the skills they need to work in their chosen career. While the main goal of this process would be to bring faculty and students together in a structured environment, student affairs professionals also could play an essential role in creating this environment. Faculty may not have the financial resources or event planning background to create some of the interaction opportunities. For instance, taking a group of students on a field trip requires money and time. Student affairs professionals could set aside a portion of their programming budgets to help fund these kinds of trips, and then work with the student leaders of the organization to plan and execute the details. Getting the student affairs professionals involved would create a seamless partnership where the students received guidance on leadership skills, and then benefit from the knowledge provided by the faculty member who is escorting them on the trip.

Of course, this idea may have some inherent challenges to overcome. Einarson and Clarkberg (2004) concluded that some students and faculty need direct coaching on how to interact with each other. For instance, students may not understand the life of a faculty member, so they may benefit from a step-by-step process on what to say to faculty members such as providing a proper greeting and asking questions about faculty research and teaching interests. On the other side, faculty may not understand the interests and needs of current students. Faculty may benefit from information regarding the technological devices used by students, the kinds of activities students participate in outside of class, and where students work. Faculty could even be provided tours of campus residence halls so they can understand a little more about students' lives. By providing a little information prior to the actual interactions, administrators can help students and faculty feel more comfortable with each other.

Another challenge might be faculty interaction styles. Snow (1973) found that faculty exhibited two different styles when they interacted with students. One group of faculty members was receptive to interacting with students outside of class and learning more about their personal lives. The other group of faculty members did not wish to move beyond only serving as a resource for students and taking care of needed tasks. These different approaches may need to be recognized by distributing roles based on style. For instance, the more interaction-oriented faculty members may do well serving as an organizational advisor while the task-oriented faculty members may appreciate serving as speakers for organizational events.

While students will learn from the faculty with whom they interact, Lohr (2004) found that faculty benefit from these interactions as well. Faculty who interacted with students expressed feeling a sense of purpose, having more enthusiasm for the college environment, receiving feedback about what students want from their instructors, and increasing their understanding of young people. Unfortunately, these are intangible rewards that may not be viewed as necessary by faculty reward structures. If academic student organizations are to be used as a primary vehicle for bringing students and faculty together, the faculty reward structures must be reviewed to ensure at least some recognition that there are benefits to spending time with students.

Addressing the reward structures and encouraging faculty to engage with students outside of class can also benefit the academic mission of the institution. One

study found that when faculty build relationships with students, those relationships can lead to additional financial resources or professional connections after those students graduate and become employed in a related field (Lohr, 2004). In essence, encouraging and rewarding faculty for interacting with students can help a department or college expand its professional network, thereby providing additional opportunities for academic growth and development.

While there are challenges to creating a structured environment for all students and faculty to interact, it may be easiest to begin with what already is happening on the campus. For instance, as this study demonstrated, academic student organizations may already have a faculty advisor. The student affairs division may already track this information, so these individuals could be easily identified. In addition, student affairs professionals could provide oversight and coordination in bringing these advisors together and discovering what kinds of interactions are already occurring and what additional financial or planning resources are needed in order to create more interaction opportunities. The identified interactions could be assessed and improved over time to ensure that both students and faculty benefit in ways that are important to the institutional mission.

Recommendations for Further Research

While it does not provide conclusive evidence, this descriptive study does provide a benchmark for the important issues that need to be addressed in future research on the topic of student-faculty interactions in academic student organizations. It is clear from this study that student-faculty interactions are occurring in the context of academic student organizations and that contextual environments matter in the questions that are asked. Based on the findings of this study, a thorough research plan can be devised for further investigations on the topic of student-faculty interactions within involvement experiences. The research plan should address barriers encountered in this study, identify all interactions with faculty that are occurring for students in all contexts, provide results that can be generalized to modern campuses, and consider additional variables that may impact student-faculty interactions in involvement activities.

The first part of the research plan should account for and remedy all barriers encountered in this study. One of the first problems discovered with examining studentfaculty interactions in academic student organizations was the administrative oversight provided by the institutions. On many college campuses, student organizations must follow a formal recognition process that assists the institution in identifying what groups are active and what leaders can be contacted for that group if someone needs more information. However, the recognition process varies among institutions. For this study, one of the originally selected institutions did not have a formal process. Identifying the appropriate organizational contacts on this campus became an insurmountable barrier because there was no deadline for organizations to update their information. Some of the contacts were more than three years old and the campus administrators providing oversight to the recognition process could not identify any current leaders in the organization or whether or not the organization had any active membership. Future research should use this basic administrative function as a way to identify where subjects should be recruited. If the institutions do not collect and

maintain this information on a regular basis, assessing the student experience on those campuses is challenging at best.

A related problem was finding another organizational contact when the student representative did not respond to initial requests for participation in the study. In these cases, the next line of contact was to the staff advisor. Campuses with good tracking systems for student organizations typically provide contact information for the organizational advisor. For all but one organization where the advisor had to be contacted for assistance in reaching the student representative, communicating with the advisor proved to be very beneficial in securing student participation. Future research should obtain both student and staff contact information prior to recruiting subjects in order to maximize participation. Researchers should not be hesitant about contacting the advisors, some of whom may be faculty members who understand and promote research and may encourage their students to participate.

While the administrative oversight of a student organizations database was a predictable problem, one issue was completely unexpected for this study. It initially was assumed that student organizations identified for this study would only have membership rosters containing current students. Both undergraduate and graduate classifications were expected and the study instrument was specifically designed to identify these classifications so only undergraduate responses would be used for the study. Therefore, it was quite surprising when alumni and faculty responded to the instrument. Further exploration revealed that some of the organizations in the study were of a professional nature, with full-fledged chapters being located on the campuses on behalf of a national organization rather than student-only chapters. Future research needs to account for this possibility by asking each organization if they have faculty and alumni within their membership rosters and recruiting appropriate subjects for the research questions being asked.

Classification also was a barrier for this study because only a handful of freshmen and sophomores responded to the instrument. While it is possible that recruiting more subjects would remedy the small cell sizes caused by this low underclassman participation, there may be another explanation for the low response rate. According to existing literature and developmental theory, as students progress in their college career, they become more interested in vocationally driven activities (Arminio & Loflin, 2003; Chickering & Reisser, 1993; Terenzini & Wright, 1987; Wessel, et al., 2003). The few studies that examined what kinds of students chose to become involved in academic student organizations discovered that they were dominated by upperclassmen (Arminio & Loflin, 2003; Holzweiss, et al., 2008). More research is needed to fully explore this possibility but it does appear that upperclassmen will be the likely subjects in future studies because underclassmen are in short supply in this context. Future researchers should consider that recruiting underclassmen may be too much of a challenge and that focusing on upperclassmen, at least at this stage of knowledge, would present a better use of resources.

A final barrier was the instrument itself. The instrument was structured in a way that participants may have found tedious. The participants who had to be removed from the database because of incomplete responses all dropped out of the survey by the third time through the interactions list and skipped questions in the other sections prior to dropping out. It may be more effective to redesign the survey to address each interaction one at a time starting with whether or not the participants had the interaction. If they say no, they can go on to the next interaction. If they say yes, they can be directed to the quantity and quality questions. While this also could be interpreted as tedious by some participants, they may move through some of the interactions at a faster pace and only spend time answering the questions that they resonate with. This could engage them long enough to complete the survey and provide the necessary data for examination.

The second part of the research plan should identify all interactions that students are having with faculty, in both the academic student organization context and within other college activities. The best way to identify the interactions would be to conduct indepth, individual interviews with several subjects at different institutions. The instrument developed for this study would provide a good resource for developing the interview protocol. Subjects could respond to the instrument prior to the interview, and then the researcher could conduct the interview by further exploring each interaction and context the subjects identified. Follow-up questions could be asked regarding the context in which specific interactions occurred, how long the interactions lasted, who else was present during the interactions, who the faculty member was, whether or not they have had additional contact with the faculty member since the initial interaction, and what kind of impact the subjects believe the interaction had on their own learning.

Conducting in-depth interviews could help identify how many faculty members students are interacting with while they are in college and how those faculty members might or might not be related to the academic student organization. For instance, are all of the interactions occurring with the faculty advisor or do they originate with other faculty members? Being able to question students further regarding their experiences would help enlighten the conversation regarding student-faculty interactions.

The interviews also could examine the debate over quantity and quality, and enlist students' assistance with clearly defining what constitutes a powerful interaction with faculty. One way to accomplish this goal would be to explain this study's use of a composite score and ask for a reaction and suggestions for improvement. For instance, what is an acceptable frequency of interactions for students? Does the frequency matter to students? How do they define a quality interaction? Does the length of the interaction factor into how powerful it is to students? Even though the range of quality and quantity measures used in this study reflected prior literature (Nadler & Nadler, 2001; Pascarella & Terenzini, 1979a; Terenzini & Pascarella, 1980; Terenzini, et al., 1984a, 1984b; Volkwein, et al., 1986), it may be that these ranges do not adequately capture the student experience.

Another possibility is that a new definition of "quality" may need to be created. The current reliance on time to define the quality of interactions misses other important characteristics such as how engaged the faculty member is with the student, the significance of the interaction to the student, and whether or not the interaction helped the student resolve an issue. Refining these measures, or creating new ones that are more relevant to the current student experience, may be necessary to truly understand what is occurring on today's college campus.

It also may be useful to try and classify student responses within the framework of Bloom's Taxonomy. It may be possible that the powerful interactions that students describe could fit into the higher orders of both the cognitive and affective taxonomies. Understanding what students perceive as important aspects of interactions with faculty could help researchers create a meaningful measure that could be used widely.

Another issue that should be considered in terms of developing an interview protocol is whether or not students are involved in living-learning communities and how those communities provide interactions with faculty, if at all. While only a small proportion of participants in this study belonged to such communities, a few did report interacting with faculty as part of those communities. Living-learning communities should not be overlooked as a possible way students and faculty interact. It is important to ask students specific questions about where they might be experiencing interactions with faculty because they may not recall this information without being prompted for it.

Similarly, the interview protocol should specifically ask about the use of technology to interact with faculty. Modern students use technology on a daily basis and may be more likely to contact faculty through technological channels than approach them in person. These interactions should be accounted for in the interviews so the role of technology can be better understood in the context of student-faculty interactions.

Once the interviews have been conducted and the full nature of interactions are better understood, the research plan can move into the third phase of providing results that can be generalized to modern campuses. The interview findings could inform the development of a new instrument or a revision of the current one in order to provide an

exhaustive list of interactions that students can experience with faculty. Each interaction could then be asked in terms of where it occurred such as in a classroom, during office hours, in an academic student organization, etc. Depending on responses, follow-up questions could explore how long the interaction lasted, how often students had experienced the interaction, whether or not they interacted with that faculty member more than once, and how they would rate the power of the interaction (as defined through the interviews).

In addition to the information regarding student-faculty interactions, institutional size should be collected as a data field on the instrument. In this study there were a few significant differences and a pattern of responses that indicated participants from small institutions may have a higher quality and quantity of interactions with faculty than their peers from large institutions. This idea is supported in the existing literature (Alderman, 2008; Dilley, 1967; Kezar & Kinzie, 2006; Tinto, 1975). However, participants in this study who attended large institutions also were experiencing interactions with faculty in both the academic student organization context and within other college activities, so it is important not to dismiss what is occurring in both types of institutions. Institutional size should be considered a factor worth exploring in future research until a better understanding can be developed of what may be different about the interactions provided within both sizes of institutions.

When an instrument has been created and tested, it should be implemented on a variety of campuses, in different regions of the country. The purpose of such a national implementation would be to gather definitive benchmark data that would explain what is

happening on different campuses and whether or not there are significant variations between variables. Once the benchmark data is obtained, the next phase would be to administer the instrument for academic student organizations on several different college campuses in order to understand what is happening in those contexts and how students and faculty build relationships in those environments.

To better inform the data collection, the implementation of the instrument should collect descriptive data regarding student's role in the organization, the length of his or her membership in the organization, and how many hours per week the student devotes to the organization. This study demonstrated that organizational role had a relationship to the quantity and quality of student-faculty interactions. The length of membership and time spent with organizational activities also may be factors for these interactions and it is important to continue exploring them until the impact is understood.

Another piece of information that should be collected on the instrument is students' plans for future academic preparation or career advancement. It may be that academic student organizations attract students who are already inclined to pursue advanced degrees or seek top leadership positions in their companies. Evaluating how these future plans may motivate a student to join academic student organizations could help identify which students are likely to become involved and how to encourage other students to consider joining.

The final part of the research plan should move beyond this study's focus on academic student organizations. For instance, other types of organizations also could provide students with opportunities to interact with faculty. One example is service based organizations. Faculty members may be involved in community groups that seek the assistance of these organizations to provide volunteers for events. Or faculty members could turn to these organizations to help with service learning projects associated with an academic course. How do these faculty roles impact the students? What do they learn from working side by side with a faculty member on a volunteer project?

In addition to service organizations, other types of organizations may have some inherent faculty interactions. Student governance groups often tackle important campus issues that need faculty input or guidance. Do the students who consult faculty members learn more about the faculty role on campus? How do their interactions with faculty in this context promote understanding between the two parties? Programming councils design activities that complement some academic courses such as featuring speakers that address issues important to some classrooms. Faculty may be contacted for speaker ideas or to market the program to their students. What do students take away from these speaker experiences that they may not obtain in other environments? Greek societies often ask faculty members to serve as academic advisors to their chapter members. How does this advising role impact the student members? Other organizations and activities that could be explored further include recreational and sports groups, study abroad programs, and religious organizations. It is important to continue identifying and exploring the different contextual environments where students and faculty might come together for a common purpose.

Obtaining the faculty perspective for interactions in different contextual environments is another important issue to examine. It may be useful to start with the faculty who are serving as advisors for academic student organizations. Based on this study, faculty advisors may be the norm for academic student organizations. Nineteen organizations on four campuses were part of this study and almost all of them had faculty advisors. This one factor could have a real impact on students because participating in this type of campus organization automatically puts them in contact with faculty members. In this informal context, students could feel more comfortable in approaching these faculty advisors and forming long-term relationships with them outside of the classroom.

More needs to be understood about who these faculty advisors are and what kinds of relationships they are forming with students. Are they tenured faculty? Do they belong to the same academic department or are the elsewhere in the college? Do they have some of the students in their classes as well as the organization? How does that student-faculty relationship evolve over time? What topics of discussion are frequently encountered by these faculty advisors? Are they professionally rewarded in some manner for their service to these organizations? Do they receive any personal rewards for serving in this organizational role? What is their motivation for taking on the role of organizational advisor? Do they have specific personality characteristics that encourage them to seek out opportunities to work with students in this context? Does gender or ethnicity matter when faculty members serve as organizational advisors? Do they develop relationships with specific types of students such as male students when the

134

advisor is male? What is the culture of their academic discipline and how does that culture impact their role as an advisor?

If research reveals that other faculty members outside of advisors are interacting with student members of the organizations, who are these faculty members and what role do they play with the organization? How many times do they interact with students in the organization and what kind of interactions do they participate in? Asking all of these questions is essential to understanding the faculty perspective of the interactions that are occurring in the academic student organization environment.

In addition to understanding the faculty perspective in greater detail, more information is needed on the impact of classification. It would be helpful to understand how students become engaged in career development issues and when they decide to join academic student organizations. This study clearly demonstrated that students do have opportunities to interact with faculty in the context of these organizations but the opportunities may only be reaching upperclassmen since they seem more likely to join the organizations. It is possible that if freshmen can be connected to these kinds of organizations during their first year in college, more interactions with faculty could be possible throughout their college career.

This possibility should be explored further by conducting a longitudinal study with incoming freshmen. They could be interviewed two or three times per year during their college career in order to discover how they view career development over time and what might motivate them to become involved in organizations that can help prepare them for future jobs. Identifying when students begin seriously focusing on career development issues could help colleges connect students to appropriate resources at the right time or help them become interested in career development activities at an earlier stage.

While all freshmen may need to be encouraged to explore career development opportunities such as joining academic student organizations, it could be especially important for freshmen who come from lower socio-economic backgrounds. It has been demonstrated in previous research that students from low socio-economic backgrounds are less involved in student organizations than students from higher socio-economic backgrounds (Walpole, 2003). These students may need to be examined further to determine if they differ from the general freshman population in terms of getting involved in academic student organizations or other activities that put them into contact with faculty. Interviewing them several times a year during their college career could help identify any special needs they have and how institutions can best address these needs.

Community colleges may present another opportunity for research. Academic student organizations exist on community college campuses but may experience low membership due to the varying needs of the non-traditional population that attends these institutions. Further exploration regarding the student-faculty interactions that may be occurring in the existing organizations could provide an opportunity to identify any beneficial outcomes students may experience. If benefits do exist, community college administrators may be able to use that information to promote more involvement in

136

academic student organizations and increase the opportunities for students and faculty to come together.

Finally, the student-faculty interactions in the different contextual environments should be examined for long-term benefits. For instance, do students who join academic student organizations and have interactions with faculty benefit in terms of increased job offers or better preparation for employment? Are they more likely to contribute financial support to their alma mater because of their experiences? What role do these interactions play in academic performance, satisfaction with the college experience, retention, and career aspirations? Understanding the long-term impact of the contextual environments and the interactions found within them is important for identifying what collegiate activities provide the greatest potential for development given existing resources.

Conclusions

This study highlighted the importance of considering the contextual environment when researching student-faculty interactions. When the context of academic student organizations was brought into focus, students reported interactions with faculty that were both frequent and lengthy. Future research must focus on isolating these contextual environments and asking specific questions about the experiences of students in these environments.

The importance of isolating contextual environments and trying to understand them in detail can be best summarized by Kuh (1995):

Colleges cannot force students to participate in organized campus activities or perform leadership roles. However, they can and should be accountable for

creating the conditions that promote such behavior. Policies and practices should be designed to encourage students to take responsibility for their own affairs, interact frequently with members of different groups in various settings, and apply knowledge gained in the classroom to other areas (p. 150).

This study is the first step in discovering how academic student organizations and other involvement activities can be utilized for increasing student learning. For instance, new students receive a variety of resources to help them prepare for college. Those resources can incorporate recommendations for what students need to do to achieve success in college including why they should initiate contact with faculty members, what that process might look like, and what can be gained from it. Explaining to them that they could find faculty members in academic student organizations might lead more freshmen to join these groups.

Implementing all of the recommendations for further research could help colleges and universities identify the opportunities that are already bringing students and faculty together. Doing so would benefit institutions through better management of current resources and providing further evidence that these resources are benefiting student learning.

REFERENCES

About Chemistry Club (n.d.). Retrieved September 29, 2009, from http://www.studentorgs.txstate.edu/chemclub/about.html

About KDP (2008). Retrieved May 13, 2009, from

http://www.kdp.org/aboutkdp/index.php

About Psi Chi (2009). Retrieved May 13, 2009, from http://www.psichi.org/About/

ACS Profile (2009). Retrieved May 13, 2009, from

http://studentactivities.tamu.edu/online/organization/OTQ5NDIw/profile

- Alderman, R. V. (2008). Faculty and student out-of-classroom interaction: Student perceptions of quality of interaction. Unpublished Dissertation, Texas A&M University, College Station, TX.
- Altbach, P. G. (2005). Harsh realities: The professoriate faces a new century. In P. G.
 Altbach, R. O. Berdahl & P. J. Gumport (Eds.), *American higher education in the twenty-first century: Social, political, and economic challenges* (2nd ed.).
 Baltimore, MD: The Johns Hopkins University Press.
- Anaya, G., & Cole, D. G. (2001). Latina/o student achievement: Exploring the influence of student-faculty interactions on college grades. *Journal of College Student Development*, 42(1), 3-14.
- Anderson, E. L. (2002). The new professoriate: Characteristics, contributions, and compensation (No. ED478300). Washington, DC: American Council on Education, Center for Policy Analysis.

- Anderson, L. E., & Carta-Falsa, J. (2002). Factors that make faculty and student relationships effective. *College Teaching*, *50*(4), 134-138.
- Arminio, J., & Loflin, S. E. (2003). The impact of involvement on students at various points in their college careers. In T. L. Skipper & R. Argo (Eds.), *Involvement in campus activities and the retention of first-year college students (Monograph No. 36)* (pp. 35-49). Columbia, SC: University of South Carolina, National Resource Center for The First-Year Experience and Students in Transition.
- Arnold, K., & Kuh, G. D. (1999). What matters in undergraduate education? Mental models, student learning, and student affairs. In E. J. Whitt (Ed.), *Student learning as student affairs work* (pp. 11-34). Washington, DC: National Association of Student Personnel Administrators.
- Association of Texas Professional Educators (2009). Retrieved May 13, 2009, from http://www.atpe.org/
- Astin, A. W. (1984). Student involvement: A developmental theory for higher education. *Journal of College Student Personnel*, 25(4), 297-308.
- Astin, A. W. (1993). *What matters in college? Four critical years revisited* (1st ed.). San Francisco: Jossey-Bass.
- Astin, A. W. (1999). Involvement in learning revisited: Lessons we have learned. Journal of College Student Development, 40(5), 587-598.
- Baxter Magolda, M. B. (1987). The affective dimension of learning: Faculty-student relationships that enhance intellectual development. *College Student Journal*, 21, 46-58.

- Bean, J. P., & Kuh, G. D. (1984). The reciprocity between student-faculty informal contact and academic performance of university undergraduate students. *Research in Higher Education*, 21(4), 461-477.
- Beeny, C. K. (2003). Perceptions of learning in the co-curriculum: A study of involvement and expectations. Unpublished dissertation, University of Georgia, Athens.
- Berger, J. B., & Braxton, J. M. (1998). Revising Tinto's interactionalist theory of student departure through theory elaboration: Examining the role of organizational attributes in the persistence process. *Research in Higher Education, 39*(2), 103-119.
- Boyer, E. L. (1987). *College the undergraduate experience in America*. New York: The Carnegie Foundation for the Advancement of Teaching.
- Burke, J. C. (2004). The many faces of accountability. In J. C. Burke (Ed.), *Achieving accountability in higher education: Balancing public, academic, and market demands*. San Francisco: Jossey-Bass.
- Carnegie Foundation for the Advancement of Teaching (2009). *Size and setting description*. Stanford, CA: Carnegie Foundation for the Advancement of Teaching.
- Chickering, A. W., & Gamson, Z. F. (1987). Seven principles for good practice in undergraduate education. *AAHE Bulletin*, *17*(3), 140-141.
- Chickering, A. W., & Reisser, L. (1993). *Education and identity* (2nd ed.). San Francisco, CA: Jossey-Bass.

- Clark, R. K., Walker, M., & Keith, S. (2002). Experimentally assessing the student impacts of out-of-class communication: Office visits and the student experience. *Journal of College Student Development*, 43(6), 824-837.
- Cleveland-Innes, M. F., & Emes, C. (2005). Social and academic interaction in higher education contexts and the effect on deep learning. *NASPA Journal*, 42(2), 241-262.
- Cohen, A. M. (1998). The shaping of American higher education: Emergence and growth of the contemporary system. San Francisco: Jossey-Bass.
- Cotten, S., & Wilson, B. (2006). Student-faculty interactions: Dynamics and determinants. *Higher Education*, *51*(4), 487-519.
- Cutting, G. R. (1871). *Student life at Amherst College*. Amherst, MA: Hatch & Williams.
- Dilley, J. S. (1967). Student-faculty noncommunication. *Journal of College Student Personnel, 8,* 282-285.
- Einarson, M. K., & Clarkberg, M. E. (2004). Understanding faculty out-of-class interaction with undergraduate students at a research university. Paper presented at the Annual Meeting of the Association for the Study of Higher Education, Kansas City, MO.
- Endo, J. J., & Harpel, R. L. (1982). The effect of student-faculty interaction on students' educational outcomes. *Research in Higher Education*, *16*(2), 115-136.
- Ewell, P. T. (1991). Assessment and public accountability: Back to the future. *Change*, *23*(6), 12-17.

- Feldman, K., & Newcomb, T. (1969). The impact of college on students. San Francisco: Jossey-Bass.
- Foubert, J. D., & Grainger, L. U. (2006). Effects of involvement in clubs and organizations on the psychosocial development of first-year and senior college students. *NASPA Journal*, 43(1), 166-182.
- Frankel, R., & Swanson, S. R. (2002). The impact of faculty-student interactions on teaching behavior: An investigation of perceived student encounter orientation, interactive confidence, and interactive practice. *Journal of Education for Business*, 78(2), 85-91.
- Fusani, D. (1994). Extra-class communication: Frequency, immediacy, self-disclosure, and satisfaction in student-faculty interaction outside the classroom. *Journal of Applied Communication Research*, 22, 232-255.
- Gaff, J. G. (1973). Making a difference: The impacts of faculty. *Journal of Higher Education, 44*, 605-622.
- Garrett, M. D., & Zabriskie, M. S. (2003). The influence of living-learning program participation on student-faculty interaction. *Journal of College and University Student Housing*, *32*(2), 38-44.
- Gellin, A. (2003). The effect of undergraduate student involvement on critical thinking:
 A meta-analysis of the literature from 1991-2000. *Journal of College Student Development*, 44(6), 746-762.
- Gilligan, C. (1982). In a different voice: Psychological theory and women's development. Cambridge, MA: Harvard University Press.

- Golde, C. M., & Pribbenow, D. A. (2000). Understanding faculty involvement in residential learning communities. *Journal of College Student Development*, 41(1), 27-40.
- Graham, S. W., & Gisi, S. L. (2000). Adult undergraduate students: What role does college involvement play? *NASPA Journal*, *38*(1), 99-121.
- Guba, E. (1990). The alternative paradigm dialog. In E. Guba (Ed.), *Paradigm dialogue* (pp. 17-27). Thousand Oaks, CA: Sage.
- Hagedorn, L. S., Maxwell, W., Rodriguez, P., Hocevar, D., & Fillpot, J. (2000). Peer and student-faculty relations in community colleges. *Community College Journal of Research and Practice*, 24(7), 587-598.
- Halawah, I. (2006). The impact of student-faculty informal interpersonal relationships on intellectual and personal development. *College Student Journal*, 40(3), 670-678.
- Health Information Management Student Association (n.d.). Retrieved September 28, 2009, from https://sa.txstate.edu/studentorgs/public/index.asp?Org=582
- Hernandez, K., Hogan, S., Hathaway, C., & Lovell, C. D. (1999). Analysis of the literature on the impact of student involvement on student development and learning: More questions than answers? *NASPA Journal*, *36*(3), 184-197.
- Holzweiss, P., Rahn, R., & Wickline, J. (2008). Are all student organizations created equal? The differences and implications of student participation in academic versus non-academic organizations. *College Student Affairs Journal*, 27(1), 136-159.

- Huang, Y., & Chang, S. (2004). Academic and cocurricular involvement: Their relationship and the best combinations for student growth. *Journal of College Student Development*, 45(4), 391-406.
- Inkelas, K. K., & Weisman, J. L. (2003). Different by design: An examination of student outcomes among participants in three types of living-learning programs. *Journal* of College Student Development, 44(3), 335-368.
- Isaac, S., & Michael, W. B. (1990). *Handbook in research and evaluation* (2nd ed.). San Diego, CA: EdITS.
- Iverson, B. K., Pascarella, E. T., & Terenzini, P. T. (1984). Informal faculty-student contact and commuter college freshmen. *Research in Higher Education*, 21(2), 123-136.
- Jaasma, M. A., & Koper, R. J. (1999). The relationship of student/faculty out-of-class communication to instructor immediacy and trust and to student motivation. *Communication Education*, 48(1), 41-47.
- Johnson, D. I. (2004). Relationships between college experiences and alumni participation in the community. *The Review of Higher Education*, 27(2), 169-185.
- Johnson, J. L. (1997). Commuter college students: What factors determine who will persist or who will drop out? *College Student Journal*, *31*(3), 323-334.
- Keeling, R. P. (Ed.). (2004). Learning reconsidered: A campus-wide focus on the student experience. Washington, DC: National Association of Student Personnel Administrators and American College Personnel Association.

- Kezar, A., & Kinzie, J. (2006). Examining the ways institutions create student engagement: The role of mission. *Journal of College Student Development*, 47(2), 149-171.
- Kuh, G. (1995). The other curriculum: Out-of-class experiences associated with student learning and personal development. *Journal of Higher Education*, 66(2), 123-155.
- Kuh, G., Palmer, M., & Kish, K. (2003). The value of educationally purposeful out-ofclass experiences. In T. L. Skipper & R. Argo (Eds.), *Involvement in campus activities and the retention of first-year college students* (pp. 1-18). Columbia, SC: University of South Carolina, National Resource Center for the First-Year Experience and Students in Transition.
- Kuh, G. D. (1996). Guiding principles for creating seamless learning environments for undergraduates. *Journal of College Student Development*, 37(2), 135-148.
- Kuh, G. D., & Hu, S. (2001). The effects of student-faculty interaction in the 1990s.*Review of Higher Education*, 24(3), 309-332.
- Kuh, G. D., Kinzie, J., Schuh, J. H., & Whitt, E. J. (2005). Student success in college: Creating conditions that matter (1st ed.). San Francisco: Jossey-Bass.
- Kuh, G. D., Schuh, J. H., Whitt, E. J., Andreas, R. E., Lyons, J. W., Strange, C. C., et al. (1991). *Involving colleges: Successful approaches to fostering student learning and development outside the classroom*. San Francisco: Jossey-Bass.
- Lamport, M. A. (1993). Student-faculty informal interaction and the effect on college student outcomes: A review of the literature. *Adolescence*, *28*(112), 971-990.

- Lewallen, W. C. (1995). Students decided and undecided about career choice: A comparison of college achievement and student involvement. *NACADA Journal*, *15*(1), 22-30.
- Light, R. J. (2001). *Making the most of college: Students speak their minds*. Cambridge, MA: Harvard University Press.
- Lohr, S. C. (2004). *Student faculty connections: Grasping the essence of informal interaction.* Unpublished Dissertation, University of Southern California.
- Love, P. G. (1995). Exploring the impact of student affairs professionals on student outcomes. *Journal of College Student Development*, *36*(2), 162-170.
- Maisto, A. A., & Tammi, M. W. (1991). The effect of a content-based freshman seminar on academic and social integration. *Journal of the Freshman Year Experience*, 3(2), 29-47.
- Marsick, V. J., & Watkins, K. E. (2001). Informal and incidental learning. In S. B. Merriam (Ed.), *The new update on adult learning theory* (pp. 25-34). San Francisco: Jossey-Bass.
- Massy, W. F., & Zemsky, R. (1994). Faculty discretionary time: Departments and the academic ratchet. *Journal of Higher Education*, 65(1), 1-22.
- Merriam, S. B. (1991). How research produces knowledge. In J. M. Peters & P. Jarvis (Eds.), *Adult education* (pp. 42-65). San Francisco: Jossey-Bass.
- Milem, J. F., & Berger, J. B. (1997). A modified model of college student persistence: Exploring the relationship between Astin's theory of involvement and Tinto's

theory of student departure. *Journal of College Student Development, 38*(4), 387-400.

- Moore, J., Lovell, C. D., McGann, T., & Wyrick, J. (1998). Why involvement matters: A review of research on student involvement in the collegiate setting. *College Student Affairs Journal*, *17*(2), 4-17.
- Morrill, R. L. (2007). *Strategic leadership: Integrating strategy and leadership in colleges and universities*. Portsmouth, NH: Greenwood Publishing Group.
- Nadler, M. K., & Nadler, L. B. (2001). The roles of sex, empathy, and credibility in outof-class communication between faculty and students. *Women's Studies in Communication*, 24(2), 241-261.
- Nikolova Eddins, S. G., Williams, D. F., Bushek, D., Porter, D., & Kineke, G. (1997). Searching for a prominent role of research in undergraduate education: Project interface. *Journal on Excellence in College Teaching*, 8(1), 69-81.
- O'Meara, K., & Braskamp, L. (2005). Aligning faculty reward systems and development to promote faculty and student growth. *NASPA Journal*, *42*(2), 223-240.
- Pascarella, E. T., Duby, P. B., Terenzini, P. T., & Iverson, B. K. (1983). Student-faculty relationships and freshman year intellectual and personal growth in a nonresidential setting. *Journal of College Student Personnel*, 24(5), 395-402.
- Pascarella, E. T., & Terenzini, P. T. (1975). Informal interaction with faculty and freshman ratings of academic and non-academic experience of college. *Journal* of Educational Research, 70(1), 35-41.

- Pascarella, E. T., & Terenzini, P. T. (1979a). Interaction effects in Spady and Tinto's conceptual models of college dropout. *Sociology of Education*, 52(4), 197-210.
- Pascarella, E. T., & Terenzini, P. T. (1979b). Student-faculty informal contact and college persistence: A further investigation. *Journal of Educational Research*, 72(4), 214-218.
- Pascarella, E. T., & Terenzini, P. T. (1980a). Predicting freshman persistence and voluntary dropout decisions from a theoretical model. *Journal of Higher Education*, 51(1), 60-75.
- Pascarella, E. T., & Terenzini, P. T. (1980b). Student-faculty and student-peer relationships as mediators of the structural effects of undergraduate residence arrangement. *Journal of Educational Research*, 73(6), 344-352.
- Pascarella, E. T., & Terenzini, P. T. (1981). Residence arrangement, student/faculty relationships, and freshman-year educational outcomes. *Journal of College Student Personnel*, 22(2), 147-156.
- Pascarella, E. T., & Terenzini, P. T. (1991). *How college affects students: Findings and insights from twenty years of research*. San Francisco, CA: Jossey-Bass.
- Pascarella, E. T., & Terenzini, P. T. (1998). Studying college students in the 21st century: Meeting new challenges. *Review of Higher Education*, 21(2), 151-165.
- Pascarella, E. T., & Terenzini, P. T. (2005). *How college affects students: A third decade of research*. San Francisco: Jossey-Bass.

- Pascarella, E. T., Terenzini, P. T., & Wolfe, L. M. (1986). Orientation to college and freshman year persistence/withdrawal decisions. *Journal of Higher Education*, 57(2), 155-175.
- Phi Epsilon Kappa (2009). Retrieved May 13, 2009, from http://www.phiepsilonkappa.org/
- Phillips, D. C., & Burbules, N. C. (2000). *Postpositivism and educational research*.Lanham, MD: Rowman & Littlefield Publishers, Inc.
- Pike, G. R. (1999). The effects of residential learning communities and traditional residential living arrangements on educational gains during the first year of college. *Journal of College Student Development*, 40(3), 269-284.
- Pike, G. R., & Kuh, G. D. (2005). A typology of student engagement for American colleges and universities. *Research in Higher Education*, *46*(2), 185-209.
- Plecha, M. (2002). The impact of motivation, student-peer, and student-faculty interaction on academic self-confidence. Paper presented at the Annual Meeting of the American Educational Research Association (New Orleans, LA, April 1-5, 2002).
- Psychology Club/Psi Chi (n.d.). Retrieved September 29, 2009, from http://psycclub.tamu.edu
- Romanski, C. B. (1987). The impact of faculty-student interaction on student academic performance and persistence two years and four years after matriculation.
 Unpublished Dissertation, University of California, Los Angeles.

- Rosenthal, G. T., Folse, E. J., Alleman, N. W., Boudreaux, D., Soper, B., & Von Bergen,
 C. (2000). The one-to-one survey: Traditional versus non-traditional student
 satisfaction with professors during one-to-one contacts. *College Student Journal*, 34(2), 315-320.
- Rudolph, F. (1990). *The American college and university: A history*. Athens, GA: The University of Georgia Press.
- Sarraf, S., Johnson, S., Davis, L., & Ahren, C. (2008, March 31). No time on my side: Examining factors affecting co-curricular involvement. Paper presented at the Association of College Personnel Administrators, Atlanta, GA.
- Sax, L. J., Bryant, A. N., & Harper, C. E. (2005). The differential effects of studentfaculty interaction on college outcomes for women and men. *Journal of College Student Development*, 46(6), 642-659.
- Sax, L. J., Gilmartin, S. K., & Bryant, A. N. (2003). Assessing response rates and nonresponse bias in web and paper surveys. *Research in Higher Education*, 44(4), 409-432.
- Schwandt, T. A. (2001). *Dictionary of qualitative inquiry* (2nd ed.). Thousand Oaks, CA: Sage Publications.
- Sheldon, H. D. (1901). The history and pedagogy of American student societies.Worchester, MA: Clark University.
- Sigma Tau Delta profile (2009). Retrieved May 13, 2009, from http://studentactivities.tamu.edu/online/organization/OTUyMTIw/profile

- Skipper, T. L., & Argo, R. (2003). Shifting our focus on retaining first-year students and campus activities: An introduction. In T. L. Skipper & R. Argo (Eds.), *Involvement in campus activities and the retention of first-year college students*(pp. ix-xx). Columbia, SC: University of South Carolina, National Resource
 Center for the First-Year Experience and Students in Transition.
- Smith, J. K. (1983). Quantitative versus qualitative research: An attempt to clarify the issue. *Educational Researcher*, *12*(3), 6-13.
- Snow, S. G. (1973). Correlates of faculty-student interaction. *Sociology of Education*, *46*(4), 489-498.
- Spatz, C. (2005). *Basic statistics: Tales of distribution* (Eighth ed.). Belmont, CA: Thomson Wadsworth.
- Stanford, S. W. (1992). Extracurricular involvement and development among undergraduate student leaders. *College Student Affairs Journal, 12*, 17-24.
- Strange, C. C. (2004). Constructions of student development across the generations. In
 M. D. Coomes & R. Debard (Eds.), *New directions for student services: Serving the millennial generation* (pp. 47-57). San Francisco, CA: Wiley Periodicals, Inc.
- Strauss, L. C., & Volkwein, J. F. (2004). Predictors of student commitment at two-year and four-year institutions. *Journal of Higher Education*, 75(2), 203-227.
- Strayhorn, T. L. (2008). How college students' engagement affects personal and social learning outcomes. *Journal of College & Character*, *10*(2), 1-16.

- Terenzini, P. T., & Pascarella, E. T. (1980). Student/faculty relationships and freshman year educational outcomes: A further investigation. *Journal of College Student Personnel*, 21(6), 521-528.
- Terenzini, P. T., Pascarella, E. T., & Bliming, G. S. (1996). Students' out-of-class experiences and their influence on learning and cognitive development: A literature review. *Journal of College Student Development*, 37(2), 149-162.
- Terenzini, P. T., Springer, L., Pascarella, E. T., & Nora, A. (1995). Academic and outof-class influences on students' intellectual orientations. *Review of Higher Education, 19*(1), 23-44.
- Terenzini, P. T., Theophilides, C., & Lorang, W. G. (1984a). Influences on students' perceptions of their academic skill development during college. *Journal of Higher Education*, 55(5), 621-636.
- Terenzini, P. T., Theophilides, C., & Lorang, W. G. (1984b). Influences on students' perceptions of their personal development during the first three years of college. *Research in Higher Education*, 21(2), 178-194.
- Terenzini, P. T., & Wright, T. M. (1987). Influences on students' academic growth during four years of college. *Research in Higher Education*, *26*(2), 161-179.
- Thompson, M. D. (2001). Informal student-faculty interaction: Its relationship to educational gains in science and mathematics among community college students. *Community College Review*, 29(1), 35-57.
- Tinto, V. (1975). Dropouts from higher education: A theoretical synthesis of recent research. *Review of Educational Research*, *45*(1), 89-125.

- Underwood, D., Kim, H., & Matier, M. (2000, May 21-24). To mail or to web: Comparisons of survey response rates and respondent characteristics. Paper presented at the Association for Institutional Research, Cincinnati, OH.
- Volkwein, J. F., King, M. C., & Terenzini, P. T. (1986). Student-faculty relationships and intellectual growth among transfer students. *Journal of Higher Education*, 57(4), 413-430.
- Walpole, M. (2003). Socioeconomic status and college: How SES affects college experiences and outcomes. *Review of Higher Education*, 27(1), 45-73.
- Wessel, R. D., Christian, N. T., & Hoff, A. K. (2003). Enhancing career development through the career success club. *Journal of Career Development*, 29(4), 265-276.
- What Is HOSA? (n.d.). Retrieved May 13, 2009, from http://www.hosa.org/whatis.html
- Wilson, R. C., Woods, L., & Gaff, J. (1974). Social-psychological accessibility and faculty-student interaction beyond the classroom. *Sociology of Education*, 47(1), 74-92.

APPENDIX A

SOURCE OF ITEMS USED IN SURVEY INSTRUMENT

1. Met informally with faculty to help resolve a personal problem

(Anaya & Cole, 2001; Hagedorn, et al., 2000; Pascarella, et al., 1983; Pascarella & Terenzini, 1979a; Pascarella & Terenzini, 1980b; Pascarella & Terenzini, 1981; Terenzini & Pascarella, 1980; Terenzini, Springer, Pascarella, & Nora, 1995; Terenzini, et al., 1984a; Terenzini & Wright, 1987; Thompson, 2001; Volkwein, et al., 1986; Wilson, et al., 1974)

2. Met informally with faculty to get basic information and advice about my academic program

(Nadler & Nadler, 2001; Pascarella, et al., 1983; Pascarella & Terenzini, 1979a; Pascarella & Terenzini, 1980b; Pascarella & Terenzini, 1981; Terenzini & Pascarella, 1980; Terenzini, et al., 1984a; Terenzini & Wright, 1987; Volkwein, et al., 1986; Wilson, et al., 1974)

3. Met informally with faculty to discuss a campus issue or problem

(Pascarella, et al., 1983; Pascarella & Terenzini, 1979a; Pascarella & Terenzini, 1980b; Pascarella & Terenzini, 1981; Terenzini & Pascarella, 1980; Terenzini, et al., 1984a; Terenzini & Wright, 1987; Volkwein, et al., 1986; Wilson, et al., 1974)

4. Met informally with faculty to discuss matters related to my future career such as career plans and ambitions

(Anaya & Cole, 2001; Hagedorn, et al., 2000; Kuh, et al., 2005; Pascarella, et al., 1983; Pascarella & Terenzini, 1979a, 1981; Terenzini & Pascarella, 1980; Terenzini, et al., 1995; Terenzini, et al., 1984a; Terenzini & Wright, 1987; Thompson, 2001; Volkwein, et al., 1986; Wilson, et al., 1974)

5. Asked a faculty member for information or guidance related to a course you were taking (grades, make-up work, assignments, etc.)

(Anaya & Cole, 2001; Berger & Braxton, 1998; Kuh, et al., 2005; Nadler & Nadler, 2001; Pascarella, et al., 1983; Pascarella & Terenzini, 1979a, 1981; Plecha, 2002; Terenzini & Pascarella, 1980; Terenzini, et al., 1995; Terenzini, et al., 1984a; Terenzini & Wright, 1987; Thompson, 2001; Volkwein, et al., 1986; Wilson, et al., 1974)

6. Had a faculty member advise or supervise you on a faculty research project (Anaya & Cole, 2001; Astin, 1993; Einarson & Clarkberg, 2004; Kuh, et al., 2005; Lewallen, 1995; Terenzini, et al., 1995; Walpole, 2003) 7. Had a faculty member advise or supervise you on a student research project (e.g., Honors thesis, independent study, etc.)

(Anaya & Cole, 2001; Astin, 1993; Einarson & Clarkberg, 2004; Kuh, et al., 2005; Lewallen, 1995; Terenzini, et al., 1995; Walpole, 2003)

- 8. Was a guest in a professor's home for a meal or social function (Astin, 1993; Einarson & Clarkberg, 2004; Lewallen, 1995; Plecha, 2002; Walpole, 2003)
- **9. Worked with a faculty member outside of class on a committee or project** (Kuh, et al., 2005)

10. Had coffee, cokes, snacks, or meals with a professor (either in on-campus dining locations or off-campus at restaurants)

(Anaya & Cole, 2001; Einarson & Clarkberg, 2004; Terenzini, et al., 1995)

11. Talked informally (and outside of class) with an instructor about current events, campus activities, or other common interests

(Hagedorn, et al., 2000; Lewallen, 1995; Pascarella, et al., 1983; Pascarella & Terenzini, 1979a, 1981; Terenzini & Pascarella, 1980; Terenzini, et al., 1984a; Terenzini & Wright, 1987; Thompson, 2001; Volkwein, et al., 1986; Wilson, et al., 1974)

12. Assisted a faculty member in teaching a class

(Astin, 1993; Lewallen, 1995; Walpole, 2003)

13. Asked a faculty member for comments or criticisms about your work (tests, papers, etc.)

(Anaya & Cole, 2001; Astin, 1993; Terenzini, et al., 1995; Thompson, 2001)

14. Discussed ideas for a term paper or other class project with a faculty member

(Anaya & Cole, 2001; Terenzini, et al., 1995; Thompson, 2001)

15. Had a faculty member accompany you to an athletic competition or event (this could include other people as well)

(Einarson & Clarkberg, 2004)

16. Had a faculty member attend an arts or cultural event with you (this could include other people as well)

(Einarson & Clarkberg, 2004)

17. Had a faculty member supervise a student social function (e.g., party) that you attended

(Einarson & Clarkberg, 2004)

18. Had a faculty advisor for a student organization or committee that you belong(ed) to

(Einarson & Clarkberg, 2004)

19. Personally invited a faculty member to speak at an event sponsored by your organization

(Einarson & Clarkberg, 2004)

20. Had a faculty member help organize or attend a field trip with your student organization

(Einarson & Clarkberg, 2004)

21. Had a faculty member assist in organizing a presentation or workshop sponsored by your organization

(Einarson & Clarkberg, 2004)

22. Had a faculty member speak at a presentation or workshop sponsored by your organization

(Einarson & Clarkberg, 2004)

23. Had a faculty member attend meetings for your student organization (Einarson & Clarkberg, 2004)

24. Had a faculty member actively participate in meetings for your student organization

(Einarson & Clarkberg, 2004)

25. Worked with a faculty member outside of class on a committee or project sponsored by your organization

(Kuh, et al., 2005)

APPENDIX B

SURVEY INSTRUMENT

ACADEMIC STUDENT ORGANIZATION SURVEY

THIS INFORMATION IS REQUIRED BY THE HUMAN SUBJECTS BOARD. PLEASE READ THEN CONTINUE ON TO THE QUESTIONS.

This survey is designed to collect information on what type of interactions students have with faculty members. Your responses will provide information about the quantity and quality of these interactions as well as under what circumstances they might occur.

You were selected for participation in this study because of your enrollment at one of four preselected colleges and universities as well as your membership in an academic student organization on the selected campus.

It is important that you are aware of the following factors about this survey:

- 1. You were specifically selected to participate in this study. Four colleges and universities were pre-selected to participate in the study because of specific institutional characteristics. After the institutions were selected, several academic student organizations were identified as common to each of the institutions. You were identified as a member of one of these organizations and therefore you are receiving this survey to request your participation in the study. Your participation is very important to the success of the project because only a limited number of students in the state of Texas were selected to contribute their experiences.
- 2. **Participation is confidential.** While email addresses are known at the beginning, they will be separated from the rest of the data prior to analysis so no one will how you responded to specific questions. In addition, results will only be reported in groups so your individual information will never be shared.
- 3. **Participation is low-risk and completely voluntarily.** You are under no obligation to participate and will not receive any benefits to do so. You may skip any question that makes you uncomfortable and you may stop responding at any time.
- 4. All individually identifying information will be removed from the database prior to analysis and destroyed. No personal information will be stored with the data files.
- 5. By completing the survey, you certify that you are *18 years of age or older* and that you are consenting to participate in this study. Please print off this information for future reference as your consent form.
- 6. The survey will take approximately 10 minutes to complete depending on your responses.

- 7. Results of the survey will be available to you at the conclusion of the project. If you are interested in receiving a copy of the final dissertation, the last question on the survey allows you to put in an email address where the final dissertation can be sent to you.
- 8. Responses to the survey need to be completed by November 13, 2009.
- 9. This project will contribute towards the fulfillment of a doctoral degree for the researcher and may be published in the future for use by other researchers.

If you have questions or concerns about this project, please contact Peggy Holzweiss at (979) 575-2585 or peggy-h@tamu.edu.

Sincerely,

Peggy Holzweiss Doctoral Student Department of Educational Administration and Human Resource Development Texas A&M University

This research study has been reviewed by the Institutional Review Board – Human Subjects in Research, Texas A&M University. For research-related problems or questions regarding subjects' rights, you can contact the Institutional Review Board through Ms. Melissa McIlhaney, IRB Program Coordinator, Office of Research Compliance, (979) 458-4067, mcilhaney@tamu.edu.

<page break>

What college or university do you attend?

McMurry University Texas A&M University Texas State University – San Marcos University of St. Thomas

What is your classification?

Freshman Sophomore Junior Senior Other: <text box>

What is your gender? Female Male Are you currently involved with a student organization that is focused on academic topics and/or professions (e.g., chemistry, science, psychology, health, education, English, etc.)? These organizations are typically affiliated with an academic college or department.

Yes No <goto learning community question>

<page break>

What academic student organization are you involved with on your campus? <text box>

Are you a member or a leader (e.g., in a specific, formal position) in the organization? Member Leader

Approximately how many active members are in your organization this semester? <text box>

Do you have a faculty advisor for your academic student organization? Yes No

<page break>

Do you currently participate in a formal living/learning community (e.g., you enrolled or were selected to participate in a residential program where you live with other students in your major, take classes with them, and participate in academic activities with them)?

Yes

No

(If yes) Do you interact with faculty as part of that living/learning community? Yes No

<page break>

The rest of the survey will ask you to consider a list of interactions students may have with faculty members when they come to college. The list will appear three different times then the survey will conclude. For the first section, you will be asked if the interaction occurred since you started college and, if so, if the interaction occurred because of your academic student organization or other college experience.

The second section will ask you to estimate how many times the interaction has occurred since you started college.

The third section will ask you to estimate the average length of the interactions.

Please continue to the rest of the survey.

<page break>

Section 1: Why Did This Interaction Occur? (Select All That Apply)

Scale: 2=Occurred because of my academic student organization, 1=Occurred within my other experiences as a college student, 0=I have never experienced this interaction with a faculty member

- Asked a faculty member for comments or criticisms about your work (tests, papers, etc.)
- Asked a faculty member for information or guidance related to a course you were taking (grades, make-up work, assignments, etc.)
- Assisted a faculty member in teaching a class
- Discussed ideas for a term paper or other class project with a faculty member
- Had a faculty member advise or supervise you on a faculty research project
- Had a faculty member advise or supervise you on a student research project (e.g., Honors thesis, independent study, etc.)
- Had coffee, cokes, snacks, or meals with a professor (either in on-campus dining locations or off-campus at restaurants)
- Met informally with faculty to discuss a campus issue or problem
- Met informally with faculty to discuss matters related to my future career such as career plans and ambitions
- Met informally with faculty to get basic information and advice about my academic program
- Met informally with faculty to help resolve a personal problem
- Talked informally (and outside of class) with a faculty member about current events, campus activities, or other common interests
- Was a guest in a professor's home for a meal or social function
- Worked with a faculty member outside of class on a committee or project
- Had a faculty member accompany you to an athletic competition or event (this could include other people as well)

- Had a faculty member attend an arts or cultural event with you (this could include other people as well)
- Had a faculty member supervise a student social function (e.g., party) that you attended
- Personally invited a faculty member to speak at an event sponsored by a group with which you were affiliated
- Had a faculty member help organize or attend a field trip with a group with which you were affiliated
- Had a faculty member assist in organizing a presentation or workshop sponsored by a group with which you were affiliated
- Had a faculty member speak at a presentation or workshop sponsored by a group with which you were affiliated
- Had a faculty member attend meetings for a group with which you were affiliated
- Had a faculty member actively participate in meetings for a group with which you were affiliated
- Worked with a faculty member outside of class on a committee or project sponsored by a group with which you were affiliated

If there are other interactions you have had with faculty members that are not covered by this list, please explain them here: <text box>

<page break>

Section 2: How Many Times Has This Interaction Occurred Between You and a Faculty Member Since You Started College?

Scale: 5=5 or more times, 4=4 times, 3=3 times, 2=2 times, 1=1 time, 0=I have never experienced this interaction with faculty

- Asked a faculty member for comments or criticisms about your work (tests, papers, etc.)
- Asked a faculty member for information or guidance related to a course you were taking (grades, make-up work, assignments, etc.)
- Assisted a faculty member in teaching a class
- Discussed ideas for a term paper or other class project with a faculty member
- Had a faculty member advise or supervise you on a faculty research project
- Had a faculty member advise or supervise you on a student research project (e.g., Honors thesis, independent study, etc.)
- Had coffee, cokes, snacks, or meals with a professor (either in on-campus dining locations or off-campus at restaurants)
- Met informally with faculty to discuss a campus issue or problem

- Met informally with faculty to discuss matters related to my future career such as career plans and ambitions
- Met informally with faculty to get basic information and advice about my academic program
- Met informally with faculty to help resolve a personal problem
- Talked informally (and outside of class) with a faculty member about current events, campus activities, or other common interests
- Was a guest in a professor's home for a meal or social function
- Worked with a faculty member outside of class on a committee or project
- Had a faculty member accompany you to an athletic competition or event (this could include other people as well)
- Had a faculty member attend an arts or cultural event with you (this could include other people as well)
- Had a faculty member supervise a student social function (e.g., party) that you attended
- Personally invited a faculty member to speak at an event sponsored by a group with which you were affiliated
- Had a faculty member help organize or attend a field trip with a group with which you were affiliated
- Had a faculty member assist in organizing a presentation or workshop sponsored by a group with which you were affiliated
- Had a faculty member speak at a presentation or workshop sponsored by a group with which you were affiliated
- Had a faculty member attend meetings for a group with which you were affiliated
- Had a faculty member actively participate in meetings for a group with which you were affiliated
- Worked with a faculty member outside of class on a committee or project sponsored by a group with which you were affiliated

<page break>

Section 3: How Long was the Average Interaction of this Type That You Have Had Since You Started College?

Scale: 4=Over 60 minutes, 3=Between 31 and 60 minutes, 2=Between 16 and 30 minutes, 1=Less than 10 minutes, 0=I have never experienced this interaction with faculty

- Asked a faculty member for comments or criticisms about your work (tests, papers, etc.)
- Asked a faculty member for information or guidance related to a course you were taking (grades, make-up work, assignments, etc.)

- Assisted a faculty member in teaching a class
- Discussed ideas for a term paper or other class project with a faculty member
- Had a faculty member advise or supervise you on a faculty research project
- Had a faculty member advise or supervise you on a student research project (e.g., Honors thesis, independent study, etc.)
- Had coffee, cokes, snacks, or meals with a professor (either in on-campus dining locations or off-campus at restaurants)
- Met informally with faculty to discuss a campus issue or problem
- Met informally with faculty to discuss matters related to my future career such as career plans and ambitions
- Met informally with faculty to get basic information and advice about my academic program
- Met informally with faculty to help resolve a personal problem
- Talked informally (and outside of class) with a faculty member about current events, campus activities, or other common interests
- Was a guest in a professor's home for a meal or social function
- Worked with a faculty member outside of class on a committee or project
- Had a faculty member accompany you to an athletic competition or event (this could include other people as well)
- Had a faculty member attend an arts or cultural event with you (this could include other people as well)
- Had a faculty member supervise a student social function (e.g., party) that you attended
- Personally invited a faculty member to speak at an event sponsored by a group with which you were affiliated
- Had a faculty member help organize or attend a field trip with a group with which you were affiliated
- Had a faculty member assist in organizing a presentation or workshop sponsored by a group with which you were affiliated
- Had a faculty member speak at a presentation or workshop sponsored by a group with which you were affiliated
- Had a faculty member attend meetings for a group with which you were affiliated
- Had a faculty member actively participate in meetings for a group with which you were affiliated
- Worked with a faculty member outside of class on a committee or project sponsored by a group with which you were affiliated

<page break>

This concludes your participation in the study. If you would like a copy of the finished report, expected in May 2010, please enter an email address where you would like it to be sent in pdf format. Any email address you provide will be removed from the database

prior to analysis and will not be used for any purpose other than sending you the final report when it is ready. <text box>

<submit survey>

Thank you for taking time to respond to these questions. Your input will help create a greater understanding of the interactions between students and faculty members on college campuses. If you have any concerns about the questions or your participation in the study, please contact Peggy Holzweiss at (979) 778-2614 or peggy-h@tamu.edu.

VITA

Name:	Peggy Carol Holzweiss
Address:	Texas A&M University Allen Building 1082 College Station, TX 77843-4220
Education:	B.S., Psychology, Texas A&M University, 1991 M.S., Educational Administration, Texas A&M University, 1994 Ph.D., Educational Administration, Texas A&M University, 2010
Professional Experience: Assistant Director, Texas A&M University, 2010-present	
	Assessment Coordinator, Texas A&M University, 2002-2010
	Student Development Specialist, Texas A&M University, 1996-2002
	Director of Student Activities, Navarro College, 1994-1995
Professional Affiliations:	

National Association of Student Personnel Administrators